

# International American Education Federation, Inc., d/b/a International Leadership of Texas

August 27, 2020 Special Board Meeting

#### **Date and Time**

Thursday August 27, 2020 at 6:15 PM CDT

#### **Meeting Notice & Mission Statement**

In compliance with the Texas Open Meetings Act, the Texas Government Code, Chapter 551, timely public advance written notice (at least 72 hours before the scheduled time of the meeting) is given of the subjects to be considered by the Board of Directors of International Leadership of Texas (the "Board") and the Board will convene a Special Open Meeting of the Board of Directors of International Leadership of Texas on the date and time set forth herein.

#### **Special Notice: Public Meetings at Headquarters are Suspended Until Further Notice:**

International Leadership of Texas Board of Directors will be holding the above identified public board meeting at the time and place stated above. Members of the public will be able to watch the meeting via the link <a href="https://zoom.us/j/801651349">https://zoom.us/j/801651349</a> or by a link that will be posted on the ILTexas.org website (click the button "Board Meetings" to view any updates to this Notice).

If you would like to sign up to speak at the meeting, please send your name to <a href="mailto:board@iltexas.org">board@iltexas.org</a>, 24-Hours in advance so that we will be able to promote you to a panelist, which will allow you to speak with our board members.

It normally is the intent of the Board to have a quorum physically present at the above address and to allow any Board members not physically present to participate by live two-way video and audio feed in accordance with the Texas Open Meetings Act. However, due to the COVID-19 Coronavirus Pandemic and the governmental and administrative responses to that Pandemic, the Board intends to conduct the Meeting via videoconference. A quorum of the Board will not be physically present at the usual Headquarters address. Neither will the presiding officer physically present there. Nevertheless, the Meeting will be open to the public. Every effort will be made to facilitate public viewing of the Meeting while it is in progress via live video feed and, subject to limitations of the technology and logistics, to allow the usual opportunities for those who wish to speak to do so. Please access this Agenda via the ILTexas website as the time of the Meeting approaches, in order to find any additional information or updated links concerning the Meeting.

The Board hereby certifies that this notice was posted on a bulletin board or on something akin thereto or at a place readily accessible and convenient to the public at 1820 N. Glenville Dr., #100, Richardson, TX 75081, as well as online at www.ILTexas.org. The items on this Agenda may be taken in any order. The mission of ILTexas is to prepare students for exceptional leadership roles in the international community by emphasizing servant leadership, mastering the English, Spanish, and Chinese languages, and strengthening the mind, body and character.

/s/ Finn Simmensen, For ILTexas' Board

Αg	en	da
----	----	----

Purpose Presenter Time

I. Opening Items 6:15 PM

Opening Items

- A. Record Attendance and Guests
- B. Call the Meeting to Order

#### **II. Public Speakers**

**Board Services** 

A. Guests who spoke, if any

FYI

#### **III. Executive Session**

#### A. AUTHORIZATION

FYI

Closed Session for Any and All Reasons Permissible by Texas Law, including, but not limited to, Texas Government Code Sections 551.071, 551.072, 551.073, 551.074, 551.075, 551.076, 551.082, 551.083, 551.084, pertaining to any item listed on this agenda, as permitted by applicable law.

#### IV. Board Items for Discussion/Action

A. CONSIDER/ACT ON TIER ONE SECURITY CONTRACT

Vote James (Tim) Brightman

Hear presentation by Tier One Holdings, LLC for security services.

Discuss/Act to approve authorizing administration to contract with Tier One Holdings, LLC for security services.

**B.** CONSIDER/ACT ON CONTRACT WITH KEY CONSTRUCTION FOR PORTABLES INSTALLATION AT ILTEXAS COLLEGE STATION K-8

Vote

Charles Klein

Discuss/Take action to authorize administration to enter into a final Guaranteed Maximum Price (GMP) contract in the amount of \$63,599 with Key Construction for the portables project at ILTexas College Station K-8.

#### V. Closing Items

A. Adjourn Meeting

Vote

# **Cover Sheet**

## CONSIDER/ACT ON TIER ONE SECURITY CONTRACT

Section: IV. Board Items for Discussion/Action

Item: A. CONSIDER/ACT ON TIER ONE SECURITY CONTRACT

Purpose: Vote

Submitted by:

Related Material: Teir One Contract as of 20200827.pdf

### **Services Agreement**

This Services Agreement ("Agreement") is made this 12<sup>TH</sup> DAY OF AUGUST, 2020 (the "Effective Date") between International Leadership of Texas ("Customer"), with offices located at 1820 N Glennville Drive #100, Richardson, TX 75081 and Tier One Holdings, LLC dba Tier One ("Service Provider") with offices located at 7500 W Camp Wisdom Road, Suite A3-132 Hunt Building, Dallas, Texas 75236. The Customer and Service Provider are sometimes collectively referred to herein as the "Parties" and individually as a "Party".

Customer wishes to be provided with the services (collectively the "Services" as defined below) by Service Provider and the Service Provider agrees to provide the Services to the Customer in accordance with the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants and promises contained herein, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound, the Parties agree as follows:

#### 1. Services

- 1.01. The Customer hereby retains the Service Provider to perform the following Services on behalf of the Customer:
  - Full Time Contract Employees (Texas State Certified TCOLE Police Officers) during the school year from 2020 - 2025 school years (a five year contract) for the fall and spring semesters for approximately 10 hours per day during the scheduled school year (see Addendum A, attached hereto, and incorpated as part of this Agreement, for further details which include duty assignments and cost per officer as well as estimated number of officers per year).
  - Extra-curricular events are covered in the addendum to reflect officers hourly rate for anything outside of their normal daytime school hours.

Service Provider will determine the method, details, and means of performing the Services. Service provider respectfully requests break area for employee, dedicated workspace (may be shared space).

- **1.02.** The Parties acknowledge and agree that during the term of the Agreement the Services may be modified and/or expanded from time to time upon a written agreement executed by authorized representatives of the Parties expressly referencing this Agreement.
- **1.03.** Customer acknowledges and agrees that Service Provider may, at its sole discretion, use subcontractors and consultants to perform some of the Services to be provided under this Agreement. In the event Service Provider utilizes subcontractors or consultants to perform any of the Services, Service Provider shall remain responsible to Customer for performance under this Agreement and that they must meet the same requirements listed in section 1.01.
- **1.04.** Service Provider may represent, perform services for, and contract with other additional clients, persons, or companies as Service Provider, in its sole discretion, sees fit, provided those services do not pose a conflict of interest with the services performed for the Customer.

#### 2. Fees

**2.01. Fees.** As full consideration for the provision of the Services, Customer shall pay Service Provider fees monthly in advance. Extra curricular event hours will be billed separately (see addendum for details).

#### 2.02. Omitted.

- **2.03. Billing and Payment.** Service Provider shall prepare and submit an invoice to the Customer via email on or about the 15<sup>th</sup> of each month. Customer shall pay the billed amount not later than the 2<sup>nd</sup> day of the following month. The fees will cover the estimated hours for that forthcoming month. These fees will be paid in 12 monthly installments per year in an annualized manner.
- **2.04. Compliance with Laws; Permits and Licenses.** Customer agrees, at its own expense, to operate in full compliance with all governmental laws, regulations and requirements applicable to the duties conducted hereunder.

It shall be the responsibility of each party to pay for any necessary licenses, permits, insurance and approvals as may be necessary for the performance of the Services under this Agreement, unless otherwise specified in writing and agreed to by the Service Provider.

**3. Warranty.** The Services to be performed hereunder are security services. Service Provider does not warrant in any form the results or achievements of the Services provided or the resulting work product and deliverables. Service Provider warrants that that the Services will be performed by qualified personnel in a professional and workmanlike manner in accordance with the generally accepted industry standards and practices. Service Provider shall comply with all statutes, ordinances, regulations and laws of all international, federal, state, county, municipal or local governments applicable to performing the Services hereunder.

LIMITATION OF WARRANTY. THE WARRANTY SET FORTH IN THIS SECTION 3 IS EXCLUSIVE AND IS IN LIEU OF ALL OTHER WARRANTIES, EXPRESS, IMPLIED, STATUTORY OR OTHERWISE WITH RESPECT TO THE SERVICES, WORK PRODUCT OR DELIVERABLES PROVIDED UNDER THIS AGREEMENT, OR AS TO THE RESULTS WHICH MAY BE OBTAINED THEREFROM. SERVICE PROVIDER DISCLAIMS ALL IMPLIED WARRANTIES INCLUDING, BUT NOT LIMITED TO, THE WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, OR AGAINST INFRINGEMENT. SERVICE PROVIDER SHALL NOT BE LIABLE FOR ANY SERVICES OR WORK PRODUCT OR DELIVERABLES PROVIDED BY THIRD PARTY VENDORS IDENTIFIED OR REFERRED TO THE CUSTOMER BY THE SERVICE PROVIDER DURING THE TERM OF THIS AGREEMENT, PURSUANT TO ANY SOW OR OTHERWISE. CUSTOMER'S EXCLUSIVE REMEDY FOR BREACH OF THIS WARRANTY IS REPERFORMANCE OF THE SERVICES, OR IF REPERFORMANCE IS NOT POSSIBLE OR CONFORMING, REFUND OF AMOUNTS PAID UNDER THIS AGREEMENT FOR SUCH NON-CONFORMING SERVICES.

**4. Ownership of Work Product.** This is not a work-for-hire agreement. The copyright in all deliverables created hereunder for Customer shall belong to the Service Provider. All intellectual property rights in all pre-existing works and derivative works of such pre-existing works and other deliverables and developments made, conceived, created, discovered, invented or reduced to practice in the performance of the Services hereunder are and shall remain the sole and absolute property of Service Provider, subject to a worldwide, non-exclusive license to Customer for its internal use as intended under this Agreement, and the Service Provider retains all moral rights therein. This Agreement does not grant Customer any license to any of the Service Provider's products, which products must be separately licensed.

#### 5. Confidential Information

5.01. Confidential Information. The Parties acknowledge that by reason of their relationship to the other hereunder, each may disclose or provide access (the "Disclosing Party") to the other Party (the "Receiving Party") certain Confidential Information. "Confidential Information" shall mean (i) information concerning a Party's products, business and operations including, but not limited to, information relating to business plans, financial records, customers, suppliers, vendors, products, product samples, costs, sources, strategies, inventions, procedures, sales aids or literature, technical advice or knowledge, contractual agreements, pricing, price lists, product white paper, product specifications, trade secrets, procedures, distribution methods, inventories, marketing strategies and interests, algorithms, data, designs, drawings, work sheets, blueprints, concepts, samples, inventions, manufacturing processes, computer programs and systems and know-how or other intellectual property, of a Party and its affiliates that may be at any time furnished, communicated or delivered by the Disclosing Party to the Receiving Party, whether in oral, tangible, electronic or other form; (ii) the terms of any agreement, including this Agreement, and the discussions, negotiations and proposals related to any agreement; (iii) information acquired during any tours of or while present at a Party's facilities; and (iv) all other non-public information provided by the Disclosing Party hereunder. In no event shall Service Provider's use or disclosure of information regarding or relating to the development, improvement or use of any of Service Provider's products be subject to any limitation or restriction. To the extent allowed by law, all Confidential Information shall remain the property of the Disclosing Party.

**5.02. Use of Confidential Information; Standard of Care.** The Receiving Party shall, to the extent permitted by law, maintain the Confidential Information in strict confidence and disclose the Confidential Information only to its

employees, subcontractors, consultants and representatives who have a need to know such Confidential Information to fulfill the business affairs and transactions between the Parties contemplated by this Agreement. The Receiving Party shall always remain responsible for breaches of this Agreement arising from the acts of its employees, subcontractors, consultants and representatives. Receiving Party shall use the same degree of care as it uses with respect to its own similar information, but no less than a reasonable degree of care, to protect the Confidential Information from any unauthorized use, disclosure, dissemination, or publication. Receiving Party shall only use the Confidential Information in furtherance of its performance of its obligations under this Agreement, and agrees not to use the Disclosing Party's Confidential Information for any other purpose or for the benefit of any third party, without the prior written approval of the Disclosing Party. The Receiving Party shall not decompile, disassemble, or reverse engineer all or any part of the Confidential Information.

- **5.03. Exceptions.** Confidential Information does not include information that: (a) was lawfully in Receiving Party's possession before receipt from Disclosing Party; (b) at or after the time of disclosure, becomes generally available to the public other than through any act or omission of the Receiving Party; (c) is developed by Receiving Party independently of any Confidential Information it receives from Disclosing Party; (d) Receiving Party receives from a third party free to make such disclosure without, to the best of Receiving Party's knowledge, breach of any legal or contractual obligation, (e) required to disclose such information pursuant to the Texas Public Information Act, or (f) is disclosed by Receiving Party with Disclosing Party's prior written approval.
- **5.04. Required Disclosures.** If the Receiving Party is confronted with legal action to disclose Confidential Information received under this Agreement, the Receiving Party shall, unless prohibited by applicable law, provide prompt written notice to the Disclosing Party to allow the Disclosing Party an opportunity to seek a protective order or other relief it deems appropriate, and Receiving Party shall reasonably assist disclosing Party in such efforts. If disclosure is nonetheless required, the Receiving Party shall limit its disclosure to only that portion of the Confidential Information which it is advised by its legal counsel must be disclosed.
- **5.05.** Unauthorized Use or Disclosure of Confidential Information; Equitable Relief. In the event the Receiving Party discovers that any Confidential Information has been used, disseminated or accessed in violation of this Agreement, it will immediately notify the Disclosing Party, take all commercially reasonable actions available to minimize the impact of the use, dissemination or publication, and take all necessary steps to prevent any further breach of this Agreement. The Parties agree and acknowledge that any breach or threatened breach regarding the treatment of the Confidential Information may result in irreparable harm to the Disclosing Party for which there may be no adequate remedy at law. In such event the Disclosing Party shall be entitled to seek an injunction, without the necessity of posting a bond, to prevent any further breach of this Agreement, in addition to all other remedies available in law or at equity.
- **5.06. Return of Confidential Information; Survival.** Receiving Party shall promptly return or, at Disclosing Party's option, certify destruction of all copies of Confidential Information at any time upon request or within 30 days following the expiration or earlier termination of this Agreement. Notwithstanding any expiration or termination of this Agreement, Receiving Party's obligations to protect the Confidential Information pursuant to this Section will survive for two years after the expiration or earlier termination of this Agreement.
- **6. Indemnification.** Each party (the "Indemnifying Party"), to the extent allowed by law, agrees to indemnify, defend and hold the other party and its affiliates and their respective officers, directors, employees and agents harmless from and against all third-party claims, losses, liabilities, damages, expenses and costs, including attorney's fees and court costs, arising out of the Indemnifying Party's (i) negligence or willful misconduct or (ii) its material breach of any of the terms of this Agreement. The Indemnifying Party's liability under this Section shall be reduced proportionally to the extent that any act or omission of the other Party, or its employees or agents, contributed to such liability. The party seeking indemnification shall provide the Indemnifying Party with prompt written notice of any claim and give complete control of the defense and settlement of the Indemnifying Party, and shall cooperate with the Indemnifying Party, its insurance company and its legal counsel in its defense of such claim(s). This indemnity shall not cover any claim in which there is a failure to give the Indemnifying Party prompt notice to the extent such lack of notice prejudices the defense of the claim.

SECTION 6 STATES THE ENTIRE OBLIGATION AND THE EXCLUSIVE REMEDIES WITH RESPECT TO THE PARTIES' INDEMNIFICATION OBLIGATIONS PURSUANT TO THIS AGREEMENT.

#### 7. LIMITATION OF LIABILITY; ACTIONS

EXCEPT FOR THE PARTIES CONFIDENTIALITY OBLIGATIONS UNDER SECTION 5 OF THIS AGREEMENT AND INDEMNIFICATION OBLIGATIONS UNDER SECTION 6 OF THIS AGREEMENT, IN NO EVENT SHALL EITHER PARTY BE LIABLE UNDER THIS AGREEMENT TO THE OTHER PARTY FOR ANY INCIDENTAL, CONSEQUENTIAL, INDIRECT, STATUTORY, SPECIAL, EXEMPLARY OR PUNITIVE DAMAGES, INCLUDING, BUT NOT LIMITED TO, LOST PROFITS, LOSS OF USE, LOSS OF TIME, INCONVENIENCE, LOST BUSINESS OPPORTUNITIES, DAMAGE TO GOOD WILL OR REPUTATION, AND COSTS OF COVER, REGARDLESS OF WHETHER SUCH LIABILITY IS BASED ON BREACH OF CONTRACT, TORT, STRICT LIABILITY OR OTHERWISE, AND EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGES OR SUCH DAMAGES COULD HAVE BEEN REASONABLY FORESEEN. SUBJECT TO THE CUSTOMER'S OBLIGATION TO PAY THE FEES TO THE SERVICE PROVIDER, EACH PARTY'S ENTIRE AGGREGATE LIABILITY FOR ANY CLAIMS RELATING TO THE SERVICES OR THIS AGREEMENT SHALL NOT EXCEED THE FEES PAID OR PAYABLE BY THE CUSTOMER TO THE SERVICE PROVIDER UNDER THIS AGREEMENT IN THE 12 MONTH PERIOD IMMEDIATELY PRECEDING THE EVENTS GIVING RISE TO SUCH LIABILITY. THIS SECTION SHALL SURVIVE THE TERMINATION OF THE AGREEMENT.

NO ACTION SHALL BE BROUGHT FOR ANY CLAIM RELATING TO OR ARISING OUT OF THIS AGREEMENT MORE THAN ONE (1) YEAR AFTER THE ACCRUAL OF SUCH CAUSE OF ACTION, EXCEPT FOR MONEY DUE ON AN OPEN ACCOUNT.

- **8. Cooperation of Customer.** Customer agrees to comply with all reasonable requests of Service Provider and shall provide Service Provider's personnel with access to all documents and facilities as may be reasonably necessary for the performance of the Services under this Agreement.
- **9. Term.** This Agreement shall have an initial term of five (5) years from the Effective Date (the "Initial Term"), unless earlier terminated in accordance with the provisions in Section 10. Thereafter, the Parties will negotiate a new agreement prior to the summer of 2025.

#### 10. Termination

- **10.01. Termination for Breach.** Either Party may terminate this Agreement at any time in the event of a breach by the other Party of a material covenant, commitment or obligation under this Agreement that remains uncured: (i) in the event of a monetary breach, 10 calendar days following written notice thereof; and (ii) in the event of a non-monetary breach after 30 days following written notice thereof. Such termination shall be effective immediately and automatically upon the expiration of the applicable notice period, without further notice or action by either Party. Termination shall be in addition to any other remedies that may be available to the non-breaching Party.
- **10.02. Termination for Bankruptcy, Insolvency or Financial Insecurity.** Either Party may terminate this Agreement immediately at its option upon written notice if the other Party: (i) becomes or is declared insolvent or bankrupt; (ii) is the subject of a voluntary or involuntary bankruptcy or other proceeding related to its liquidation or solvency, which proceeding is not dismissed within 90 calendar days after its filing; (iii) ceases to do business in the normal course; (iv) Customer does not receive reasonably adequate funding from the state to meet its basic expenditures; or iv) makes an assignment for the benefit of creditors. This Agreement shall terminate immediately and automatically upon any determination by a court of competent jurisdiction that either Party is excused or prohibited from performing in full all obligations hereunder, including, without limitation, rejection of this Agreement pursuant to 11 U.S.C. §365).

#### 10.03. Omitted.

**10.04. Obligations upon Termination.** Termination of this Agreement for any reason shall not discharge either Party's liability for obligations incurred hereunder and amounts unpaid at the time of such termination. Customer shall pay Service Provider for all Services rendered prior to the effective date of termination. Upon termination, each Party shall return the other Party's Confidential Information that is in its possession at the time of termination. Upon

the termination of the Agreement, the Customer shall promptly return to Service Provider any equipment, materials or other property of the Service Provider relating to the terminated Services which are in Customer's possession or control.

- **11. Non-Solicitation.** During the term of this Agreement and for one year following the expiration or termination date of the Agreement, each Party agrees not to directly solicit or induce any person who performs Services hereunder to leave the employ of the other Party. The Parties are not prohibited from responding to or hiring the other's employees who inquire about employment on their own accord or in response to a public advertisement or employment solicitation in general.
- 12. Relationship of the Parties. The relationship of the Parties hereto is that of independent contractors. Nothing in this Agreement, and no course of dealing between the Parties, shall be construed to create or imply an employment or agency relationship or a partnership or joint venture relationship between the Parties or between one Party and the other Party's employees or agents. Each of the Parties is an independent contractor and neither Party has the authority to bind or contract any obligation in the name of or on account of the other Party or to incur any liability or make any statements, representations, warranties or commitments on behalf of the other Party, or otherwise act on behalf of the other. Each Party shall be solely responsible for payment of the salaries of its employees and personnel (including withholding of income taxes and social security), workers' compensation, and all other employment benefits.
- 13. Force Majeure. Neither Party shall be liable hereunder for any failure or delay in the performance of its obligations under this Agreement, except for the payment of money, which is not excused, if such failure or delay is on account of causes beyond its reasonable control, including civil commotion, war, fires, floods, accident, earthquakes, inclement weather, telecommunications line failures, electrical outages, network failures, governmental regulations or controls, casualty, strikes or labor disputes, terrorism, acts of God, or other similar or different occurrences beyond the reasonable control of the Party so defaulting or delaying in the performance of this Agreement, for so long as such force majeure event is in effect. Each Party shall use reasonable efforts to notify the other Party of the occurrence of such an event within five (5) business days of its occurrence. Customer shall not be excused from any payment due or accruing under this Agreement, nor should service privider be excused from performing it's obligations regardless of any cause contained in this Paragraph 13, so long as Customer is still receiving adequate funding.
- **14. Governing Law and Venue.** This Agreement will be governed by and interpreted in accordance with the laws of the State of Texas, without giving effect to the principles of conflicts of law of such state. The UN Convention on Contracts for the International Sale of Goods shall not apply to this Agreement. The Parties hereby agree that any action arising out of this Agreement will be brought solely in any state or federal court located in Tarrant County, Texas. Both Parties hereby submit to the exclusive jurisdiction and venue of any such court.
- **15. Attorney's Fees.** If either Party incurs any legal fees associated with the enforcement of this Agreement or any rights under this Agreement, the prevailing Party shall be entitled to recover its reasonable attorney's fees and any court, arbitration, mediation, or other litigation expenses from the other Party.
- **16. Collection Expenses.** If Service Provider incurs any costs, expenses, or fees, including reasonable attorney's fees and professional collection services fees, in connection with the collection or payment of any amounts due it under this Agreement, Customer agrees to reimburse Service Provider for all such costs, expenses and fees.
- Assignment. The Service Provider may subcontract its obligations and rights to a third-party.
- **18. Severability.** If any provision or portion of this Agreement shall be rendered by applicable law or held by a court of competent jurisdiction to be illegal, invalid, or unenforceable, the remaining provisions or portions shall remain in full force and effect.
- **19. Headings; Construction.** The headings/captions appearing in this Agreement have been inserted for the purposes of convenience and ready reference, and do not purport to and shall not be deemed to define, limit or extend the scope or intent of the provisions to which they appertain. This Agreement is the result of negotiations between the Parties and their counsel. Accordingly, this Agreement shall not be construed more strongly against

either Party regardless of which Party is more responsible for its preparation, and any ambiguity that might exist herein shall not be construed against the drafting Party.

- **20. Survival.** Each term and provision of this Agreement that should by its sense and context survive any termination or expiration of this Agreement, shall so survive regardless of the cause and even if resulting from the material breach of either Party to this Agreement.
- **21. Rights Cumulative.** The rights and remedies of the Parties herein provided shall be cumulative and not exclusive of any rights or remedies provided by law or equity.
- **22. Counterparts.** This Agreement may be executed in one or more counterparts, each of which will be deemed to be an original, but all of which together will constitute one and the same instrument, without necessity of production of the others. An executed signature page delivered via facsimile transmission or electronic signature shall be deemed as effective as an original executed signature page.
- **23. Authorized Signatories.** It is agreed and warranted by the Parties that the individuals singing this Agreement on behalf of the respective Parties are authorized to execute such an agreement. No further proof of authorization shall be required.
- **24. Notices.** All notices or other communications required under this Agreement shall be in writing and shall be deemed effective when received and made in writing by either (i) hand delivery, (ii) registered mail, (iii) certified mail, return receipt requested, or (iv) overnight mail, addressed to the Party to be notified at the following address or to such other address as such Party shall specify by like notice to the addresses in in the Preamble of this Agreement.
- **25. Waiver.** No waiver of any term or right in this Agreement shall be effective unless in writing, signed by an authorized representative of the waiving Party. The failure of either Party to enforce any provision of this Agreement shall not be construed as a waiver or modification of such provision, or impairment of its right to enforce such provision or any other provision of this Agreement thereafter.
- **26. Entire Agreement; Modification.** This Agreement, and any addendum(s) and/or exhibit(s) attached, is the entire Agreement between the Parties with respect to the subject matter hereof and supersedes any prior agreement or communications between the Parties, whether written, oral, electronic or otherwise. No change, modification, amendment, or addition of or to this Agreement or any part thereof shall be valid unless in writing and signed by authorized representatives of the Parties. Each Party hereto has received independent legal advice regarding this Agreement and their respective rights and obligations set forth herein. The Parties acknowledge and agree that they are not relying upon any representations or statements made by the other Party or the other Party's employees, agents, representatives or attorneys regarding this Agreement, except to the extent such representations are expressly set forth in this Agreement. This Agreement contains one (1) addendum and no exhibits.

In witness whereof, the Parties hereto have executed this Services Agreement on the date set forth below.

# CUSTOMER By: \_\_\_\_\_\_ Name: \_\_\_\_\_ Title: \_\_\_\_\_ Date: \_\_\_\_\_ SERVICE PROVIDER By: \_\_\_\_\_ Name: \_\_\_\_\_ Title: \_\_\_\_\_

national American	Education Federation, Inc.	, d/b/a International Leadersh	ip of Texas - Augu	st 27, 2020 Special Board	d Meeting - Agenda - T	hursday August 27, 20	)20 at 6:18
						Page 7	

Date:
-------

#### **OUTLINE OF TIER ONE RESPONSIBILITIES, AND PROCEDURES**

- 1. Tier One shall provide contracted Licensed and Commissioned Peace Officers who will be assigned to Campuses. Each of them shall be duly sworn and:
  - In possession of and consistently maintains current Texas Commission on Law Enforcement (TCOLE) certification.
  - In "good standing" with his/her employing police agency.
  - Not the subject of a current or imminent administrative internal investigation involving a significant violation of a policy or procedure with his/her employing police agency.
  - Trained in his/her agency's active shooter response procedures.
- 2. Tier One shall coordinate its Services and implementation thereof through Customer's Director of Safety and Security ("Director"). Any issues regarding the implementation of the Services shall be addressed with the Director. All reports, MOUs, and practices, procedures and directives given to the Licensed and Commissioned Peace Officers shall be provided to the Director as soon as reasonably possible.
- 3. TIER ONE shall attempt to ensure, as much as reasonably possible, that the same Licensed and Commissioned Peace Officer is assigned to the Campus on a dayto-day basis, contingent upon officers' work schedules. However, consistency in assignment is not necessary for a Licensed and Commissioned Peace Officer assigned to traffic control duties.
- 4. Licensed and Commissioned Peace Officers will comply with their employing police agency's policies and procedures, especially regarding detentions, arrests, and the use of force.'
- 5. Licensed and Commissioned Peace Officers assigned to Campus security will work a 10-hour shift, with the shift hours determined by the Campus schedule in consultation with the Campus principal. Within the 10-hour work shift, Licensed and Commissioned Peace Officers may take a 30-minute lunch break, while remaining on the school Campus depending upon Campus activities. While on the lunch break, Licensed and Commissioned Peace Officers remain responsible for responding to Campus activities that would require their involvement.
- 6. Licensed and Commissioned Peace Officers assigned to Campus security may be required to assist with traffic direction duties within their 10-hour work shift. If assisting with traffic direction, the Licensed and Commissioned Peace Officer assigned to Campus security shall be expected to respond as soon as reasonably possible back to the Campus if his/her assistance is needed for an incident.
- 7. Licensed and Commissioned Peace Officers may be requested to provide Services at school activities outside the normal school schedule. If so, Licensed and Commissioned and Commissioned Peace Officers shall be compensated in

- accordance with the hourly rates described in the Fee Schedule on the addendum for the duty type listed.
- Licensed and Commissioned Peace Officers assigned to traffic direction shall comply with their employing agency's policies and procedures as well as state law regarding traffic direction.
- 9. Licensed and Commissioned Peace Officers shall wear their agency-approved uniform in compliance with their police agency standards and state law.
- 10. Licensed and Commissioned Peace Officers shall on a reasonably appropriate basis maintain high visibility by patrolling and monitoring hallways, stairwells, restrooms, outside facilities and recreational areas, and all public or unsupervised areas of the Campus to ensure the safety and security of students, staff, and visitors.
- 11. Licensed and Commissioned Peace Officers may detain and/or arrest any individual, including students, staff, or visitors for a violation of criminal law, consistent with their training and the policy of their employing police agency and state law. The Campus principal shall be notified in advance, or as soon as reasonably possible, of any detention and/or arrest made by a Licensed and Commissioned Peace Officer on the school Campus.
- 12. Licensed and Commissioned Peace Officers shall ensure the security of the facility by making reasonably appropriate checks of the locked condition of the access points to the facility.
- 13. Licensed and Commissioned Peace Officers shall monitor the Campus security cameras on a reasonably appropriate basis to detect and respond to potentially suspicious or unusual circumstances.
- 14. Licensed and Commissioned Peace Officers shall monitor the movement and traffic flow of students through the hallways as reasonably appropriate to assist students in arriving on time for classes or to assigned locations. They will also monitor the lunch room as reasonably requested by the Campus Principal.
- 15. Licensed and Commissioned Peace Officers shall assist, when directed to do so, Campus administrative staff with the control and/or removal of disruptive or violent students utilizing verbal commands and/or reasonably appropriate force based upon factors including but not limited to the age of the student, the physical size of the student, and/or the nature of the behavior of the student. Licensed and Commissioned Peace Officers are not responsible for the administration or enforcement of student disciplinary actions.
- 16. Licensed and Commissioned Peace Officers shall provide reasonably appropriate assistance to the Campus staff, other law enforcement personnel, and/or other

- emergency first responders in responding to and/or handling emergency situations.
- 17. Licensed and Commissioned Peace Officers shall provide reasonable assistance to Campus visitors including directions and obtaining appropriate identification from visitors.
- 18. Licensed and Commissioned Peace Officers shall deny entry into the Campus to unauthorized visitors and shall direct them to exit the Campus facility and grounds.
- 19. Licensed and Commissioned Peace Officers shall ensure that unauthorized persons are not loitering on the Campus grounds.
- 20. Licensed and Commissioned Peace Officers shall notify the appropriate law enforcement agency for assistance if an incident occurs on Campus that is either beyond the scope of their employment as a security agent for the Campus or that exceeds their ability to respond appropriately.
- 21. Licensed and Commissioned Peace Officers shall offer recommendations to the Assistant Principal responsible for Campus security regarding any deficiencies in security issues that involve any potential dangerous or unusual circumstances.
- 22. Licensed and Commissioned Peace Officers shall make appropriate notifications, consistent with their sworn duties and state law, of evidence of substance abuse, child abuse, child neglect, severe medical conditions, or potential suicidal behavior; or individuals exhibiting objective physical symptoms of being under the influence of an alcoholic beverage or any controlled substance.
- 23. Licensed and Commissioned Peace Officers are not responsible for facility maintenance. However, Licensed and Commissioned Peace Officers may at their own discretion provide reasonable assistance to the Campus staff if providing such assistance does not interfere with their primary security duties.
- 24. Licensed and Commissioned Peace Officers shall ensure the confidentiality of Campus records and information regarding students and staff which the Licensed and Commissioned Peace Officer may come into possession of in the performance of its duties, and shall comply with legal constraints regarding the dissemination of that information.
- 25. Licensed and Commissioned Peace Officers should work with the Campus administration, especially the Campus principal or assistant principal(s), in a cooperative manner to develop pro-active strategies to prevent and/or mitigate potentially dangerous situation on the Campus.
- 26. Although the Licensed and Commissioned Peace Officers are not performing all the duties normally assigned to a trained School Resource Officer, specifically

- mentoring and counseling, they should as much as reasonably possible strive to serve as a positive role model for student behavior and attitude.
- 27. Licensed and Commissioned Peace Officers shall be aware that any dispute that may arise between the Campus staff, students, or visitors regarding a Licensed and Commissioned Peace Officer's attitude or behavior will be resolved with consultation among Contractor's administration, the Campus administration, and a member or members of the District-level administrative staff appointed by the Superintendent. Resolution of the dispute may include removal of the Licensed and Commissioned Peace Officer from his/her assignment on the Campus at the sole discretion of the Superintendent or his/her designee. Licensed and Commissioned Peace Officers are not guaranteed a position on the Campus.
- 28. The Services shall be rendered in a professional manner consistent with quality security services, and in a manner that is courteous and polite to the public to the extent possible under the circumstances. The Services shall be rendered in accordance with applicable laws, rules and regulations.

# Addendum A to Services Agreement Between International Leadership of Texas and Tier One Holdings, LLC

(ADDENDUM TO TIER ONE SERVICES AGREEMENT DATED AUGUST 12<sup>TH</sup>, 2020)

#### COSTS:

- Each TCOLE certified police officer will cost \$89,900 per school year paid monthly annualized over 12 months.
- Each year will have a 3% increase in this cost unless the contract is written for a five-year term. This was discussed and a five-year term was decided upon which will save the district approximately \$491,042 over the term of the five-year contract. The 3% adjustment will be made only during the third year of the five year contract for each TCOLE certified police officer that increases the cost to \$92,597 per school year paid monthly annualized over 12 months.
- Extra-Curricular hours will be billed in addition at a rate of \$40 per hour while most departments charge \$50-\$60 per hour.

#### BILLING:

- The customer will receive two separate invoices per month and will receive these invoices on or about the 15<sup>th</sup> of each month. The customer will pay the invoice no later than the 2<sup>nd</sup> of each month for the current month to be worked, essentially paying in advance for the given month. This is discussed in further detail in the service agreement. These invoices will be prorated during the phase in process discussed below.
  - Invoice #1 will be for police services which is essentially the total of \$89,900 per officer divided by 12 months until the third year as discussed above when the cost increases to \$92,597.
  - Invoice #2 will be for extra curricular coverage at \$40 per hour. Each campus that has this coverage for any given month will have its own line item in the invoice with the number of hours x \$40.

#### 3. PHASED APPROACH:

- It is estimated that there are 20 campuses that need coverage for 2020-2021 school year.
- In the short term, the two campuses at College Station will be exempted from this
  contract due to College Station Police off duty officer coverage. This can be easily be
  added to the contract at any time. The bill rate will be prorated as determined by the
  starting month and end month of the given school year in which these two campuses are
  added.
- The estimated additional 18 campuses will be contracted by Tier One and designated law enforcement agency and staffed through as phased in process with a timeline of 60-90 or sooner when possible.
  - Tier One would like to do a phased in approach which equates to attempting to find high quality officers at an estimated a rate of five per month. If Tier One is able to contract more than five, we will notify ILT to discuss placement.
  - Tier One will provide ILT a 30-day notice that an officer is available and ready to start. This notice allows ILT's current vendor, ASG Security, a transition period for their employees to be reallocated or provided a potential opportunity with Tier One
  - If ASG Security fails to maintain staffing during the transition phase, Tier One may fill the vacant police officer position of the campuses in question sooner than

Initials		

- 30 days based upon officer availability. The intent is to cover campuses quickly should ASG Security be unable to staff effectively due to non-renewal of contract.
- Note as stated previously that the bill rate will be prorated as determined by the starting month of the given school year in which each officer is added.
- ILT plans on adding an estimated 2 campuses per year until they get to 27 campuses. Tier One will accommodate as many campuses as ILT adds with proper communication and reasonable expectations in regard to length of time it takes to fill those positions. The cost for these police officers will be the same as all others in the contract and addendum.

CUSTOMER
Ву:
Name:
Title:
Date:
SERVICE PROVIDER
SERVICE PROVIDER  By:
Ву:
By:

Initials		
----------	--	--

# **Cover Sheet**

# CONSIDER/ACT ON CONTRACT WITH KEY CONSTRUCTION FOR PORTABLES INSTALLATION AT ILTEXAS COLLEGE STATION K-8

Section: IV. Board Items for Discussion/Action

Item: B. CONSIDER/ACT ON CONTRACT WITH KEY CONSTRUCTION FOR

PORTABLES INSTALLATION AT ILTEXAS COLLEGE STATION K-8

Purpose: Vote

Submitted by:

Related Material: ILT CS Portable Classrooms CM Contract 8-27-20.pdf

# Standard Form of Agreement Between Owner and Construction Manager as Constructor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price

**AGREEMENT** made as of the 25<sup>th</sup> day of August in the year 2020 (In words, indicate day, month and year.)

**BETWEEN** the Owner:

(Name, legal status and address)

International Leadership of Texas 1820 N. Glenville Dr. #100 Richardson, TX 75081

and the Construction Manager: (Name, legal status and address)

Key Construction, Inc. 741 W. 2<sup>nd</sup>, Wichita, KS 67203

for the following Project: (Name and address or location)

International Leadership of Texas Charter Schools Portable Classrooms 3610 Longmire Drive, College Station, TX 77840

The Architect:

(Name, legal status and address)

Pivot North Architecture 1101 W. Grove St., Boise, ID 83702

The Owner's Designated Representative: (Name, address and other information)

Charles Klein 1820 N. Glenville Dr., #100 Richardson, TX 75081

The Construction Manager's Designated Representative: (Name, address and other information)

Brent Weckar 741 W. 2<sup>nd</sup>., Wichita, KS 67203

The Architect's Designated Representative: (Name, address and other information)

Jared Schmidt 1101 W. Grove St., Boise, ID 83702

The Owner and Construction Manager agree as follows.

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

AIA Document A201™-2007, General Conditions of the Contract for Construction, is adopted in this document by reference. Do not use with other general conditions unless this document is modified.

Init.

**User Notes:** 

AIA Document A133™ - 2009 (formerly A121™CMc - 2003). Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:05:37 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

(1651987766)



#### TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 CONSTRUCTION MANAGER'S RESPONSIBILITIES
- 3 OWNER'S RESPONSIBILITIES
- 4 COMPENSATION AND PAYMENTS FOR PRECONSTRUCTION PHASE SERVICES
- 5 COMPENSATION FOR CONSTRUCTION PHASE SERVICES
- 6 COST OF THE WORK FOR CONSTRUCTION PHASE
- 7 PAYMENTS FOR CONSTRUCTION PHASE SERVICES
- 8 INSURANCE AND BONDS
- 9 DISPUTE RESOLUTION
- 10 TERMINATION OR SUSPENSION
- 11 MISCELLANEOUS PROVISIONS
- 12 SCOPE OF THE AGREEMENT

#### EXHIBIT A GUARANTEED MAXIMUM PRICE AMENDMENT

#### ARTICLE 1 GENERAL PROVISIONS

#### § 1.1 The Contract Documents

The Contract Documents consist of this Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to the execution of this Agreement, other documents listed in this Agreement, and Modifications issued after execution of this Agreement, all of which form the Contract and are as fully a part of the Contract as if attached to this Agreement or repeated herein. Upon the Owner's acceptance of the Construction Manager's Guaranteed Maximum Price proposal, the Contract Documents will also include the documents described in Section 2.2.3 and identified in the Guaranteed Maximum Price Amendment and revisions prepared by the Architect and furnished by the Owner as described in Section 2.2.8. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. If anything in the other Contract Documents, other than a Modification, is inconsistent with this Agreement, this Agreement shall govern.

#### § 1.2 Relationship of the Parties

The Construction Manager accepts the relationship of trust and confidence established by this Agreement and covenants with the Owner to cooperate with the Architect and exercise the Construction Manager's skill and judgment in furthering the interests of the Owner; to furnish efficient construction administration, management services and supervision; to furnish at all times an adequate supply of workers and materials; and to perform the Work in an expeditious and economical manner consistent with the Owner's interests. The Owner agrees to furnish or approve, in a timely manner, information required by the Construction Manager and to make payments to the Construction Manager in accordance with the requirements of the Contract Documents.

#### § 1.3 General Conditions

For the Preconstruction Phase, AIA Document A201<sup>TM</sup>–2007, General Conditions of the Contract for Construction, shall apply only as specifically provided in this Agreement. For the Construction Phase, the general conditions of the contract shall be as set forth in A201–2007, which document is incorporated herein by reference. The term "Contractor" as used in A201–2007 shall mean the Construction Manager. All references to the A201<sup>TM</sup>–2007, General Conditions of the Contract for Construction shall mean the A201<sup>TM</sup>–2017, General Conditions of the Contract for Construction as amended.

#### ARTICLE 2 CONSTRUCTION MANAGER'S RESPONSIBILITIES

The Construction Manager's Preconstruction Phase responsibilities are set forth in Sections 2.1 and 2.2. The Construction Manager's Construction Phase responsibilities are set forth in Section 2.3. The Owner and Construction Manager may agree, in consultation with the Architect, for the Construction Phase to commence prior to completion of the Preconstruction Phase, in which case, both phases will proceed concurrently. The Construction Manager shall identify a representative authorized to act on behalf of the Construction Manager with respect to the Project.

#### § 2.1 Preconstruction Phase

**§ 2.1.1** The Construction Manager shall provide a preliminary evaluation of the Owner's program, schedule and construction budget requirements, each in terms of the other.

#### § 2.1.2 Consultation

At least monthly, the Construction Manager shall schedule and conduct meetings with the Architect and Owner to discuss such matters as procedures, progress, coordination, and scheduling of the Work. The Construction Manager shall advise the Owner and the Architect on proposed site use and improvements, selection of materials, and building systems and equipment. The Construction Manager shall also provide recommendations consistent with the Project requirements to the Owner and Architect on constructability; availability of materials and labor; time requirements for procurement, installation and construction; and factors related to construction cost including, but not limited to, costs of alternative designs or materials, preliminary budgets, life-cycle data, and possible cost reductions.

§ 2.1.3 When Project requirements in Section 3.1.1 have been sufficiently identified, the Construction Manager shall prepare and periodically update a Project schedule for the Architect's review and the Owner's acceptance. The Construction Manager shall obtain the Architect's approval for the portion of the Project schedule relating to the performance of the Architect's services. The Project schedule shall coordinate and integrate the Construction Manager's services, the Architect's services, other Owner consultants' services, and the Owner's responsibilities and identify items that could affect the Project's timely completion. The updated Project schedule shall include the following: submission of the Guaranteed Maximum Price proposal; components of the Work; times of commencement and completion required of each Subcontractor; ordering and delivery of products, including those that must be ordered well in advance of construction; and the occupancy requirements of the Owner.

#### § 2.1.4 Phased Construction

The Construction Manager shall provide recommendations with regard to accelerated or fast-track scheduling, procurement, or phased construction. The Construction Manager shall take into consideration cost reductions, cost information, constructability, provisions for temporary facilities and procurement and construction scheduling issues.

#### § 2.1.5 Preliminary Cost Estimates

§ 2.1.5.1 Based on the preliminary design and other design criteria prepared by the Architect, the Construction Manager shall prepare preliminary estimates of the Cost of the Work or the cost of program requirements using area, volume or similar conceptual estimating techniques for the Architect's review and Owner's approval. If the Architect or Construction Manager suggests alternative materials and systems, the Construction Manager shall provide cost evaluations of those alternative materials and systems.

§ 2.1.5.2 As the Architect progresses with the preparation of the Schematic Design, Design Development and Construction Documents, the Construction Manager shall prepare and update, at appropriate intervals agreed to by the Owner, Construction Manager and Architect, estimates of the Cost of the Work of increasing detail and refinement and allowing for the further development of the design until such time as the Owner and Construction Manager agree on a Guaranteed Maximum Price for the Work. Such estimates shall be provided for the Architect's review and the Owner's approval. The Construction Manager, as soon as feasible, shall inform the Owner and Architect when estimates of the Cost of the Work exceed the latest approved Project budget and make recommendations for corrective action.

#### § 2.1.6 Subcontractors and Suppliers

The Construction Manager shall develop bidders' interest in the Project.

§ 2.1.7 The Construction Manager shall prepare, for the Architect's review and the Owner's acceptance, a procurement schedule for items that must be ordered well in advance of construction. The Construction Manager shall expedite and coordinate the ordering and delivery of materials that must be ordered well in advance of construction. If the Owner agrees to procure any items prior to the establishment of the Guaranteed Maximum Price, the Owner shall procure the items on terms and conditions acceptable to the Construction Manager. Upon the establishment of the Guaranteed Maximum Price, the Owner shall assign all contracts for these items to the Construction Manager and the Construction Manager shall thereafter accept responsibility for them.

#### § 2.1.8 Extent of Responsibility

The Construction Manager shall exercise reasonable care in preparing schedules and estimates. The Construction Manager, however, does not warrant or guarantee estimates and schedules except as may be included as part of the Guaranteed Maximum Price. The Construction Manager is not required to ascertain that the Drawings and Specifications are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Construction Manager shall promptly report to the Architect and Owner any nonconformity discovered by or made known to the Construction Manager as a request for information in such form as the Architect may require.

#### § 2.1.9 Notices and Compliance with Laws

The Construction Manager shall comply with applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to its performance under this Contract, and with equal employment opportunity programs, and other programs as may be required by governmental and quasi governmental authorities for inclusion in the Contract Documents.

#### § 2.2 Guaranteed Maximum Price Proposal and Contract Time

§ 2.2.1 When the Drawings and Specifications for the construction of the entire Work of the Project, including all agreed-upon pre construction packages (if any), are sufficiently complete, and all subcontractor bids or proposals have been received and evaluated, the Construction Manager shall propose a maximum sum of the Cost of the Work (the "Guaranteed Maximum Price Proposal"), including the Construction Manager's General Conditions Costs as provided herein, the contingencies described in Section 2.2.4, and the Construction Manager's Fee that, upon acceptance by the Owner, the Construction Manager will guarantee not to exceed, subject only to those additions and deductions by changes in the Work as provided in the Contract Documents. Change Orders shall be allowed for unforeseen or changed conditions, including, but not limited to, subsurface conditions not discovered or reported by the Owner's geotechnical engineer, changes in conditions required by any permitting or regulatory agencies, scope changes requested by Owner, or concealed conditions or as otherwise allowed under § 3.7.4 of the A201<sup>TM</sup>-2017 General Conditions, except to the extent such conditions were discovered or should have been discovered by the Construction Manager using reasonable means prior to fixing the Guaranteed Maximum Price. Construction Manager shall promptly notify the Owner of any anticipated or actual needed Change Order.

#### § 2.2.2

§ 2.2.3 The Construction Manager shall include with the Guaranteed Maximum Price Proposal a written statement of its basis, which shall include the following:

- .1 A list of the Drawings and Specifications, including all Addenda thereto, and the Conditions of the Contract:
- .2 A list of the clarifications and assumptions made by the Construction Manager in the preparation of the Guaranteed Maximum Price proposal to supplement the information provided by the Owner and contained in the Drawings and Specifications;
- .3 The proposed Guaranteed Maximum Price, including the items set forth in Section 2.2.3.6 below and as otherwise approved by Owner;
- .4 The Date of Substantial Completion upon which the proposed Guaranteed Maximum Price is based, together with such Interim Completion Dates made a part thereof, and a schedule of the Construction Documents issuance dates upon which the date of Substantial Completion is based;
- .5 A date by which the Owner must accept the Guaranteed Maximum Price.
- .6 The Guaranteed Maximum Price shall be the sum of the following:
  - a. the sum of all accepted subcontractor or trade bids or proposals, including, if acceptable to Owner, allowances as allowed under § 2.2.10, below, compensation for any preconstruction services, and proposals from the Construction Manager to self perform portions of the work;

- b. the amount of the Construction Manager's General Conditions Costs;
- c. the Construction Manager's Contingency in an amount stated in the GMP Addendum
- d. the Owner's Contingency in an amount stated in the GMP Addendum
- e. the Construction Manager's Fee, which may not be calculated or included on any work performed by the Construction Manager's own forces; and
- f. Allowances, if any, as approved by the Owner.
- § 2.2.4 In preparing the Construction Manager's Guaranteed Maximum Price Proposal, the Construction Manager shall include its contingency for the Construction Manager's exclusive use to cover those costs considered reimbursable as the Cost of the Work but not included in a Change Order. Notwithstanding the foregoing, overruns of actual Cost of the Work beyond Construction Manager's Contingency shall not affect the Guaranteed Maximum Price, and shall not be the basis for a Change Order, unless otherwise agreed to in writing by the Owner.
- § 2.2.5 The Construction Manager shall meet with the Owner and Architect to review the Guaranteed Maximum Price proposal. In the event that the Owner and Architect discover any inconsistencies or inaccuracies in the information presented, they shall promptly notify the Construction Manager, who shall make appropriate adjustments to the Guaranteed Maximum Price proposal, its basis, or both.
- § 2.2.6 If the Owner notifies the Construction Manager that the Owner has accepted the Guaranteed Maximum Price proposal in writing before the date specified in the Guaranteed Maximum Price proposal, the Guaranteed Maximum Price proposal shall be deemed effective without further acceptance from the Construction Manager. Following acceptance of a Guaranteed Maximum Price, the Owner and Construction Manager shall execute the Guaranteed Maximum Price Amendment amending this Agreement, a copy of which the Owner shall provide to the Architect. The Guaranteed Maximum Price Amendment shall set forth the agreed upon Guaranteed Maximum Price with the information and assumptions upon which it is based.
- § 2.2.7 The Construction Manager shall not incur any cost to be reimbursed as part of the Cost of the Work prior to the commencement of the Construction Phase, unless the Owner provides prior written authorization for such costs.
- § 2.2.8 The Owner shall authorize the Architect to provide the revisions to the Drawings and Specifications to incorporate the agreed upon assumptions and clarifications contained in the Guaranteed Maximum Price Amendment. The Owner shall promptly furnish those revised Drawings and Specifications to the Construction Manager as they are revised. The Construction Manager shall notify the Owner and Architect of any inconsistencies between the Guaranteed Maximum Price Amendment and the revised Drawings and Specifications.
- § 2.2.9 The Construction Manager shall include in the Guaranteed Maximum Price all sales, consumer, use and similar taxes for the Work provided by the Construction Manager that are legally enacted, whether or not yet effective, at the time the Guaranteed Maximum Price Amendment is executed. Since the Owner is a tax exempt entity, no sales tax or other tax from which the Owner is exempt shall be included in the proposal or any portion of the Cost of the Work, except where required by law.
- § 2.2.10 The Construction Manager and the Owner agree and acknowledge that the cost of certain portions of the Work may be incapable of exact determination at the time that the Guaranteed Maximum Price is established and accepted by the Owner. In such event, the Construction Manager and the Owner will establish reasonable estimates of these costs based upon availability of information for such portions of the Work. Such estimates are herein called "Allowances" and, if established by the parties as provided herein, will be shown in the Schedule of Values and clearly identified therein as "Allowances" and subject to Section 3.8.2 of A201<sup>TM</sup>-2017. Allowances, if any, are to be enumerated on a List of Allowances set out at the time the Guaranteed Maximum Price is established.

#### § 2.3 Construction Phase

§ 2.3.1 General

**User Notes:** 

- § 2.3.1.1 For purposes of Section 8.1.2 of A201–2007, the date of commencement of the Work shall mean the date of commencement of the Construction Phase.
- § 2.3.1.2 The Construction Phase shall commence upon the later to occur of the Owner's acceptance of the Construction Manager's Guaranteed Maximum Price proposal or the Owner's issuance of a Notice to Proceed, or the Construction Manager's receipt of a full building permit and commitment of financing.

Init.

AIA Document A133™ – 2009 (formerly A121™CMc – 2003). Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:05:37 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

#### § 2.3.2 Administration

- § 2.3.2.1 Those portions of the Work that the Construction Manager does not customarily perform with the Construction Manager's own personnel shall be performed under subcontracts or by other appropriate agreements with the Construction Manager.
- § 2.3.2.2 If the Guaranteed Maximum Price has been established and when a specific bidder (1) is recommended to the Owner by the Construction Manager, (2) is qualified to perform that portion of the Work, and (3) is qualified to perform that portion of the Work, and (4) has submitted a bid that conforms to the requirements of the Contract Documents without reservations or exceptions, but the Owner requires that another bid be accepted, then the Construction Manager may require that a Change Order be issued to adjust the Contract Time and the Guaranteed Maximum Price by the difference between the bid of the person or entity recommended to the Owner by the Construction Manager and the amount and time requirement of the subcontract or other agreement actually signed with the person or entity designated by the Owner.
- § 2.3.2.3 Subcontracts or other agreements shall conform to the applicable payment provisions of this Agreement, and shall not be awarded on the basis of cost plus a fee without the prior consent of the Owner. If the Subcontract is awarded on a cost plus fee basis, the Construction Manager shall provide in the Subcontract for the Owner to receive the same audit rights with regard to the Subcontractor as the Owner receives with regard to the Construction Manager in Section 6.11 below.
- § 2.3.2.4 If the Construction Manager recommends a specific bidder that may be considered a "related party" according to Section 6.10, then the Construction Manager shall promptly notify the Owner in writing of such relationship and notify the Owner of the specific nature of the contemplated transaction, according to Section 6.10.2.
- § 2.3.2.5 The Construction Manager shall schedule and conduct meetings to discuss such matters as procedures, progress, coordination, scheduling, and status of the Work. The Construction Manager shall prepare and promptly distribute minutes to the Owner and Architect.
- § 2.3.2.6 Upon the execution of the Guaranteed Maximum Price Amendment, the Construction Manager shall prepare and submit to the Owner and Architect a construction schedule for the Work and submittal schedule in accordance with Section 3.10 of A201–2007.
- § 2.3.2.7 The Construction Manager shall record the progress of the Project. On a monthly basis, or otherwise as agreed to by the Owner, the Construction Manager shall submit written progress reports to the Owner and Architect, showing percentages of completion and other information required by the Owner. The Construction Manager shall also keep, and make available to the Owner and Architect, a daily log containing a record for each day of weather, portions of the Work in progress, number of workers on site, identification of equipment on site, problems that might affect progress of the work, accidents, injuries, work in progress, work completed, and by whom and other information required by the Owner. Such log entries shall be transmitted to Owner on at least a monthly basis, or as otherwise requested by Owner.
- § 2.3.2.8 The Construction Manager shall develop a system of cost control for the Work, including regular monitoring of actual costs for activities in progress and estimates for uncompleted tasks and proposed changes. The Construction Manager shall identify variances between actual and estimated costs and report the variances to the Owner and Architect and shall provide this information in its monthly reports to the Owner and Architect, in accordance with Section 2.3.2.7 above.

#### § 2.4 Professional Services

Section 3.12.10 of A201–2007 shall apply to both the Preconstruction and Construction Phases.

#### § 2.5 Hazardous Materials

**User Notes:** 

Section 10.3 of A201–2007 shall apply to both the Preconstruction and Construction Phases.

24 of 99

#### ARTICLE 3 OWNER'S RESPONSIBILITIES

#### § 3.1 Information and Services Required of the Owner

- § 3.1.1 The Owner shall provide information with reasonable promptness, regarding requirements for and limitations on the Project, including a written program which shall set forth the Owner's objectives, constraints, and criteria, including schedule, space requirements and relationships, flexibility and expandability, special equipment, systems, sustainability and site requirements.
- § 3.1.2 Prior to the execution of the Guaranteed Maximum Price Amendment, the Construction Manager may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Construction Manager may only request such evidence if (1) the Owner fails to make payments to the Construction Manager as the Contract Documents require, (2) a change in the Work materially changes the Contract Sum, or (3) the Construction Manager identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Construction Manager and Architect.
- § 3.1.3 The Owner shall establish and periodically update the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1.1, (2) the Owner's other costs, and (3) reasonable contingencies related to all of these costs. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Construction Manager and Architect. The Owner and the Architect, in consultation with the Construction Manager, shall thereafter agree to a corresponding change in the Project's scope and quality.
- § 3.1.4 Structural and Environmental Tests, Surveys and Reports. During the Preconstruction Phase, the Owner shall furnish the following information or services with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Construction Manager's performance of the Work with reasonable promptness after receiving the Construction Manager's written request for such information or services. The Construction Manager shall be entitled to rely on the accuracy of information and services furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.
- § 3.1.4.1 The Owner shall furnish tests, inspections and reports required by law and as otherwise agreed to by the parties, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.
- § 3.1.4.2 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions and necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.
- § 3.1.4.3 The Owner, when such services are requested, shall furnish services of geotechnical engineers, which may include but are not limited to test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 3.1.4.4 During the Construction Phase, the Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Construction Manager's performance of the Work with reasonable promptness after receiving the Construction Manager's written request for such information or services.

#### § 3.2 Owner's Designated Representative

The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. All parties acknowledge that only the Board of Directors for the Owner, acting as a body corporate, has the authority to bind the Owner with respect to all matters requiring the Board's approval under School District Board Policy, including without limitation, Changes in the Work, except for minor changes in the

Work to be paid out of Owner's contingency in an amount not to exceed the amount identified in the GMP Addendum for any individual change in the aggregate amount identified in the GMP Addendum. Except as otherwise provided in Section 4.2.1 of A201<sup>TM</sup> 2017, the Architect does not have authority to bind the Owner with respect to matters requiring the Owner's approval or authorization. The term "Owner" means the Owner or the Owner's authorized representative, whom the Owner shall designate in writing, and who shall have authority to make decisions on behalf of the Owner concerning change orders to be paid out of Owner's Contingency not to exceed the amount identified in the GMP Addendum for any individual change or the amount identified in the GMP Addendum in the aggregate and extensions in the Contract Time consistent with the Contract Documents, and shall render such decisions promptly and furnish information expeditiously, so as to avoid unreasonable delay in the services or Work of the Construction Manager.

§ 3.2.1 Legal Requirements. The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.

#### § 3.3 Architect

The Owner shall retain an Architect to provide services, duties and responsibilities according to an agreement between Owner and Architect executed by such parties, including any additional services requested by the Construction Manager that are necessary for the Preconstruction and Construction Phase services under this Agreement. Such services shall be provided in accordance with time schedules agreed to by the Owner, Architect and the Construction Manager. Upon request, the Owner shall provide the Construction Manager a copy of the executed agreement between the Owner and the Architect, and any further modifications to such agreement.

#### COMPENSATION AND PAYMENTS FOR PRECONSTRUCTION PHASE SERVICES ARTICLE 4

§ 4.1 Compensation

- § 4.1.1 For the Construction Manager's Preconstruction Phase services, the Owner shall compensate the Construction Manager as follows:
- § 4.1.2 For the Construction Manager's Preconstruction Phase services described in Sections 2.1 and 2.2: (Insert amount of, or basis for, compensation and include a list of reimbursable cost items, as applicable.)

Not applicable. Preconstruction Phase compensation is included in a separate agreement.

- § 4.1.3 Preconstruction Phase services are performed under a separate agreement.
- § 4.1.4 Compensation based on Direct Personnel Expense includes the direct salaries of the Construction Manager's personnel providing Preconstruction Phase services on the Project and the Construction Manager's costs for the mandatory and customary contributions and benefits related thereto, such as employment taxes and other statutory employee benefits, insurance, sick leave, holidays, vacations, employee retirement plans and similar contributions.

#### § 4.2 Payments

- § 4.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed.
- § 4.2.2 Payments are due and payable upon presentation of the Construction Manager's invoice. Amounts unpaid thirty (30) days after the invoice date shall bear interest (Paragraphs deleted) at the rate of 6.5% per annum.

#### ARTICLE 5 COMPENSATION FOR CONSTRUCTION PHASE SERVICES

§ 5.1 For the Construction Manager's performance of the Work as described in Section 2.3, the Owner shall pay the Construction Manager the Contract Sum in current funds. The Contract Sum is the Cost of the Work as defined in Section 6.1.1 plus the Construction Manager's Fee.

#### § 5.1.1 The Construction Manager's Fee:

(State a lump sum, percentage of Cost of the Work or other provision for determining the Construction Manager's Fee.)

Ten Percent (10%)

Init.

AIA Document A133™ - 2009 (formerly A121™CMc - 2003). Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:05:37 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. **User Notes:** 

§ 5.1.2 The method of adjustment of the Construction Manager's Fee for changes in the Work:

The Construction Manager's overhead and profit attributable to increases in the Cost of the Work, as evidenced by executed Change Order(s) identifying the same, shall not exceed the percentage of the Construction Manager's Fee, and no markup shall be allowed on self-performed work.

§ 5.1.3 Limitations, if any, on a Subcontractor's overhead and profit for increases in the cost of its portion of the Work:

A Subcontractor's overhead and profit attributable to increases in the cost of its portion of the Work, as evidenced by executed Change Order(s) identifying the same, shall not exceed the rate of mark-up for the Subcontractor's base scope of Work.

§ 5.1.4 Rental rates for Construction Manager-owned equipment shall not exceed one hundred percent (100 %) of the standard rate paid at the place of the Project

#### § 5.1.5 Unit prices, if any:

(Identify and state the unit price; state the quantity limitations, if any, to which the unit price will be applicable.)

Item	Units and Limitations	Price per Unit (\$0.00)	
N/A	N/A	N/A	

#### § 5.2 Guaranteed Maximum Price

§ 5.2.1 The Construction Manager guarantees that the Contract Sum shall not exceed the Guaranteed Maximum Price set forth in the Guaranteed Maximum Price Amendment, as it is amended from time to time. To the extent the Cost of the Work exceeds the Guaranteed Maximum Price, the Construction Manager shall bear such costs in excess of the Guaranteed Maximum Price without reimbursement or additional compensation from the Owner. (Insert specific provisions if the Construction Manager is to participate in any savings.)

N/A

§ 5.2.2 The Guaranteed Maximum Price is subject to additions and deductions by Change Order as provided in the Contract Documents and the Date of Substantial Completion shall be subject to adjustment as provided in the Contract Documents.

#### § 5.3 Changes in the Work

- § 5.3.1 The Owner may, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions. The Owner shall issue such changes in writing. The Construction Manager shall be entitled to an equitable adjustment in the Contract Time as a result of changes in the Work and shall notify the Owner in advance of any anticipated extension of the Contract Time. The Architect may make minor changes in the Work as provided in Section 7.4 of AIA Document A201–2007, General Conditions of the Contract for Construction not affecting the Contract Time or the Cost of the Work. Subject to the Owner's consent for minor changes which may affect the Contract Time or the Cost of the Work, the Architect may make minor changes in the Work as provided in Section 7.4 of AIA Document A201<sup>TM</sup>-2017, General Conditions of the Contract for Construction, as amended by the Owner and provided the Construction Manager is provided an equitable adjustment to the Contract time and/or the Cost of the Work.
- § 5.3.2 Adjustments to the Guaranteed Maximum Price on account of changes in the Work subsequent to the execution of the Guaranteed Maximum Price Amendment, including changes under § 5.3.1, above, may be determined by any of the methods listed in Section 7.3.3 of AIA Document A201–2007, General Conditions of the Contract for Construction.
- § 5.3.3 In calculating adjustments to subcontracts (except those awarded with the Owner's prior consent on the basis of cost plus a fee), the terms "cost" and "fee" as used in Section 7.3.3.3 of AIA Document A201–2007 and the term "costs" as used in Section 7.3.7 of AIA Document A201–2007 shall have the meanings assigned to them in AIA

lnit.

**User Notes:** 

AIA Document A133™ – 2009 (formerly A121™CMc – 2003). Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:05:37 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

Document A201–2007 and shall not be modified by Sections 5.1 and 5.2, Sections 6.1 through 6.7, and Section 6.8 of this Agreement. Adjustments to subcontracts awarded with the Owner's prior consent on the basis of cost plus a fee shall be calculated in accordance with the terms of those subcontracts.

- § 5.3.4 In calculating adjustments to the Guaranteed Maximum Price, the terms "cost" and "costs" as used in the above-referenced provisions of AIA Document A201–2007 shall mean the Cost of the Work as defined in Sections 6.1 to 6.7 of this Agreement and the term "fee" shall mean the Construction Manager's Fee as defined in Section 5.1 of this Agreement.
- § 5.3.5 If no specific provision is made in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case of changes in the Work, or if the extent of such changes is such, in the aggregate, that application of the adjustment provisions of Section 5.1.2 will cause inequity to the Owner or Construction Manager, the Construction Manager's Fee shall be equitably adjusted on the same basis that was used to establish the Fee for the original Work, and the Guaranteed Maximum Price shall be adjusted accordingly.

#### ARTICLE 6 COST OF THE WORK FOR CONSTRUCTION PHASE

#### § 6.1 Costs to Be Reimbursed

- § 6.1.1 The term Cost of the Work shall mean costs necessarily incurred by the Construction Manager in the proper performance of the Work. Such costs shall be at rates not higher than the standard paid at the place of the Project except with prior consent of the Owner. The Cost of the Work shall include only the items set forth in Sections 6.1 through 6.7.
- § 6.1.2 Where any cost is subject to the Owner's prior approval, the Construction Manager shall obtain this approval prior to incurring the cost. The parties shall endeavor to identify any such costs prior to executing Guaranteed Maximum Price Amendment.

#### § 6.2 Labor Costs

- § 6.2.1 Wages of construction workers directly employed by the Construction Manager to perform the construction of the Work at the site per Exhibit E.
- § 6.2.2 Wages or salaries, excluding bonuses, of the Construction Manager's supervisory and administrative personnel when stationed at the site with the Owner's prior written agreement, and only to the extent that such personnel perform supervisory or administrative work directly related to the Project, or other personnel as identified in writing in the Construction Manager's Fee proposal, attached hereto as Exhibit E.
- § 6.2.3 Wages and salaries of the Construction Manager's supervisory or administrative personnel engaged at factories, workshops or on the road, in expediting the production or transportation of materials or equipment required for the Work, but only for that portion of their time required for the Work. The only personnel reimbursable hereunder are the personnel set out in writing in the Contract Documents and the Construction Manager's Proposal.
- § 6.2.4 Costs paid or incurred by the Construction Manager for taxes, insurance, contributions, assessments and benefits required by law or collective bargaining agreements and, for personnel not covered by such agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such costs are based on wages and salaries included in the Cost of the Work under Sections 6.2.1 through 6.2.3.
- § 6.2.5 Profit sharing and incentive compensation paid to anyone hired by the Construction Manager or paid to any Subcontractor or vendor, with the Owner's prior approval.

#### § 6.3 Subcontract Costs

Payments made by the Construction Manager to Subcontractors in accordance with the requirements of the subcontracts.

#### § 6.4 Costs of Materials and Equipment Incorporated in the Completed Construction

**§ 6.4.1** Costs, including transportation and storage, of materials and equipment incorporated or to be incorporated in the completed construction.

§ 6.4.2 Costs of materials described in the preceding Section 6.4.1 in excess of those actually installed to allow for reasonable waste and spoilage. Unused excess materials, if any, shall become the Owner's property at the completion of the Work or, at the Owner's option, shall be sold by the Construction Manager. Any amounts realized from such sales shall be credited to the Owner as a deduction from the Cost of the Work.

#### § 6.5 Costs of Other Materials and Equipment, Temporary Facilities and Related Items

- § 6.5.1 Costs of transportation, storage, installation, maintenance, dismantling and removal of materials, supplies, temporary facilities, machinery, equipment and hand tools not customarily owned by construction workers that are provided by the Construction Manager at the site and fully consumed in the performance of the Work. Costs of materials, supplies, temporary facilities, machinery, equipment and tools that are not fully consumed shall be based on the cost or value of the item at the time it is first used on the Project site less the value of the item when it is no longer used at the Project site. Costs for items not fully consumed by the Construction Manager shall mean fair market value.
- § 6.5.2 Rental charges for temporary facilities, machinery, equipment and hand tools not customarily owned by construction workers that are rented from unrelated third parties and provided by the Construction Manager at the site, and costs of transportation, installation, minor repairs, dismantling and removal thereof. Rates and quantities of equipment rented shall be subject to the Owner's prior written approval. The Cost of the Work described in this subsection shall include only straight rental costs and will not include any costs associated with the purchase of, or an option to purchase, the machinery or equipment. The total rental cost of any Construction Manager-owned item may not exceed the purchase price of any comparable item.
- § 6.5.3 Costs of removal of debris from the site of the Work and its proper and legal disposal.
- **§ 6.5.4** Costs of document reproductions, facsimile transmissions and long-distance telephone calls, postage and parcel delivery charges, telephone service at the site and reasonable petty cash expenses of the site office.
- § 6.5.5 That portion of the reasonable travel and subsistence expenses of the Construction Manager's supervisory or administrative personnel incurred while traveling in discharge of duties connected with the Work in accordance with the Construction Manager's current standard written policy for travel reimbursement. Such expenses, together with the method of calculating reimbursable travel and subsistence expenses incurred by employees of the Construction Manager must receive written approval in advance from the Owner.
- § 6.5.6 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the Owner's prior approval. Such costs, to be reimbursable, must comply with Section 9.3.2 of AIA Document A201<sup>TM</sup>-2017 General Conditions.

#### § 6.6 Miscellaneous Costs

- § 6.6.1 Premiums for that portion of insurance and bonds required by the Contract Documents that can be directly attributed to this Contract. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval. If the Construction Manager requires Subcontractors to furnish bonds or participate in a subcontractor default insurance program, the cost of such bonds or insurance shall be included in the Cost of the Work.
- § 6.6.2 Sales, use or similar taxes imposed by a governmental authority that are related to the Work and for which the Construction Manager is liable, except those taxes which the Owner is not required to pay under applicable law.
- § 6.6.3 Fees and assessments for the building permit and for other permits, licenses and inspections for which the Construction Manager is required by the Contract Documents to pay.
- **§ 6.6.4** Fees of laboratories for tests required by the Contract Documents, except those related to defective or nonconforming Work for which reimbursement is excluded by Section 13.5.3 of AIA Document A201–2017 or by other provisions of the Contract Documents, and which do not fall within the scope of Section 6.7.3.
- § 6.6.5 Royalties and license fees paid for the use of a particular design, process or product required by the Contract Documents; the cost of defending suits or claims for infringement of patent rights arising from such requirement of the Contract Documents; and payments made in accordance with legal judgments against the Construction Manager resulting from such suits or claims and payments of settlements made with the Owner's consent. However, such costs

**User Notes:** 

29 of 99

of legal defenses, judgments and settlements shall not be included in the calculation of the Construction Manager's Fee or subject to the Guaranteed Maximum Price. If such royalties, fees and costs are excluded by the last sentence of Section 3.17 of AIA Document A201–2007 or other provisions of the Contract Documents, then they shall not be included in the Cost of the Work.

- § 6.6.6 Costs for electronic equipment and software, directly related to the Work with the Owner's prior approval.
- § 6.6.7 Deposits lost for causes other than the Construction Manager's negligence or failure to fulfill a specific responsibility in the Contract Documents.
- § 6.6.8 Legal, mediation and arbitration costs, including attorneys' fees, other than those arising from disputes between the Owner and Construction Manager, reasonably incurred by the Construction Manager after the execution of this Agreement in the performance of the Work and with the Owner's prior approval, which shall not be unreasonably withheld.
- § 6.6.9 Subject to the Owner's prior approval, expenses incurred in accordance with the Construction Manager's standard written personnel policy for relocation and temporary living allowances of the Construction Manager's personnel required for the Work.

#### § 6.7 Other Costs and Emergencies

- § 6.7.1 Other costs incurred in the performance of the Work if, and to the extent, approved in advance in writing by the Owner.
- § 6.7.2 Costs incurred in taking action to prevent threatened damage, injury or loss in case of an emergency affecting the safety of persons and property, as provided in Section 10.4 of AIA Document A201–2007.
- § 6.7.3 Costs of repairing or correcting damaged or nonconforming Work executed by the Construction Manager, Subcontractors or suppliers, provided that such damaged or nonconforming Work was not caused by negligence or failure to fulfill a specific responsibility of the Construction Manager, a Subcontractor or a supplier and only to the extent that the cost of repair or correction is not recoverable by the Construction Manager from insurance, sureties, Subcontractors, suppliers, or others.
- § 6.7.4 The costs described in Sections 6.1 through 6.7 shall be included in the Cost of the Work, notwithstanding any provision of AIA Document A201–2007 or other Conditions of the Contract which may require the Construction Manager to pay such costs, unless such costs are excluded by the provisions of Section 6.8.

#### § 6.8 Costs Not To Be Reimbursed

- § 6.8.1 The Cost of the Work shall not include the items listed below:
  - .1 Salaries and other compensation of the Construction Manager's personnel stationed at the Construction Manager's principal office or offices other than the site office, except as specifically provided in Section 6.2, or as may be provided in Article 11;
  - 2 Expenses of the Construction Manager's principal office and offices other than the site office except as may be allowed elsewhere in the Contract Documents;
  - .3 Overhead and general expenses, except as may be expressly included in Sections 6.1 to 6.7;
  - 4 The Construction Manager's capital expenses, including interest on the Construction Manager's capital employed for the Work;
  - 5 Except as provided in Section 6.7.3 of this Agreement, costs due to the negligence or failure of the Construction Manager, Subcontractors and suppliers or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable to fulfill a specific responsibility of the Contract;
  - **.6** Any cost not specifically and expressly described in Sections 6.1 to 6.7;
  - .7 Costs, other than costs included in Change Orders approved by the Owner, that would cause the Guaranteed Maximum Price to be exceeded; and
  - **.8** Costs for services incurred during the Preconstruction Phase.
  - Except as provided in Section 6.7.3, costs to repair defective Work and other costs to comply with Contractor's warranty obligations under the Contract.
  - .10 Subject to and as limited by Section 9.3.3 of the AIA Document A201<sup>TM</sup> 2017, costs and expenses arising from Construction Manager's indemnity obligations, including without limitation, Construction

- Manager's costs and expenses in removing or defending against a mechanic's lien or surety bond claim asserted against the Owner and/or its property or the Construction Manager.
- .11 Costs incurred in the submission process relating to the Request for Proposals issued by the Owner.
- Rental costs or lease payments for vehicles used solely for commuting, together with any and all automobile liability premiums.

#### § 6.9 Discounts, Rebates and Refunds

§ 6.9.1 Cash discounts obtained on payments made by the Construction Manager shall accrue to the Owner if (1) before making the payment, the Construction Manager included them in an Application for Payment and received payment from the Owner, or (2) the Owner has deposited funds with the Construction Manager with which to make payments; otherwise, cash discounts shall accrue to the Construction Manager. Trade discounts, rebates, refunds and amounts received from sales of surplus materials and equipment shall accrue to the Owner, and the Construction Manager shall make provisions so that they can be obtained.

§ 6.9.2 Amounts that accrue to the Owner in accordance with the provisions of Section 6.9.1 shall be credited to the Owner as a deduction from the Cost of the Work.

#### § 6.10 Related Party Transactions

§ 6.10.1 For purposes of Section 6.10, the term "related party" shall mean a parent, subsidiary, affiliate or other entity having common ownership or management with the Construction Manager; any entity in which any stockholder in, or management employee of, the Construction Manager owns any interest in excess of ten percent in the aggregate; or any person or entity which has the right to control the business or affairs of the Construction Manager. The term "related party" includes any member of the immediate family of any person identified above.

§ 6.10.2 If any of the costs to be reimbursed arise from a transaction between the Construction Manager and a related party, the Construction Manager shall notify the Owner of the specific nature of the contemplated transaction, including the identity of the related party and the anticipated cost to be incurred, before any such transaction is consummated or cost incurred. If the Owner, after such notification, authorizes the proposed transaction, then the cost incurred shall be included as a cost to be reimbursed, and the Construction Manager shall procure the Work, equipment, goods or service from the related party, as a Subcontractor, according to the terms of Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3. If the Owner fails to authorize the transaction, the Construction Manager shall procure the Work, equipment, goods or service from some person or entity other than a related party according to the terms of Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3.

#### § 6.11 Accounting Records

The Construction Manager shall keep full and detailed records and accounts related to the cost of the Work and exercise such controls as may be necessary for proper financial management under this Contract and to substantiate all costs incurred. The accounting and control systems shall be satisfactory to the Owner. The Owner and the Owner's auditors shall, during regular business hours and upon reasonable notice, be afforded access to, and shall be permitted to audit and copy, the Construction Manager's records and accounts, including complete documentation supporting accounting entries, books, correspondence, instructions, drawings, receipts, subcontracts, Subcontractor's proposals, purchase orders, vouchers, memoranda and other data relating to this Contract. The Construction Manager shall preserve these records for a period of three years after final payment, or for such longer period as may be required by law.

#### ARTICLE 7 PAYMENTS FOR CONSTRUCTION PHASE SERVICES

#### § 7.1 Progress Payments

§ 7.1.1 Based upon Applications for Payment submitted to the Architect by the Construction Manager and Certificates for Payment issued by the Architect, the Owner shall make progress payments on account of the Contract Sum to the Construction Manager as provided below and elsewhere in the Contract Documents.

§ 7.1.2 The period covered by each Application for Payment shall be one calendar month ending on the last day of the

§ 7.1.3 Provided that an Application for Payment is received by the Architect not later than the 25<sup>th</sup> day of a month, the Owner shall make payment of the certified amount to the Construction Manager not later than the 25th day of the following month. If an Application for Payment is received by the Architect after the application date fixed above,

lnit.

**User Notes:** 

AIA Document A133<sup>™</sup> – 2009 (formerly A121 <sup>™</sup>CMc – 2003). Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:05:37 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

payment shall be made by the Owner not later than thirty (30) days after the Architect receives the Application for Payment.

(Federal, state or local laws may require payment within a certain period of time.)

- § 7.1.4 With each Application for Payment, the Construction Manager shall submit such evidence as may be reasonably necessary to demonstrate costs incurred or to be incurred by the Construction Manager on account of the Cost of Work during the current month, which may include payrolls, receipted invoices or invoices with check vouchers attached, for individual costs in excess of \$25,000. Such application for payment shall be delivered to Owner along with the log referenced in 2.3.2.7, above, and pictures documenting any completed Work.
- § 7.1.5 Each Application for Payment shall be based on the most recent schedule of values submitted by the Construction Manager in accordance with the Contract Documents. The schedule of values shall allocate the entire Guaranteed Maximum Price among the various portions of the Work, except that the Construction Manager's Fee shall be shown as a single separate item. The schedule of values shall be prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Construction Manager's Applications for Payment. The allocation of the Guaranteed Maximum Price under this Section 7.1.5 is to assist the Owner and Architect in evaluation of Construction Manager's applications for payment as to the percentage of work completed and shall not constitute a separate guaranteed maximum price for the Cost of the Work of each individual line item in the schedule of values. Owner shall not be entitled to a reduction in the Guaranteed Maximum Price except to the extent the aggregate price for the various portions of the Work is less than the Guaranteed Maximum Price as evaluated on final completion of the Work, prior to which time, all savings on any particular line item in the schedule of values may be treated as a contingency fund for cost overruns on any other line item.
- § 7.1.6 Applications for Payment shall show the percentage of completion of each portion of the Work as of the end of the period covered by the Application for Payment. The percentage of completion shall be the lesser of (1) the percentage of that portion of the Work which has actually been completed, or (2) the percentage obtained by dividing (a) the expense that has actually been incurred by the Construction Manager on account of that portion of the Work for which the Construction Manager has made or intends to make actual payment prior to the next Application for Payment by (b) the share of the Guaranteed Maximum Price allocated to that portion of the Work in the schedule of values.
- § 7.1.7 Subject to other provisions of the Contract Documents, the amount of each progress payment shall be computed as follows:
  - .1 Take that portion of the Guaranteed Maximum Price properly allocable to completed Work as determined by multiplying the percentage of completion of each portion of the Work by the share of the Guaranteed Maximum Price allocated to that portion of the Work in the schedule of values. Pending final determination of cost to the Owner of changes in the Work, amounts not in dispute shall be included as provided in Section 7.3.9 of AIA Document A201–2007;
  - Add that portion of the Guaranteed Maximum Price properly allocable to materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work, or if approved in advance by the Owner, suitably stored off the site at a location agreed upon in writing, plus any contingency allowance as may be provided in the Contract Documents;
  - .3 Add the Construction Manager's Fee, less retainage of ten percent (10%). The Construction Manager's Fee shall be computed upon the Cost of the Work at the rate stated in Section 5.1 or, if the Construction Manager's Fee is stated as a fixed sum in that Section, shall be an amount that bears the same ratio to that fixed-sum fee as the Cost of the Work bears to a reasonable estimate of the probable Cost of the Work upon its completion;
  - .4 Subtract retainage of N/A percent (N/A%) from that portion of the Work that the Construction Manager self-performs;
  - .5 Subtract the aggregate of previous payments made by the Owner;
  - **.6** Subtract the shortfall, if any, indicated by the Construction Manager in the documentation required by Section 7.1.4 to substantiate prior Applications for Payment, or resulting from errors subsequently discovered by the Owner's auditors in such documentation; and
  - .7 Subtract amounts, if any, for which the Architect has withheld or nullified a Certificate for Payment as provided in Section 9.5 of AIA Document A201–2007.

- § 7.1.8 The Owner and Construction Manager shall agree upon (1) a mutually acceptable procedure for review and approval of payments to Subcontractors and (2) the percentage of retainage held on Subcontracts, and the Construction Manager shall execute subcontracts in accordance with those agreements. Provided release of retainage is consented to by Owner, which consent shall not be unreasonably withheld, and funded by Owner's lender, payment of previously held retainage shall be made to those Subcontractors whose work is fully completed or reduced with respect to Subcontractors whose Work is substantially complete.
- § 7.1.9 Except with the Owner's prior approval, the Construction Manager shall not make advance payments to suppliers for materials or equipment which have not been delivered and stored at the site.
- § 7.1.10 In taking action on the Construction Manager's Applications for Payment, the Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Construction Manager and shall not be deemed to represent that the Architect has made a detailed examination, audit or arithmetic verification of the documentation submitted in accordance with Section 7.1.4 or other supporting data; that the Architect has made exhaustive or continuous on-site inspections; or that the Architect has made examinations to ascertain how or for what purposes the Construction Manager has used amounts previously paid on account of the Contract. Such examinations, audits and verifications, if required by the Owner, will be performed by the Owner's auditors acting in the sole interest of the Owner.

#### § 7.2 Final Payment

- § 7.2.1 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be made by the Owner to the Construction Manager when
  - .1 the Construction Manager has fully performed the Contract except for the Construction Manager's responsibility to correct Work as provided in Section 12.2.2 of AIA Document A201–2007, and to satisfy other requirements, if any, which extend beyond final payment;
  - the Construction Manager has submitted a final accounting for the Cost of the Work and a final Application for Payment; and
  - .3 a final Certificate for Payment has been issued by the Architect.

The Owner's final payment to the Construction Manager shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment.

- § 7.2.2 The Owner's auditors will review and report in writing on the Construction Manager's final accounting within 30 days after delivery of the final accounting to the Architect by the Construction Manager. Based upon such Cost of the Work as the Owner's auditors report to be substantiated by the Construction Manager's final accounting, and provided the other conditions of Section 7.2.1 have been met, the Architect will, within seven days after receipt of the written report of the Owner's auditors, either issue to the Owner a final Certificate for Payment with a copy to the Construction Manager, or notify the Construction Manager and Owner in writing of the Architect's reasons for withholding a certificate as provided in Section 9.5.1 of the AIA Document A201–2007. The time periods stated in this Section supersede those stated in Section 9.4.1 of the AIA Document A201–2007. The Architect is not responsible for verifying the accuracy of the Construction Manager's final accounting.
- § 7.2.3 If, subsequent to final payment and at the Owner's request, the Construction Manager incurs costs described in Section 6.1.1 and not excluded by Section 6.8 to correct defective or nonconforming Work, the Owner shall reimburse the Construction Manager such costs and the Construction Manager's Fee applicable thereto on the same basis as if such costs had been incurred prior to final payment, but not in excess of the Guaranteed Maximum Price. If the Construction Manager has participated in savings as provided in Section 5.2.1, the amount of such savings shall be recalculated and appropriate credit given to the Owner in determining the net amount to be paid by the Owner to the Construction Manager.

#### (Paragraph deleted)

**User Notes:** 

#### ARTICLE 8 INSURANCE AND BONDS

For all phases of the Project, the Construction Manager and the Owner shall purchase and maintain insurance, and the Construction Manager shall provide bonds as set forth in Article 11 of AIA Document A201–2007. The Construction Manager shall deliver the required bonds to the Owner not later than the 10th day after the date the Construction Manager executes the GMP Addendum

(State bonding requirements, if any, and limits of liability for insurance required in Article 11 of AIA Document A201–2007.)

#### Type of Insurance or Bond

#### Limit of Liability or Bond Amount (\$0.00)

(Row deleted)

Commercial General Liability Insurance

General policy aggregate: \$2,000,000;

Bodily injury / property damage per occurrence: \$1,000,000; Personal / advertising injury: \$1,000,000 any one person or

organization;

Products / completed ops aggregate: \$2,000,000

#### ARTICLE 9 DISPUTE RESOLUTION

§ 9.1 Any Claim between the Owner and Construction Manager shall be resolved in accordance with the provisions set forth in this Article 9 and Article 15 of A201–2007. However, for Claims arising from or relating to the Construction Manager's Preconstruction Phase services, no decision by the Initial Decision Maker shall be required as a condition precedent to mediation or binding dispute resolution, and Section 9.3 of this Agreement shall not apply.

§ 9.2 For any Claim subject to, but not resolved by mediation pursuant to Section 15.3 of AIA Document A201–2007, the method of binding dispute resolution shall be as follows:

(Check the appropriate box. If the Owner and Construction Manager do not select a method of binding dispute resolution below, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, Claims will be resolved by litigation in a court of competent jurisdiction.)

(Paragraph deleted)

[X] Litigation in a court of competent jurisdiction

#### § 9.3 Initial Decision Maker

The Architect will serve as the Initial Decision Maker pursuant to Section 15.2 of AIA Document A201–2007 for Claims arising from or relating to the Construction Manager's Construction Phase services, unless the parties appoint below another individual, not a party to the Agreement, to serve as the Initial Decision Maker.

(If the parties mutually agree, insert the name, address and other contact information of the Initial Decision Maker, if other than the Architect.)

#### ARTICLE 10 TERMINATION OR SUSPENSION

#### § 10.1 Termination Prior to Establishment of the Guaranteed Maximum Price

§ 10.1.1 Prior to the execution of the Guaranteed Maximum Price Amendment, the Owner may terminate this Agreement upon not less than seven days' written notice to the Construction Manager for the Owner's convenience and without cause, and the Construction Manager may terminate this Agreement, upon not less than seven days' written notice to the Owner, for the reasons set forth in Section 14.1.1 of A201–2007.

§ 10.1.2 In the event of termination of this Agreement pursuant to Section 10.1.1, the Construction Manager shall be equitably compensated for Preconstruction Phase services performed prior to receipt of a notice of termination. In no event shall the Construction Manager's compensation under this Section exceed the compensation set forth in Section 4.1.

§ 10.1.3 If the Owner terminates the Contract pursuant to Section 10.1.1 after the commencement of the Construction Phase but prior to the execution of the Guaranteed Maximum Price Amendment, the Owner shall pay to the Construction Manager an amount calculated as follows, which amount shall be in addition to any compensation paid to the Construction Manager under Section 10.1.2:

- .1 Take the Cost of the Work incurred by the Construction Manager to the date of termination;
- .2 Add the Construction Manager's Fee computed upon the Cost of the Work to the date of termination at the rate stated in Section 5.1 or, if the Construction Manager's Fee is stated as a fixed sum in that

- Section, an amount that bears the same ratio to that fixed-sum Fee as the Cost of the Work at the time of termination bears to a reasonable estimate of the probable Cost of the Work upon its completion; and
- Subtract the aggregate of previous payments made by the Owner for Construction Phase services.

The Owner shall also pay the Construction Manager fair compensation, either by purchase or rental at the election of the Owner, for any equipment owned by the Construction Manager which the Owner elects to retain and which is not otherwise included in the Cost of the Work under Section 10.1.3.1. To the extent that the Owner elects to take legal assignment of subcontracts and purchase orders (including rental agreements), the Construction Manager shall, as a condition of receiving the payments referred to in this Article 10, execute and deliver all such papers and take all such steps, including the legal assignment of such subcontracts and other contractual rights of the Construction Manager, as the Owner may require for the purpose of fully vesting in the Owner the rights and benefits of the Construction Manager under such subcontracts or purchase orders. All Subcontracts, purchase orders and rental agreements entered into by the Construction Manager will contain provisions allowing for assignment to the Owner as described above.

If the Owner accepts assignment of subcontracts, purchase orders or rental agreements as described above, the Owner will reimburse or indemnify the Construction Manager for all costs arising under the subcontract, purchase order or rental agreement, if those costs would have been reimbursable as Cost of the Work if the contract had not been terminated. If the Owner chooses not to accept assignment of any subcontract, purchase order or rental agreement that would have constituted a Cost of the Work had this agreement not been terminated, the Construction Manager will terminate the subcontract, purchase order or rental agreement and the Owner will pay the Construction Manager the costs necessarily incurred by the Construction Manager because of such termination, excluding any Fee, profit, overhead or other compensation for work not yet performed, or other damages excluded under the Contract.

#### § 10.2 Termination Subsequent to Establishing Guaranteed Maximum Price

Following execution of the Guaranteed Maximum Price Amendment and subject to the provisions of Section 10.2.1 and 10.2.2 below, the Contract may be terminated as provided in Article 14 of AIA Document A201–2007.

- § 10.2.1 If the Owner terminates the Contract after execution of the Guaranteed Maximum Price Amendment, the amount payable to the Construction Manager pursuant to Sections 14.2 and 14.4 of A201-2007 shall not exceed the amount the Construction Manager would otherwise have received pursuant to Sections 10.1.2 and 10.1.3 of this Agreement.
- § 10.2.2 If the Construction Manager terminates the Contract after execution of the Guaranteed Maximum Price Amendment, the amount payable to the Construction Manager under Section 14.1.3 of A201-2007 shall not exceed the amount the Construction Manager would otherwise have received under Sections 10.1.2 and 10.1.3 above. The Construction Manager shall not be entitled to any Fee, profit, overhead or other compensation for work not yet performed, or other damages excluded under the Contract.

#### § 10.3 Suspension

The Work may be suspended by the Owner as provided in Article 14 of AIA Document A201–2007. In such case, the Guaranteed Maximum Price and Contract Time shall be increased as provided in Section 14.3.2 of AIA Document A201-2007, except that the term "profit" shall be understood to mean the Construction Manager's Fee as described in Sections 5.1 and 5.3.5 of this Agreement.

#### **MISCELLANEOUS PROVISIONS**

§ 11.1 Terms in this Agreement shall have the same meaning as those in A201–2007.

#### § 11.2 Ownership and Use of Documents

Section 1.5 of A201–2007 shall apply to both the Preconstruction and Construction Phases.

#### § 11.3 Governing Law

Section 13.1 of A201–2007 shall apply to both the Preconstruction and Construction Phases.

#### § 11.4 Assignment

**User Notes:** 

The Owner and Construction Manager, respectively, bind themselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Construction Manager shall assign this Agreement

Init.

AlA Document A133™ - 2009 (formerly A121™CMc - 2003). Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The "American Institute of Architects. "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:05:37 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement. Except as provided in Section 13.2.2 of A201–2007, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

#### § 11.5 Other provisions:

- § 11.5.1 In addition to any other basis specified in the Contract Documents, the following shall constitute basis for termination by the Construction Manager:
  - (a) Owner's failure to pay Construction Manager in accordance with the terms of the Contract for a period of more than 30 calendar days;
  - (b) Owner's refusal for a period of more than 30 calendar days from written request by the Construction Manager to issue appropriate Change Orders; and
  - (c) Owner's failure to provide Construction Manager with adequate plans and specifications for Construction Manager to perform its Work in an efficient and workmanlike manner unless the Owner promptly remedies that failure through plan or specification amendments within 30 calendar days from written request by Construction Manager.
  - (d) Owner's failure to provide Construction Manager with reasonable evidence that the Owner has made satisfactory financial arrangements to fulfill the Owner's obligations under the Contract within 10 calendar days from written request by Construction Manager.

#### ARTICLE 12 SCOPE OF THE AGREEMENT

§ 12.1 This Agreement represents the entire and integrated agreement between the Owner and the Construction Manager and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both Owner and Construction Manager.

- § 12.2 The following documents comprise the Agreement:
  - AIA Document A133-2009, Standard Form of Agreement Between Owner and Construction Manager as Constructor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price.
  - .2 AIA Document A201–2007, General Conditions of the Contract for Construction.
  - .3 AIA Document E201<sup>TM</sup>\_2007, Digital Data Protocol Exhibit, if completed.
  - AIA Document E202<sup>TM</sup>–2008, Building Information Modeling Protocol Exhibit, if completed.
  - Other documents:
    - Exhibit A Guaranteed Maximum Price Amendment
    - Exhibit B Construction Schedule to be established with the GMP Amendment
    - Exhibit C Construction Manager Certifications
    - Exhibit D Certificate of Insurance
    - Exhibit E Project Proposal/Billable Rates

This Agreement is entered into as of the day and year first written above.

OWNER (Signature)	CONSTRUCTION MANAGER (Signature)
	Chris Beasley Chief Operating Officer
(Printed name and title)	(Printed name and title)

**User Notes:** 

(1651987766)

### **Guaranteed Maximum Price Amendment**

#### for the following PROJECT:

(Name and address or location)

International Leadership of Texas Charter Schools Portable Classrooms 3610 Longmire Drive, College Station, TX 77840

#### THE OWNER:

(Name, legal status and address)

International Leadership of Texas 1820 N. Glenville Dr., #100, Richardson, TX 75081

#### THE CONSTRUCTION MANAGER:

(Name, legal status and address)

Key Construction, Inc. 741 W. 2<sup>nd</sup> St, Wichita, KS 67203

#### ARTICLE A.1

#### § A.1.1 Guaranteed Maximum Price

Pursuant to Section 2.2.6 of the Agreement, the Owner and Construction Manager hereby amend the Agreement to establish a Guaranteed Maximum Price. As agreed by the Owner and Construction Manager, the Guaranteed Maximum Price is an amount that the Contract Sum shall not exceed. The Contract Sum consists of the Construction Manager's Fee plus the Cost of the Work, as that term is defined in Article 6 of this Agreement.

- **§ A.1.1.1** The Contract Sum is guaranteed by the Construction Manager not to exceed Sixty Three Thousand, Five Hundred Ninety Nine and 00/100 Dollars (\$63,599.00), subject to additions and deductions by Change Order as provided in the Contract Documents.
- § A.1.1.2 Itemized Statement of the Guaranteed Maximum Price. Provided below is an itemized statement of the Guaranteed Maximum Price organized by trade categories, allowances, contingencies, alternates, the Construction Manager's Fee, and other items that comprise the Guaranteed Maximum Price. (Provide below or reference an attachment.)
  - See Exhibit A.1 GMP Summary and Basis
- § A.1.1.3 The Guaranteed Maximum Price is based on the following alternates, if any, which are described in the Contract Documents and are hereby accepted by the Owner: (State the numbers or other identification of accepted alternates. If the Contract Documents permit the Owner to accept other alternates subsequent to the execution of this Amendment, attach a schedule of such other alternates showing the amount for each and the date when the amount expires.)
  - See Exhibit A.1- GMP Summary and Basis

§ A.1.1.4 Allowances included in the Guaranteed Maximum Price, if any: (*Identify allowance and state exclusions, if any, from the allowance price.*)

#### **ADDITIONS AND DELETIONS:**

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

AIA Document A201™–2007, General Conditions of the Contract for Construction, is adopted in this document by reference. Do not use with other general conditions unless this document is modified.

Init.

AIA Document A133" – 2009 Exhibit A. Copyright © 1991, 2003 and 2009 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:14:21 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

User Notes:

•	See Exhibit A.1 – GMP Summary and Basis	
§ A.1.	1.5 Assumptions, if any, on which the Guaranteed Maxin	mum Price is based:
•	See Exhibit A.1 – GMP Summary and Basis	
§ A.1. Contra	<b>1.6</b> The Guaranteed Maximum Price is based upon the foact:	ollowing Supplementary and other Conditions of the
•		
	<b>1.7</b> The Guaranteed Maximum Price is based upon the for list the Specifications here, or refer to an exhibit attack	
<b>,</b>	See Exhibit A.2 – List of Construction Documents	
	<b>1.8</b> The Guaranteed Maximum Price is based upon the for list the Drawings here, or refer to an exhibit attached a See Exhibit A.2 – List of Construction Documents	
	<b>1.9</b> The Guaranteed Maximum Price is based upon the formation here, or refer to an example of the company of t	
N/A		
-	LE A.2  1 The anticipated date of Substantial Completion established.	shed by this Amendment:
20 cal	endar days from the date of Commencement as defined i	in Section 2.3.1.2 of the agreement.
OM/N	IER (Signature)	CONSTRUCTION MANAGER (Signature)
OWN		
OWN		Chris Beasley Chief Operating Officer

		lege Station, 1
DESCRIPTION	BIDS	95% - AUG 2020
TEMP FENCING ALLOWANCE	Ś	2.750.
DEMO & REPLACE FIRE LANE FOR PLUMBING (INCLUDES JOINT SEALANTS)	T	With Plumb
4" SIDEWALKS		With Plumb
TRANSFORMER PAD		
PAVEMENT MARKINGS	\$	975
PLUMBING	\$	22,250
ELECTRICAL	\$	17,750
EROSION CONTROL	\$	3,019
EROSION CONTROL REPAIR AND REMOVAL ALLOWANCE	\$	425
LANDSCAPING (SOD) ALLOWANCE	\$	50
COLUMN TOTALS	\$	47,669
TAX ON MATERIALS		
SUBTOTAL (COST OF WORK)		
BOND PERF	\$	
BUILDERS RISK	\$	
CONTRACTOR FIXED COSTS		
PRE CONSTRUCTION		
STAFF	\$	8,23
GENERAL REQUIREMENTS	\$	1,91
RECOMMENDED CONTINGENCIES		
OVERTIME/ACCELERATION CONTINGENCY	Ś	
CM CONTINGENCY	\$	
OWNER CONTINGENCY (DESIGN CONTINGENCY)	\$	
CITY, COUNTY, UTILITY FEE'S		
BUILDING PERMITS	\$	
PROFIT/OVERHEAD	\$	5,78
	<u>'</u>	
TOTAL GMP	\$	63,5



#### **GMP** Clarifications:

- Assumes work performed during normal business hours
- Includes domestic water and sewer connections to restroom portable
- Includes electrical service connection from transformer to portable classrooms and portable restroom
- GMP includes pricing through Addendum 3 dated 08-10-20.
- No alternates to Contract Documents have been generated or accepted.
- Allowances Inclusions (OH&P & performance & payment bonds figured on top of allowances)
  - o Temporary Fencing Allowance: \$2,750.00
  - o Erosion Control Removal Allowance: \$425.00
  - Landscape Sod Allowance: \$500.00

#### Exclusions:

- o Builder's Risk
- Irrigation Modifications
- o Electrical Overhead Service
- Portable Classrooms and/or Restrooms
- Data/Communications for portables
- Domestic Water Line Tap (by City)
- o 3rd Party Testing
- o 3<sup>rd</sup> Party Inspections
- 3<sup>rd</sup> Party Design Review
- Dewatering due to groundwater / high water tables
- No DDC Controls

G E N E R A L CONTRACTOR 741 WEST SECOND WICHITA KS 67203 316-263-9515 FAX: 316-263-1161



# Attachment A.2

Printed on Mon Aug 24, 2020 at 10:50 am CDT

Job #: 20-03-044 International Leadership of Texas Portable Classrooms - College Station, TX 3610 Longmire Drive College Station, Texas 77840

#### **Current Drawings**

Drawing No.	Drawing Title	Revision	<b>Drawing Date</b>	Received Date	Set
Civil					
C0.00	NOTES	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
C1.00	SITE PLAN OVERALL	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
C1.01	SITE PLAN	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
C2.00	DEMO PLAN	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
C3.00	GRADING AND DRAINAGE	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
C4.00	EROSION CONTROL	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
C5.00	DETAILS	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
S1	Sewer Details	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
S2	Sewer Details	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
W1	Water Details	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)
W2	Water Details	0	07/10/2020	08/03/2020	For Permit Set (07/10/20)

Exhibit B

- o Schedule:
  - Plumbing 7 working days
  - Concrete 3 working days
  - Electrical 5 working days
  - Pavement Markings & Sealants 2 working days
  - Landscape Repairs (sod) 1 working day
- o Total Overall Schedule 10 working days

GENERAL CONTRACTOR 741 WEST SECOND WICHITA KS 67203 316-263-9515 FAX: 316-263-1161

### Construction Manager Certifications

#### **REPRESENTATIONS AND CERTIFICATIONS**

By execution and submission of this Proposal, the Offeror hereby represents and warrants to Owner as follows:

- a) The Offeror has prior experience on Construction Management at Risk projects of the same or similar scope, nature and class as the Work described.
- b) The Offeror has read and understands the Proposal Documents and the Contract Documents, and this Proposal is made in accordance with the Proposal Documents.
- c) To the fullest extent permitted by applicable law, the Offeror waives any claim it has or may have against the Owner, the Architect, and their respective officers, shareholders, directors, trustees, partners, agents, contractors, consultants and employees arising out of or in connection with the administration, evaluation or recommendation of any offers; waiver of any requirements under the Proposal Documents or the Contract Documents; acceptance or rejection of any proposals; and the award of the Contract.
- d) The Project will be undertaken in accordance with the applicable provisions of Chapter 44 of the Texas Education Code and Section 2269 of the Texas Government Code.
- e) Represents that to the best of its knowledge it is not indebted to IL Texas. Indebtedness to IL Texas shall be basis for non-award and/or cancellation of any award.
- f) Certifies that no suspension or debarment is in place, which would preclude receiving a federally funded contract.
- g) Section 44.034, of the Texas Education Code requires a person or business entity that enters into a contract with a school district or charter school must give advance notice to the district or school if the person and/or Owner or operator of the business entity has been convicted of a felony. The notice must include a general description of the conduct resulting in the conviction of a felony. Subsection (b), which also applies to charter schools, states "a school district may terminate a contract with a person or business entity if the district determines that the person or business entity failed to give notice as required by Subsection (a) or misrepresented the conduct resulting in the conviction. The district must compensate the person or business entity for services performed before the termination of the contract." THIS NOTICE IS NOT REQUIRED OF A PUBLICLY HELD CORPORATION.

I, the undersigned agent for the firm named below, certify that the information concerning notification of felony convictions has been reviewed by me and the following information furnished is true to the best of my knowledge.

#### **COMPANY NAME:**

l	_My firm is a publicly held corporation; therefore, this reporting requirement is not applicable.
2X	My firm is not owned nor operated by anyone who has been convicted of a felony.
3	_ My firm is owned or operated by the following individual(s) who has/have been convicted of a felony.

43 of 99

	Name of Felon(s):
	Felon(s):
	Details of Conviction(s):
CHECK	( (√) 1, 2, OR 3 ABOVE AND SIGN BELOW
d.	"Non-Collusion Statement": "The undersigned affirms that they are duly authorized to execute this contract, that this company, corporation, firm, partnership, etc., or individual has not prepared this proposal in collusion (An agreement between two or more persons to deceive the school district or defraud the school district of its rights) with any other bidder, school board member, or school employee, and that the contents of this proposal as to prices, quality of product, terms and/or conditions, etc., have not been communicated by the undersigned nor by any other employee, agent and/or representative of the company, corporation, firm, partnership, etc., or individual to any other person engaged in this type of business prior to the official opening of his proposal for the intent or purpose of collusion."
Bidder	Name/Title: Scott Casebolt - President Bidder Signature:
Telepho	one No. 316-263-9515 Fax No. 316-263-1161 Address 741 W. 2nd Street Wichita, KS 67203

Form (Rev. October 2018)
Department of the Treasury

Internal Revenue Service

# Request for Taxpayer Identification Number and Certification

▶ Go to www.irs.gov/FormW9 for instructions and the latest information.

Give Form to the requester. Do not send to the IRS.

	1	EY CONSTRUC	•	•	ile is req	quired on	iriis iirie, do	not leave thi	s iirie biarik										
	_	2 Business name/disregarded entity name, if different from above																	
page 3.											CE	4 Exemptions (codes apply only to certain entities, not individuals; see instructions on page 3):							
e. ns on	] [	☐ Individual/sole proprietor or ☐ C Corporation ☑ S Corporation ☐ Partnership ☐ Trust/estate									Exempt payee code (if any)								
ct ty	☐ Limited liability company. Enter the tax classification (C=C corporation, S=S corporation, P=Partnership) ▶											.							
Print or type. See Specific Instructions on page	Note: Check the appropriate box in the line above for the tax classification of the single-member owner. Do not check LLC if the LLC is classified as a single-member LLC that is disregarded from the owner unless the owner of the LLC is another LLC that is not disregarded from the owner for U.S. federal tax purposes. Otherwise, a single-member LLC that is disregarded from the owner should check the appropriate box for the tax classification of its owner.										3   -	code (if any)							
ěC <u>i</u>		Other (see instruc											(Ap	plies to a	ccounts	mainta	ained o	utside t	he U.S.)
ςS	5	Address (number, str	reet, and apt.	or suite no.) Se	e instru	ctions.				Reque	ster's	nam	e and	addre	ss (op	tiona	l)		
Še	74	1 W 2ND ST N																	
0,	6	City, state, and ZIP c	ode							]									
	W	CHITA, KS 6720	03																
	7	_ist account number(	(s) here (optio	nal)			<del></del>												
Par	tΙ	Taxpayer	Identific	ation Num	nber (	TIN)													
		r TIN in the approp									So	cial s	ecuri	y nun	ber				
		ithholding. For ind								for a									T
		ilien, sole proprieto is your employer i								et a				-		-			
TIN, la			idonimodilo	Triambor (Eli	·, , ·	,a ao 1101	t navo a no		non to ge		or		_			'			
Note:	If th	ne account is in mo	ore than on	e name, see tl	he instr	ructions	for line 1.	Also see W	hat Name	and	Em	ploy	er ide	ntifica	tion n	umb	er		
Numb	er 7	o Give the Reques	s <i>ter</i> for guic	delines on who	ose nun	mber to e	enter.							4 0			_		
											4	8	-	1 0	3	8	5	8	1
Par	t II	Certificat	ion														·		
Unde	r pe	nalties of perjury, I	certify that	:															
2. I an	<ol> <li>The number shown on this form is my correct taxpayer identification number (or I am waiting for a number to be issued to me); and</li> <li>I am not subject to backup withholding because: (a) I am exempt from backup withholding, or (b) I have not been notified by the Internal Revenue Service (IRS) that I am subject to backup withholding as a result of a failure to report all interest or dividends, or (c) the IRS has notified me that I am</li> </ol>																		
		er subject to back			•														
3. I an	n a	J.S. citizen or othe	er U.S. pers	on (defined be	elow); a	and													
4. The	FA	TCA code(s) enter	ed on this f	orm (if any) inc	dicating	g that I a	am exempt	from FATC	A reportir	ng is co	rrect.								
you ha	ive t	on instructions. You ailed to report all in a or abandonment of interest and divide	iterest and o	dividends on your property, cance	our tax r	return. Fo	or real esta contribution	te transactions to an indi	ons, item 2 ividual retir	does nement a	ot ap ırranç	ply. I geme	Form ent (IR	ortga( A), an	ge inte d gen	erest erall	paid y, pa	i, iymer	nts
Sign Here		Signature of U.S. person ▶	M	NS	52				-	Date ►	01	1	30	17	50	_0	)		
Ge	ne	ral Instruc	tions					• Form 10 funds)	99-DIV (di	vidends	, inc	ludin	g tho	se fro	m sto	cks	or n	nutua	ıl

Section references are to the Internal Revenue Code unless otherwise noted.

**Future developments.** For the latest information about developments related to Form W-9 and its instructions, such as legislation enacted after they were published, go to www.irs.gov/FormW9.

#### Purpose of Form

An individual or entity (Form W-9 requester) who is required to file an information return with the IRS must obtain your correct taxpayer identification number (TIN) which may be your social security number (SSN), individual taxpayer identification number (ITIN), adoption taxpayer identification number (ATIN), or employer identification number (EIN), to report on an information return the amount paid to you, or other amount reportable on an information return. Examples of information returns include, but are not limited to, the following.

• Form 1099-INT (interest earned or paid)

- Form 1099-MISC (various types of income, prizes, awards, or gross proceeds)
- Form 1099-B (stock or mutual fund sales and certain other transactions by brokers)
- Form 1099-S (proceeds from real estate transactions)
- Form 1099-K (merchant card and third party network transactions)
- Form 1098 (home mortgage interest), 1098-E (student loan interest), 1098-T (tuition)
- Form 1099-C (canceled debt)
- Form 1099-A (acquisition or abandonment of secured property)

Use Form W-9 only if you are a U.S. person (including a resident alien), to provide your correct TIN.

If you do not return Form W-9 to the requester with a TIN, you might be subject to backup withholding. See What is backup withholding, later.

Form W-9 (Rev. 10-2018)

	<del></del>
CONFLICT OF INTEREST QUESTIONNAIRE	FORM CIQ
For vendor doing business with local governmental entity	
	OFFICE USE ONLY
This questionnaire reflects changes made to the law by H.B. 23, 84th Leg., Regular Session.  This questionnaire is being filed in accordance with Chapter 176, Local Government Code, by a vendor who	
has a business relationship as defined by Section 176.001(1-a) with a local governmental entity and the vendor meets requirements under Section 176.006(a).	Date Received
By law this questionnaire must be filed with the records administrator of the local governmental entity not later than the 7th business day after the date the vendor becomes aware of facts that require the statement to be filed. See Section 176.006(a-1), Local Government Code.	
A vendor commits an offense if the vendor knowingly violates Section 176.006, Local Government Code, An offense under this section is a misdemeanor.	
Name of vendor who has a business relationship with local governmental entity.	
2	
Check this box if you are filing an update to a previously filed questionnaire. (The law re completed questionnaire with the appropriate filing authority not later than the 7th busines you became aware that the originally filed questionnaire was incomplete or inaccurate.)	
Name of local government officer about whom the information is being disclosed.	
Name of Officer	
4 Describe each employment or other pasiness elation hip with the local government office	per, or a family member of the
officer, as described by Section 176.0 131 (2)(A. Also describ) an atamity relationship with	the local government officer.
Complete subparts A and B for each employment or by siness plattic, ship described. Attact CIQ as necessary.	n additional pages to this Form
A. Is the local government officer or a family member of the officer receiving or hit.	rely to receive tayable income
other than investment income, from the vendor?	tely to receive taxable income,
Yes No	
Is the vendor receiving or likely to receive taxable income, other than investment.	income, from or at the direction
of the local government officer or a family member of the officer AND the taxable in local governmental entity?	ncome is not received from the
Yes No	
Describe each employment or business relationship that the vendor named in Section 1 ms other business entity with respect to which the local government officer serves as an of ownership interest of one percent or more.	intains with a corporation or ficer or director, or holds an
Check this havil the words has given the least on a manner officer and family manner	Libe officer and a second by
Check this box if the vendor has given the local government officer or a family member of as described in Section 176.003(a)(2)(B), excluding gifts described in Section 176.003(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(	of the officer one or more gifts [] (3(a-1).
7	
Signature of vendor doing business with the governmental entity  Da	de
Form provided by Texas Ethics Commission www.ethics.state.tx.us	Revised 11/30/2015

# CONFLICT OF INTEREST QUESTIONNAIRE For vendor doing business with local governmental entity

A complete copy of Chapter 176 of the Local Government Code may be found at http://www.statutes.legis.state.tx.us/ Docs/LG/htm/LG.176.htm. For easy reference, below are some of the sections cited on this form.

<u>Local Government Code § 176.001(1-a)</u>: "Business relationship" means a connection between two or more parties based on commercial activity of one of the parties. The term does not include a connection based on:

- (A) a transaction that is subject to rate or fee regulation by a federal, state, or local governmental entity or an agency of a federal, state, or local governmental entity;
- (B) a transaction conducted at a price and subject to terms available to the public; or
- (C) a purchase or lease of goods or services from a person that is chartered by a state or federal agency and that is subject to regular examination by, and reporting to, that agency.

#### Local Government Code § 176.003(a)(2)(A) and (B):

- (a) A local government officer shall file a conflicts disclosure statement with respect to a vendor if:
  - (2) the vendor:
    - (A) has an employment or other business relationship with the local government officer or a family member of the officer that results in the officer or family member receiving taxable income, other than investment income, that exceeds \$2,500 during the 12-month period preceding the date that the officer becomes aware that
      - (i) a contract between the local governmental entity and vendor has been executed;

Of

- (ii) the local governmental entity is considering entering into a contract with the vendor:
- (B) has given to the local government officer or a family member of the officer one or more gifts that have an aggregate value of more than \$100 in the 12-month period preceding the date the officer becomes aware that:
  - (i) a contract between the local governmental entity and vendor has been executed; or
  - (ii) the local governmental entity is considering entering into a contract with the vendor.

#### Local Government Code § 176.006(a) and (a-1)

- (a) Avendor shall file a completed conflict of interest questionnaire if the vendor has a business relationship with a local governmental entity and:
  - (1) has an employment or other business relationship with a local government officer of that local governmental entity, or a family member of the officer, described by Section 176.003(a)(2)(A);
  - (2) has given a local government officer of that local governmental entity, or a family member of the officer, one or more gifts with the aggregate value specified by Section 176.003(a)(2)(B), excluding any gift described by Section 176.003(a-1); or
  - (3) has a family relationship with a local government officer of that local governmental entity.
- (a-1) The completed conflict of interest questionnaire must be filed with the appropriate records administrator not later than the seventh business day after the later of:
  - (1) the date that the vendor:
    - (A) begins discussions or negotiations to enter into a contract with the local governmental entity; or
    - (B) submits to the local governmental entity an application, response to a request for proposals or bids, correspondence, or another writing related to a potential contract with the local governmental entity; or
  - (2) the date the vendor becomes aware:
    - (A) of an employment or other business relationship with a local government officer, or a family member of the officer, described by Subsection (a);
    - (B) that the vendor has given one or more gifts described by Subsection (a); or
    - (C) of a family relationship with a local government officer.

Form provided by Texas Ethics Commission

www.ethics.state.tx.us

Revised 11/30/2015

#### NON COLLUSIVE PROPOSAL CERTIFICATE & ACKNOWLEDGMENT

By submission of this proposal, the Vendor certifies that: (a) the proposal has been independently arrived at without collusion with any other vendor or with any competitor. (b) The proposal has not been knowingly disclosed and shall not be knowingly disclosed, prior to the opening of proposals for this project, to any other vendor, competitor or potential competitor. (c) No attempt has been or will be made to induce any other person, partnership or corporation to submit or not to submit a proposal. (d) The person signing this proposal certifies that he has fully informed himself regarding the accuracy of the statements contained in this certification and under the penalties being applicable to the Vendor as well as to the person signing in its behalf.

I certify that the above information is true and correct. XYES

FELONY CONVICTION NOTIFICATION
State of Texas Legislative Senate Bill No.1, Section 44.034, Notification of Criminal History, Subsection (a), states: "a Person or business entity that enters into a contract with a school district must give advance notice to the district if the Person or an Owner or operator of the business entity has been convicted of a felony. The notice must include a general description of the conduct resulting in the conviction of a felony." Subsection (b) states "a school district may terminate a contract with a person or business entity if the district determines that the person or business entity failed to give notice as required by Subsection (a) or misrepresented the conduct resulting in the conviction. The District must compensate the person or business entity for services performed before the termination of the contract." This Notice Is Not Required of a Publicly Held Corporation. Check the appropriate box and sign the form.
My firm is a publicly held corporation; therefore, this reporting requirement is not applicable.
XXMy firm is not owned or operated by anyone who has been convicted of a felony.
My firm is owned or operated by the following individual(s) who has/have been convicted of a felony:
Name of Felon:
Details of Conviction(s)

## CERTIFICATION REGARDING DEBARMENT SUSPENSION, INELIGIBILITY AND VOLUNTARY EXCLUSION - LOWER TIER COVERED TRANSACTIONS

This certificate is required by the regulations implementing Executive Order 12549, Debarment and Suspension, 7 CFR Part 3017, Section 3017.510, Participants' responsibilities. The regulations were published as Part IV of the January 30, 1989, Federal Register (pages 4722-4733). Copies of the regulations may be obtained by contacting the Department of Agriculture agency with which this transaction originated.

- 1) The prospective lower tier participant certifies, by submission of this proposal, that neither it nor its principals is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction by a Federal department or agency.
- 2) Where the prospective lower tier participant is unable to certify to any of the statements in this certification, such prospective participant shall attach an explanation to this proposal.

PR/Award # or Project Name:	
Check applicable box	V
I (We) certify that our company has not been debarred and is not participating in lobbying activities.	X
See attached explanation and complete disclosure forms.	

#### **CLEAN AIR & WATER ACT CERTIFICATION**

(This is a Federal requirement)

I certify that my company is in compliance with all applicable standards, orders or regulations issued pursuant to the Clean Air Act of 1970, as amended (42 U.S.C. 1857(h), Section 508 of the Clean Water Act, as amended (33 U.S.C. 1368), Executive Order 117389 and Environmental Protection Agency Regulation, 40 CFR Part 15 as required under OMB Circular A-102, Attachment O, Paragraph 14 (1) regarding reporting violations to the grantor agency and to the United States Environmental Protection Agency Assistant Administrator for the Enforcement.

CERTIFICATION REGARDING LOBBYING for FEDERAL FUNDS - Applicable to grants, Subgrants, Cooperative Agreements, and Contracts Exceeding \$100,000 in Federal Funds.

Submission of this certifications a prerequisite for making or entering into this transaction and is imposed by section 1352, Title 31, U.S. Code. This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

The undersigned certifies, to the best of his or her knowledge and belief, that:

- (1) No Federal appropriated funds have been paid or will be paid by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of congress, or an employee of a Member of Congress in connection with the awarding of a Federal contract, the making of a Federal grant, the making of a Federal loan, the entering into a cooperative agreement, and the extension, continuation, renewal, amendment, or modification of a Federal contract, grant, loan, or cooperative agreement.
- (2) If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of congress, or an employee of a Member of Congress in connection with this Federal grant or cooperative agreement the undersigned shall complete and submit Standard Form-LLL, "Disclosure of Lobbying Activities" Form in accordance with its instructions.
- (3) The undersigned shall required that the language of this certification be included in the award documents for all covered subawards exceeding \$100,000 in Federal funds at all appropriate tiers and that all subrecipients shall certify and disclose accordingly.

	Check applicable box	✓
•	I (We) certify that our company has not been debarred and is not participating in lobbying activities.	X
•	See attached explanation and complete disclosure forms.	

### Form to be left blank, per questions, answers and clarifications

## CERTIFICATION OF CRIMINAL HISTORY REVIEW OF EMPLOYEES OF PUBLIC WORKS CONTRACTORS AND SUBCONTRACTORS

I, the undersigned agent for	("Contractor"), certify tha
Contractor has received all criminal history record information	relating to any person to whom
Section 22.08341(b) of the Texas Education Code applies,	and that each of Contractor's
subcontractors on the Project has complied with Section 22.	.08341(e), as it relates to such
subcontractor's employees.	

I further certify that Contractor, and each of its subcontractors, does not and will not employ any person on the Project to whom Section 22.08341(b) applies that has, during the preceding 30 years, been convicted of any of the following offenses, where the victim was under 18 years of age or was enrolled in a public school:

- A felony offense under Title 5, Penal Code;
- An offense on conviction of which a defendant is required to register as a sex offender under Chapter 62, Code of Criminal Procedure; or
- An offense under the laws of another state or federal law that is equivalent to an offense under Subdivision (1) or (2).

I further certify that Contractor, and each of its subcontractors, shall comply with Section 22.08341 of the Texas Education Code in all other respects for the duration of the Project, and that:

- 1. Upon receipt of information that any employee of Contractor, or any of its subcontractors, on the Project has a reported criminal history including any of the above offenses, such employee shall be immediately removed from the Project and notice provided to the School in writing within three business days.
- 2. Upon request, Contractor will provide the School with the name and any other requested information regarding any employee of Contractor, or any of its subcontractors, so that the School may obtain criminal history record information on such person.
- 3. If the School objects to the presence of any employee of Contractor, or any of its subcontractors, on the basis of the employee's criminal history record information, Contractor agrees to immediately remove that employee from the Project, or to discontinue using that subcontractor on the Project until such employee to provide services to the School.

I further certify that Contractor has obtained written certifications and assurances from each of its subcontractors on the Project in the same manner and form as provided herein, and that the form of each subcontractor's certification shall be provided to the School.

I understand that non-compliance with this certification by Contractor may be grounds for contract termination and/or barring disqualified persons from performing the work.

Signature of Contractor Official	Date
Signature of Contractor Official	Date



# State of Texas Health & Human Services Commission

#### **Child Support Certification**

I.

Section 231,006, Texas Family Code, as amended by Section 82 of House Bill No. 433, 74th Regular Legislative Session (Acts 1995, 74th Leg., R.S., cb., 751), prohibits the payment of state funds under a grant, contract, or loan to

a person who is more than 30 days delinquent in the payment of child support, and

 a business entity in which such a person is the sole proprietor, partner, shareholder or owner with an ownership interest of at least 25%.

Section 231,006 further provides that a person or business entity that is ineligible to receive payments for the reasons stated above shall continue to be ineligible to receive payments from the state under a contract, grant, or loan until

- all arrearages have been paid, or
- the person is in compliance with a written repayment agreement or court order as to any existing delinquency.

Section 231.006 further requires each bid, or application for a contract, grant, or loan to include

- the name and social security number of the individual or sole proprietor and each partner, shareholder, or owner with an ownership interest of at least 25% of the business entity submitting the bid or application, and
- . the statement in Part III below.

Section 231,006 authorizes a state agency to terminate a contract if it determines that statement required below is inaccurate or false. In the event the statement is determined to be false, the vendor is liable to the state for attorney's fees, costs necessary to complete the contract [including the cost of advertising and awarding a second contract], and any other damages provided by taw or contract.

	11.
	d social security numbers of the individual identified in the a minimum 25% ownership interest in the business entity.  Social Security#
6-7-7-3-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1	
As required by Section 231,006, the undersigned certific	es the following:
business entity named in this contract, bid,	owledges that this contract may be terminated and
	President-Commercial & National Accounts
Scott Casebolt	8/24/20
Printed Name	Date

I (We) the undersigned, agent for the firm, named below certify that all information in the above certifications is true and correct to the best of my knowledge.

Name/Title: Scott Casebolt; President-Commercial Company Name: Key Construction, Inc.

Original and National Accounts

Signature: Date 8/24/20



### **CERTIFICATE OF LIABILITY INSURANCE**

11/1/2020

DATE (MM/DD/YYYY) 8/27/2020

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions or be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).

this certificate does not come rights to the certificate notice in	if fied of such endorsement(s).	
PRODUCER LOCKTON COMPANIES 444 W. 47TH STREET, SUITE 900	CONTACT NAME: PHONE (A/C, No, Ext): (A/C, No):	,
KANSAS CITY MO 64112-1906 (816) 960-9000	E-MAIL ADDRESS:	
(810) 900-9000	INSURER(S) AFFORDING COVERAGE	NAIC #
	INSURER A: LM Insurance Corporation	33600
INSURED KEY CONSTRUCTION, INC.	INSURER B: Liberty Mutual Fire Insurance Company	23035
741 WEST SECOND	INSURER C: XL Specialty Insurance Company	37885
WICHITA KS 67203	INSURER D: Liberty Insurance Corporation	42404
	INSURER E :	
	INSURER F:	
COVERAGES VEVCOOL CEPTIEICATE NUMBER.	16020452 PEVICION NUMBER. 37	3/3/3/3/3/

THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS.

INSR		DSIGNS AND CONDITIONS OF SUCH							
LTR		TYPE OF INSURANCE	INSD	SUBR WVD	POLICY NUMBER	POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIMITS	S
A	X	CLAIMS-MADE X OCCUR	Y	Y	TB5-Z91-468083-039	11/1/2019	11/1/2020	EACH OCCURRENCE DAMAGE TO RENTED PREMISES (Ea occurrence)	\$ 1,000,000 \$ 300,000
								MED EXP (Any one person)	\$ 10,000
								PERSONAL & ADV INJURY	\$ 1,000,000
	GEI	N'L AGGREGATE LIMIT APPLIES PER:						GENERAL AGGREGATE	\$ 2,000,000
		POLICY X PRO-						PRODUCTS - COMP/OP AGG	\$ 2,000,000
		OTHER:							\$
В	AU1	OMOBILE LIABILITY	Y	Y	AS2-Z91-468083-019	11/1/2019	11/1/2020	COMBINED SINGLE LIMIT (Ea accident)	\$ 1,000,000
	X	ANY AUTO						BODILY INJURY (Per person)	\$ XXXXXXX
		OWNED SCHEDULED AUTOS ONLY							\$ XXXXXXX
	X	AUTOS ONLY X NON-OWNED AUTOS ONLY	i					PROPERTY DAMAGE (Per accident)	\$ XXXXXXX
_									\$ XXXXXXX
C	X	UMBRELLA LIAB X OCCUR	Y	Y	US00095247LI19A	11/1/2019	11/1/2020	EACH OCCURRENCE	\$ 10,000,000
	<u> </u>	EXCESS LIAB CLAIMS-MADE	ŀ					AGGREGATE	\$ 10,000,000
		DED X RETENTION \$ 10,000							\$ XXXXXXX
D		KERS COMPENSATION EMPLOYERS' LIABILITY Y/N		Y	WC2-Z91-468083-029	11/1/2019	11/1/2020	X PER OTH-	
	ANY	PROPRIETOR/PARTNER/EXECUTIVE	N/A					E.L. EACH ACCIDENT	\$ 1,000,000
	(Man	datory in NH)						E.L. DISEASE - EA EMPLOYEE	\$ 1,000,000
	DES	CRIPTION OF OPERATIONS below						E.L. DISEASE - POLICY LIMIT	\$ 1,000,000
								'	
l .									

DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks Schedule, may be attached if more space is required)
RE: PROJECT #20-03-044, INTERNATIONAL LEADERSHIP OF TEXAS CHARTER SCHOOLS PORTABLE CLASSROOMS, 3610 LONGMIRE DRIVE, COLLEGE STATION, TX 77840, INTERNATIONAL LEADERSHIP OF TEXAS (OWNER) AND PIVOT NORTH ARCHITECTURE(THE ARCHITECT) ARE INCLUDED AS ADDITIONAL INSUREDS AS RESPECTS LIABILITY COVERAGE WHERE REQUIRED BY WRITTEN CONTRACT. A WAIVER OF SUBROGATION APPLIES IN FAVOR OF ADDITIONAL INSUREDS ON ALL POLICIES WHERE REQUIRED BY WRITTEN CONTRACT AND WHERE ALLOWED BY LAW. 30 DAYS NOTICE OF CANCELLATION APPLIES AS REQUIRED BY WRITTEN CONTRACT.

CERTIFICATE HOLDER	CANCELLATION
16929452 INTERNATIONAL LEADERSHIP OF TEXAS 1820 N. GLENVILLE DR., #100 RICHARDSON TX 75081	SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.
	AUTHORIZED REPRESENTATIVE!

© 1988 2015 ACORD CORPORATION. All rights reserved.

national American Education Federatio	n, Inc., d/b/a International Leadership of Tex	kas - August 27, 2020 Special Board Meeting	- Agenda - Thursda	ay August 27, 2020 at 6	3:15
			Г	DATE (MM/DD/YYYY)	
ACORD <sup>®</sup>	CERTIFICATE OF L	JABILITY INSURANCE		0.105.10000	

### CERTIFICATE OF LIABILITY INSURANCE

11/1/2020

DATE (MM/DD/YYYY) 8/27/2020

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder If SUBROGATION IS WAIVED, subject this certificate does not confer rights	t to t	he te	rms and conditions of th	e poli	cy, certain p	olicies may	NAL INSURED provision require an endorsemen	t. Ast	e endorsed. atement on
PRODUCER LOCKTON COMPANIES				CONTA		.,-			-
444 W. 47TH STREET, SUITE	900			PHONE (A/C, N			FAX (A/C, No):		
KANSAS CITY MO 64112-19	06			(A/C, N E-MAIL	o, Ext):		(A/C. No):		
(816) 960-9000				ADDRE					
, ,					_		RDING COVERAGE		NAIC #
				INSUR	RA: Traveler	s Property Ca	sualty Co of America		25674
INSURED KEY CONSTRUCTION, INC.				INSURE	RB:				
1072452 & SUBS & SUBCONTRACTO	RS			INSURE	RC:			i	
741 W. SECOND				INSURE	BD:			_	
WICHITA KS 67203				INSURE					
COVERAGES KEYCO01 CEF	TIE	CATI	NUMBER: 1692944	INSURE	HF:		DEVICION NUMBER.	VV	VVVVV
THIS IS TO CERTIFY THAT THE POLICIES					N ICCUED TO		REVISION NUMBER:		XXXXX
INDICATED. NOTWITHSTANDING ANY RI CERTIFICATE MAY BE ISSUED OR MAY EXCLUSIONS AND CONDITIONS OF SUCH	EQUIF PERT POLI	REME AIN,	NT, TERM OR CONDITION ( THE INSURANCE AFFORDE LIMITS SHOWN MAY HAVE (	OF AN' ED BY	Y CONTRACT THE POLICIES REDUCED BY (	OR OTHER I S DESCRIBEI PAID CLAIMS.	OCUMENT WITH RESPE	CT TO \	WHICH THIS
INSR LTR TYPE OF INSURANCE	INSD	WVD	POLICY NUMBER		POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIMIT		
COMMERCIAL GENERAL LIABILITY			NOT APPLICABLE				EACH OCCURRENCE	\$ XX	XXXXX
CLAIMS-MADE OCCUR	İ						DAMAGE TO RENTED PREMISES (Ea occurrence)	\$ XX	XXXXX
							MED EXP (Any one person)	\$ XX	XXXXX
	i						PERSONAL & ADV INJURY	\$ XX	XXXXX
GEN'L AGGREGATE LIMIT APPLIES PER:							GENERAL AGGREGATE		XXXXX
POLICY X PRO-							PRODUCTS - COMP/OP AGG		XXXXX
OTHER:							THODOOTO COMITTOT ACC	\$	7.7.7.7.7.7.
AUTOMOBILE LIABILITY			NOT APPLICABLE				COMBINED SINGLE LIMIT		XXXXX
ANY AUTO						-	(Ea accident) BODILY INJURY (Per person)		
OWNED SCHEDULED						ŀ			XXXXX
HIRED AUTOS NON-OWNED						-	PROPERTY DAMAGE		XXXXX
AUTOS ONLY AUTOS ONLY				i			(Per accident)		XXXXX
								\$ XX	XXXXX
UMBRELLA LIAB OCCUR			NOT APPLICABLE				EACH OCCURRENCE	\$ XX	XXXXX
EXCESS LIAB CLAIMS-MADE							AGGREGATE	\$ XX	XXXXX
DED RETENTION \$								\$ XX	XXXXX
WORKERS COMPENSATION AND EMPLOYERS' LIABILITY			NOT APPLICABLE				PER OTH- STATUTE ER		
ANY PROPRIETOR/PARTNER/EVECUTIVE T/N							E.L. EACH ACCIDENT	s XXX	XXXXX
OFFICER/MEMBER EXCLUDED? (Mandatory in NH)	N/A					Ì	E.L. DISEASE - EA EMPLOYEE		
If yes, describe under DESCRIPTION OF OPERATIONS below							E.L. DISEASE - POLICY LIMIT		XXXXX
A BUILDERS RISK	Y	N	QT6601J678860TIL19		11/1/2019	11/1/2020	SEE DESCRIPTION	Ψ ΛΛΩ	AAAAA
DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICI RE: PROJECT #20-03-044, INTERNATIONAL COLLEGE STATION, TX 77840, START DATE PIVOT NORTH ARCHITECTURE(THE ARCH	LEAI E: 8/31	DERS /2020	HIP OF TEXAS CHARTER S : CONTRACT AMOUNT: \$6	CHOO 3.559:	LS PORTABLI INTERNATIO	E CLASSROO	MS. 3610 LONGMIRE DR	IVE, R) AND	
CERTIFICATE HOLDER			·	CANO	ELI ATION				
			Г	CANC	ELLATION				
<b>16929446</b> INTERNATIONAL LEADERSH 1820 N. GLENVILLE DR.,#100 RICHARDSON TX 75081		F TE	EXAS	THE	<b>EXPIRATION</b>	DATE THE	SCRIBED POLICIES BE CAREOF, NOTICE WILL BY PROVISIONS.		
			ļ.	AUTHOR	IZED REPRESEN	ITATIVE!			
						- 71	M Agnello		

ACORD 25 (2016/03)

© 1988 2015 ACORD CORPORATION. All rights reserved.



## Project Billable Rates Exhibit E ILT College Station Portables

**Project Staff** 

1 Tojout Gtan	
Cost Engineer	\$53.75/hour
Estimator	\$68.75/hour
Precon Manager/VP of Precon	\$93.75/hour
Project Coordinator	\$37.50/hour
Project Engineer	\$53.75/hour
Assistant Project Manager	\$68.75/hour
Project Manager	\$87.50/hour
Senior Project Manager	\$93.75/hour
Field Engineer	\$53.75/hour
Assistant Superintendent	\$68.75/hour
Traveling Superintendent	\$97.50/hour
Project Executive/Division President	\$123.75/hour
Safety Director	\$90.00/hour
BIM/VDC Manager	\$90.00/hour
Laborer	\$37.50/hour
Junior Carpenter	\$43.75/hour
Carpenter	\$50.00/hour

<sup>\*\*</sup> The above rates are subject to 5% increase in 1st Quarter 2021

#### **Vehicles**

Superintendent/Assistant Truck	\$175/week

#### Insurance

	0.69% of total
General Liability/Project Insurance	contract amount

<sup>\*\*</sup> Does not include builder's risk, workers compensation, or any project specific policies

#### **Miscellaneous**

	1.0% of total contract
Performance & Payment Bonds	amount
	0.45% of total contract
Project Computers & Software (Tech)	amount
	0.10% of total contract
Safety Inspections	amount

741 WEST SECOND WICHITA KS 67203 3 1 6 - 2 6 3 - 9 5 1 5 FAX: 316-263-1161



### General Conditions of the Contract for Construction

#### for the following PROJECT:

(Name and location or address)

International Leadership of Texas Charter Schools Portable Classrooms 3610 Longmire Drive, College Station, TX 77840

#### THE OWNER:

(Name, legal status and address)

International Leadership of Texas 1820 N Glenville Dr. #100, Richardson, TX 75081 **THE ARCHITECT:** 

(Name, legal status and address)

Pivot North Architecture 1101 W Grove St Boise ID 83702

#### TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 OWNER
- 3 CONTRACTOR
- 4 ARCHITECT
- 5 SUBCONTRACTORS
- 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS
- 7 CHANGES IN THE WORK
- 8 TIME
- 9 PAYMENTS AND COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- 12 UNCOVERING AND CORRECTION OF WORK
- 13 MISCELLANEOUS PROVISIONS
- 14 TERMINATION OR SUSPENSION OF THE CONTRACT
- 15 CLAIMS AND DISPUTES

#### **ADDITIONS AND DELETIONS:**

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

For guidance in modifying this document to include supplementary conditions, see AIA Document A503™, Guide for Supplementary Conditions.

#### **INDEX** Architect's Authority to Reject Work (Topics and numbers in bold are Section headings.) 3.5, 4.2.6, 12.1.2, 12.2.1 Architect's Copyright 1.1.7, 1.5 Acceptance of Nonconforming Work Architect's Decisions 9.6.6, 9.9.3, **12.3** 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 6.3, Acceptance of Work 7.3.4, 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4.1, 9.5, 9.8.4, 9.9.1, 9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3, 12.3 13.4.2, 15.2 Access to Work Architect's Inspections **3.16**, 6.2.1, 12.1 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 13.4 **Accident Prevention** Architect's Instructions 3.2.4, 3.3.1, 4.2.6, 4.2.7, 13.4.2 **Acts and Omissions** Architect's Interpretations 3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 4.2.11, 4.2.12 Architect's Project Representative 10.2.8, 13.3.2, 14.1, 15.1.2, 15.2 Addenda 4.2.10 1.1.1 Architect's Relationship with Contractor Additional Costs, Claims for 1.1.2, 1.5, 2.3.3, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.7.4, 3.7.5, 10.3.2, 15.1.5 3.5, 3.7.4, 3.7.5, 3.9.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, **Additional Inspections and Testing** 3.18, 4.1.2, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.4.2, 9.8.3, 12.2.1, **13.4** 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3, 12, 13.3.2, 13.4, 15.2 Additional Time, Claims for Architect's Relationship with Subcontractors 1.1.2, 4.2.3, 4.2.4, 4.2.6, 9.6.3, 9.6.4, 11.33.2.4, 3.7.4, 3.7.5, 3.10.2, 8.3.2, **15.1.6 Administration of the Contract** Architect's Representations 3.1.3, **4.2**, 9.4, 9.5 9.4.2, 9.5.1, 9.10.1 Advertisement or Invitation to Bid Architect's Site Visits 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.4 1.1.1 Aesthetic Effect Asbestos 4.2.13 10.3.1 Allowances Attorneys' Fees 3.18.1, 9.6.8, 9.10.2, 10.3.3 Award of Separate Contracts **Applications for Payment** 4.2.5, 7.3.9, 9.2, **9.3**, 9.4, 9.5.1, 9.5.4, 9.6.3, 9.7, 9.10 6.1.1, 6.1.2 Award of Subcontracts and Other Contracts for 2.1.1, 2.3.1, 2.5, 3.1.3, 3.10.2, 3.12.8, 3.12.9, Portions of the Work 3.12.10.1, 4.2.7, 9.3.2, 13.4.1 5.2 **Basic Definitions** Arbitration 8.3.1, 15.3.2, **15.4** 1.1 **ARCHITECT** Bidding Requirements 1.1.1 Architect, Definition of Binding Dispute Resolution 4.1.1 8.3.1, 9.7, 11.5, 13.1, 15.1.2, 15.1.3, 15.2.1, 15.2.5, Architect, Extent of Authority 15.2.6.1, 15.3.1, 15.3.2, 15.3.3, 15.4.1 2.5, 3.12.7, 4.1.2, 4.2, 5.2, 6.3, 7.1.2, 7.3.4, 7.4, 9.2, Bonds, Lien 9.3.1, 9.4, 9.5, 9.6.3, 9.8, 9.10.1, 9.10.3, 12.1, 12.2.1, 7.3.4.4, 9.6.8, 9.10.2, 9.10.3 13.4.1, 13.4.2, 14.2.2, 14.2.4, 15.1.4, 15.2.1 Bonds, Performance, and Payment Architect, Limitations of Authority and Responsibility 7.3.4.4, 9.6.7, 9.10.3, **11.1.2**, 11.1.3, **11.5** 2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2, 4.2.3, **Building Information Models Use and Reliance** 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, 7.4, 9.4.2, 1.8 **Building Permit** 9.5.4, 9.6.4, 15.1.4, 15.2 Architect's Additional Services and Expenses 3.7.1 2.5, 12.2.1, 13.4.2, 13.4.3, 14.2.4 Capitalization Architect's Administration of the Contract 3.1.3, 3.7.4, 15.2, 9.4.1, 9.5 Certificate of Substantial Completion

Init.

Architect's Approvals 2.5, 3.1.3, 3.5, 3.10.2, 4.2.7

AIA Document A201® – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents registered trademarks and may not be used without permission. This document was produced by AIA software at 16:21:36 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. **User Notes:** 

9.8.3, 9.8.4, 9.8.5

59 of 99

#### **Certificates for Payment**

4.2.1, 4.2.5, 4.2.9, 9.3.3, **9.4**, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.4

Certificates of Inspection, Testing or Approval 13.4.4

Certificates of Insurance 9.10.2

#### **Change Orders**

1.1.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, **7.2**, 7.3.2, 7.3.7, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.2, 11.5, 12.1.2

Change Orders, Definition of

7.2.1

#### **CHANGES IN THE WORK**

2.2.2, 3.11, 4.2.8, **7**, 7.2.1, 7.3.1, 7.4, 8.3.1, 9.3.1.1, 11.5

Claims, Definition of

15.1.1

Claims, Notice of

1.6.2, 15.1.3

#### **CLAIMS AND DISPUTES**

3.2.4, 6.1.1, 6.3, 7.3.9, 9.3.3, 9.10.4, 10.3.3, **15**, 15.4 Claims and Timely Assertion of Claims 15.4.1

#### **Claims for Additional Cost**

3.2.4, 3.3.1, 3.7.4, 7.3.9, 9.5.2, 10.2.5, 10.3.2, **15.1.5** 

#### **Claims for Additional Time**

3.2.4, 3.3.1, 3.7.4, 6.1.1, 8.3.2, 9.5.2, 10.3.2, **15.1.6** 

### Concealed or Unknown Conditions, Claims for 3.7.4

Claims for Damages

3.2.4, 3.18, 8.3.3, 9.5.1, 9.6.7, 10.2.5, 10.3.3, 11.3, 11.3.2, 14.2.4, 15.1.7

Claims Subject to Arbitration

15.4.1

#### Cleaning Up

**3.15**, 6.3

Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.2, **15.1.5** 

Commencement of the Work, Definition of 8.1.2

#### **Communications**

3.9.1, 4.2.4

Completion, Conditions Relating to

3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 14.1.2, 15.1.2

#### **COMPLETION, PAYMENTS AND**

9

Completion, Substantial

3.10.1, 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, 9.8, 9.9.1, 9.10.3, 12.2, 15.1.2

Compliance with Laws

2.3.2, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 9.6.4, 10.2.2, 13.1, 13.3, 13.4.1, 13.4.2, 13.5, 14.1.1, 14.2.1.3, 15.2.8, 15.4.2, 15.4.3

Concealed or Unknown Conditions

3.7.4, 4.2.8, 8.3.1, 10.3

Conditions of the Contract

1.1.1, 6.1.1, 6.1.4

Consent, Written

3.4.2, 3.14.2, 4.1.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 13.2,

15.4.4.2

#### **Consolidation or Joinder**

15.4.4

## CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

1146

## **Construction Change Directive**, Definition of 7.3.1

#### **Construction Change Directives**

1.1.1, 3.4.2, 3.11, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, **7.3**, 9.3.1.1

Construction Schedules, Contractor's 3.10, 3.11, 3.12.1, 3.12.2, 6.1.3, 15.1.6.2

### **Contingent Assignment of Subcontracts**

**5.4**, 14.2.2.2

#### **Continuing Contract Performance**

15.1.4

Contract, Definition of

1.1.2

## CONTRACT, TERMINATION OR SUSPENSION OF THE

5.4.1.1, 5.4.2, 11.5, **14** 

**Contract Administration** 

3.1.3, 4, 9.4, 9.5

Contract Award and Execution, Conditions Relating

3.7.1, 3.10, 5.2, 6.1

Contract Documents, Copies Furnished and Use of 1.5.2, 2.3.6, 5.3

Contract Documents, Definition of

1.1.1

#### **Contract Sum**

2.2.2, 2.2.4, 3.7.4, 3.7.5, 3.8, 3.10.2, 5.2.3, 7.3, 7.4, **9.1**, 9.2, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.5, 12.1.2, 12.3, 14.2.4, 14.3.2, 15.1.4.2, **15.1.5, 15.2.5** 

Contract Sum, Definition of

9.1

#### Contract Time

1.1.4, 2.2.1, 2.2.2, 3.7.4, 3.7.5, 3.10.2, 5.2.3, 6.1.5, 7.2.1.3, 7.3.1, 7.3.5, 7.3.6, 7, 7, 7.3.10, 7.4, 8.1.1, 8.2.1, 8.2.3, 8.3.1, 9.5.1, 9.7, 10.3.2, 12.1.1, 12.1.2, 14.3.2, 15.1.4.2, 15.1.6.1, 15.2.5

Contract Time, Definition of

8.1.1

#### **CONTRACTOR**

3

Contractor, Definition of

3.1, 6.1.2

## Contractor's Construction and Submittal Schedules

**3.10**, 3.12.1, 3.12.2, 4.2.3, 6.1.3, 15.1.6.2

#### Init.

3

```
Damage to Construction of Owner or Separate
Contractor's Employees
2.2.4, 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2,
                                                                 Contractors
10.3, 11.3, 14.1, 14.2.1.1
                                                                 3.14.2, 6.2.4, 10.2.1.2, 10.2.5, 10.4, 12.2.4
Contractor's Liability Insurance
                                                                Damage to the Work
                                                                 3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4, 12.2.4
11.1
Contractor's Relationship with Separate Contractors
                                                                Damages, Claims for
                                                                 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.3.2,
and Owner's Forces
3.12.5, 3.14.2, 4.2.4, 6, 11.3, 12.2.4
                                                                 11.3, 14.2.4, 15.1.7
Contractor's Relationship with Subcontractors
                                                                Damages for Delay
1.2.2, 2.2.4, 3.3.2, 3.18.1, 3.18.2, 4.2.4, 5, 9.6.2, 9.6.7,
                                                                 6.2.3, 8.3.3, 9.5.1.6, 9.7, 10.3.2, 14.3.2
9.10.2, 11.2, 11.3, 11.4
                                                                Date of Commencement of the Work, Definition of
Contractor's Relationship with the Architect
1.1.2, 1.5, 2.3.3, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2,
                                                                Date of Substantial Completion, Definition of
3.5.1, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.2, 5.2, 6.2.2,
                                                                8.1.3
7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3,
                                                                 Day, Definition of
                                                                8.1.4
11.3, 12, 13.4, 15.1.3, 15.2.1
Contractor's Representations
                                                                Decisions of the Architect
3.2.1, 3.2.2, 3.5, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2
                                                                 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 6.3, 7.3.4,
                                                                 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4, 9.5.1, 9.8.4, 9.9.1, 13.4.2,
Contractor's Responsibility for Those Performing the
Work
                                                                 14.2.2, 14.2.4, 15.1, 15.2
3.3.2, 3.18, 5.3, 6.1.3, 6.2, 9.5.1, 10.2.8
                                                                Decisions to Withhold Certification
Contractor's Review of Contract Documents
                                                                 9.4.1, 9.5, 9.7, 14.1.1.3
                                                                 Defective or Nonconforming Work, Acceptance,
Contractor's Right to Stop the Work
                                                                 Rejection and Correction of
2.2.2, 9.7
                                                                2.5, 3.5, 4.2.6, 6.2.3, 9.5.1, 9.5.3, 9.6.6, 9.8.2, 9.9.3,
Contractor's Right to Terminate the Contract
                                                                9.10.4, 12.2.1
14.1
                                                                Definitions
Contractor's Submittals
                                                                 1.1, 2.1.1, 3.1.1, 3.5, 3.12.1, 3.12.2, 3.12.3, 4.1.1, 5.1,
3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 9.2, 9.3, 9.8.2,
                                                                 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1, 15.1.1
9.8.3, 9.9.1, 9.10.2, 9.10.3
                                                                Delays and Extensions of Time
Contractor's Superintendent
                                                                3.2, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, 8.3, 9.5.1, 9.7,
3.9, 10.2.6
                                                                 10.3.2, 10.4, 14.3.2, 15.1.6, 15.2.5
                                                                Digital Data Use and Transmission
Contractor's Supervision and Construction
Procedures
                                                                 1.7
1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3,
                                                                 Disputes
7.3.4, 7.3.6, 8.2, 10, 12, 14, 15.1.4
                                                                 6.3, 7.3.9, 15.1, 15.2
Coordination and Correlation
                                                                 Documents and Samples at the Site
1.2, 3.2.1, 3.3.1, 3.10, 3.12.6, 6.1.3, 6.2.1
                                                                3.11
Copies Furnished of Drawings and Specifications
                                                                Drawings, Definition of
1.5, 2.3.6, 3.11
                                                                 1.1.5
Copyrights
                                                                Drawings and Specifications, Use and Ownership of
1.5, 3.17
Correction of Work
                                                                Effective Date of Insurance
2.5, 3.7.3, 9.4.2, 9.8.2, 9.8.3, 9.9.1, 12.1.2, 12.2, 12.3,
                                                                8.2.2
15.1.3.1, 15.1.3.2, 15.2.1
                                                                 Emergencies
Correlation and Intent of the Contract Documents
                                                                 10.4, 14.1.1.2, 15.1.5
1.2
                                                                 Employees, Contractor's
Cost, Definition of
                                                                 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2,
7.3.4
                                                                 10.3.3, 11.3, 14.1, 14.2.1.1
Costs
                                                                 Equipment, Labor, or Materials
                                                                 1.1.3, 1.1.6, 3.4, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1,
2.5, 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3,
7.3.3.3, 7.3.4, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6, 11.2,
                                                                 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.4, 9.3.2, 9.3.3, 9.5.1.3,
12.1.2, 12.2.1, 12.2.4, 13.4, 14
                                                                 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2
Cutting and Patching
                                                                Execution and Progress of the Work
3.14, 6.2.5
                                                                 1.1.3, 1.2.1, 1.2.2, 2.3.4, 2.3.6, 3.1, 3.3.1, 3.4.1, 3.7.1,
                                                                 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.6, 8.2, 9.5.1,
                                                                9.9.1, 10.2, 10.3, 12.1, 12.2, 14.2, 14.3.1, 15.1.4
```

Init.

AIA Document A201® – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:21:36 ET on 08/25/2020 under Order No. 1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

User Notes:

**Extensions of Time** Insurance, Stored Materials 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 7.4, 9.5.1, 9.7, 10.3.2, 9.3.2 INSURANCE AND BONDS 10.4, 14.3, 15.1.6, **15.2.5 Failure of Payment** 9.5.1.3, **9.7**, 9.10.2, 13.5, 14.1.1.3, 14.2.1.2 Insurance Companies, Consent to Partial Occupancy Faulty Work (See Defective or Nonconforming Work) Insured loss, Adjustment and Settlement of **Final Completion and Final Payment** 4.2.1, 4.2.9, 9.8.2, **9.10**, 12.3, 14.2.4, 14.4.3 Intent of the Contract Documents Financial Arrangements, Owner's 1.2.1, 4.2.7, 4.2.12, 4.2.13 2.2.1, 13.2.2, 14.1.1.4 Interest **GENERAL PROVISIONS** 13.5 Interpretation 1.1.8, 1.2.3, **1.4**, 4.1.1, 5.1, 6.1.2, 15.1.1 **Governing Law** 13.1 Interpretations, Written 4.2.11, 4.2.12 Guarantees (See Warranty) **Hazardous Materials and Substances** Judgment on Final Award 10.2.4, **10.3** 15.4.2 Identification of Subcontractors and Suppliers Labor and Materials, Equipment 5.2.1 1.1.3, 1.1.6, **3.4**, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, Indemnification 5.2.1, 6.2.1, 7.3.4, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 3.17, **3.18**, 9.6.8, 9.10.2, 10.3.3, 11.3 10.2.4, 14.2.1.1, 14.2.1.2 Information and Services Required of the Owner Labor Disputes 2.1.2, **2.2**, 2.3, 3.2.2, 3.12.10.1, 6.1.3, 6.1.4, 6.2.5, 8.3.1 9.6.1, 9.9.2, 9.10.3, 10.3.3, 11.2, 13.4.1, 13.4.2, Laws and Regulations 1.5, 2.3.2, 3.2.3, 3.2.4, 3.6, 3.7, 3.12.10, 3.13, 9.6.4, 14.1.1.4, 14.1.4, 15.1.4 **Initial Decision** 9.9.1, 10.2.2, 13.1, 13.3.1, 13.4.2, 13.5, 14, 15.2.8, 15.2 15.4 **Initial Decision Maker, Definition of** Liens 1.1.8 2.1.2, 9.3.1, 9.3.3, 9.6.8, 9.10.2, 9.10.4, 15.2.8 Initial Decision Maker, Decisions Limitations, Statutes of 14.2.4, 15.1.4.2, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 12.2.5, 15.1.2, 15.4.1.1 Initial Decision Maker, Extent of Authority Limitations of Liability 14.2.4, 15.1.4.2, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 3.2.2, 3.5, 3.12.10, 3.12.10.1, 3.17, 3.18.1, 4.2.6, Injury or Damage to Person or Property 4.2.7, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 9.6.8, 10.2.5, 10.3.3, 10.2.8, 10.4 11.3, 12.2.5, 13.3.1 Inspections Limitations of Time 3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 2.1.2, 2.2, 2.5, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7, 5.2, 5.3, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.9.2, 9.10.1, 12.2.1, 13.4 Instructions to Bidders 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 12.2, 13.4, 14, 15, 1.1.1 15.1.2, 15.1.3, 15.1.5 Materials, Hazardous Instructions to the Contractor 3.2.4, 3.3.1, 3.8.1, 5.2.1, 7, 8.2.2, 12, 13.4.2 10.2.4, 10.3 Instruments of Service, Definition of Materials, Labor, Equipment and 1.1.7 1.1.3, 1.1.6, 3.4.1, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 5.2.1, 6.2.1, 7.3.4, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, Insurance 6.1.1, 7.3.4, 8.2.2, 9.3.2, 9.8.4, 9.9.1, 9.10.2, 10.2.5, 11 10.2.1.2, 10.2.4, 14.2.1.1, 14.2.1.2 Insurance, Notice of Cancellation or Expiration Means, Methods, Techniques, Sequences and 11.1.4, 11.2.3 Procedures of Construction Insurance, Contractor's Liability 3.3.1, 3.12.10, 4.2.2, 4.2.7, 9.4.2 11.1 Mechanic's Lien Insurance, Effective Date of 2.1.2, 9.3.1, 9.3.3, 9.6.8, 9.10.2, 9.10.4, 15.2.8 8.2.2, 14.4.2 Mediation Insurance, Owner's Liability 8.3.1, 15.1.3.2, 15.2.1, 15.2.5, 15.2.6, **15.3**, 15.4.1, 15.4.1.1 11.2 **Insurance, Property** Minor Changes in the Work **10.2.5**, 11.2, 11.4, 11.5 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1, **7.4** 

Init.

**User Notes:** 

AIA Document A201® – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:21:36 ET on 08/25/2020 under Order No.1612218839 which expires on 11/27/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

(1986546025)

#### MISCELLANEOUS PROVISIONS

13

Modifications, Definition of

1.1.1

Modifications to the Contract

1.1.1, 1.1.2, 2.5, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7, 10.3.2

**Mutual Responsibility** 

6.2

Nonconforming Work, Acceptance of

9.6.6, 9.9.3, 12.3

Nonconforming Work, Rejection and Correction of 2.4, 2.5, 3.5, 4.2.6, 6.2.4, 9.5.1, 9.8.2, 9.9.3, 9.10.4, 12.2

Notice

**1.6**, 1.6.1, 1.6.2, 2.1.2, 2.2.2., 2.2.3, 2.2.4, 2.5, 3.2.4, 3.3.1, 3.7.4, 3.7.5, 3.9.2, 3.12.9, 3.12.10, 5.2.1, 7.4, 8.2.2 9.6.8, 9.7, 9.10.1, 10.2.8, 10.3.2, 11.5, 12.2.2.1, 13.4.1, 13.4.2, 14.1, 14.2.2, 14.4.2, 15.1.3, 15.1.5, 15.1.6, 15.4.1

Notice of Cancellation or Expiration of Insurance 11.1.4, 11.2.3

**Notice of Claims** 

1.6.2, 2.1.2, 3.7.4, 9.6.8, 10.2.8, **15.1.3**, 15.1.5, 15.1.6, 15.2.8, 15.3.2, 15.4.1

Notice of Testing and Inspections

13.4.1, 13.4.2

Observations, Contractor's

3.2, 3.7.4

Occupancy

2.3.1, 9.6.6, 9.8

Orders, Written

1.1.1, 2.4, 3.9.2, 7, 8.2.2, 11.5, 12.1, 12.2.2.1, 13.4.2, 14.3.1

**OWNER** 

2

Owner, Definition of

2.1.1

Owner, Evidence of Financial Arrangements

**2.2**, 13.2.2, 14.1.1.4

Owner, Information and Services Required of the 2.1.2, **2.2**, 2.3, 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 13.4.1, 13.4.2, 14.1.4, 14.1.4, 15.1.4

Owner's Authority

1.5, 2.1.1, 2.3.32.4, 2.5, 3.4.2, 3.8.1, 3.12.10, 3.14.2, 4.1.2, 4.2.4, 4.2.9, 5.2.1, 5.2.4, 5.4.1, 6.1, 6.3, 7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.2, 9.5.1, 9.6.4, 9.9.1, 9.10.2, 10.3.2, 11.4, 11.5, 12.2.2, 12.3, 13.2.2, 14.3, 14.4, 15.2.7

**Owner's Insurance** 

11.2

Owner's Relationship with Subcontractors 1.1.2, 5.2, 5.3, 5.4, 9.6.4, 9.10.2, 14.2.2

Owner's Right to Carry Out the Work 2.5, 14.2.2

Owner's Right to Clean Up

6.3

Owner's Right to Perform Construction and to

**Award Separate Contracts** 

6.1

Owner's Right to Stop the Work

2.4

Owner's Right to Suspend the Work

14.3

Owner's Right to Terminate the Contract

14.2, 14.4

Ownership and Use of Drawings, Specifications

and Other Instruments of Service

1.1.1, 1.1.6, 1.1.7, **1.5**, 2.3.6, 3.2.2, 3.11, 3.17, 4.2.12, 5.3

**Partial Occupancy or Use** 

9.6.6, **9.9** 

Patching, Cutting and

**3.14**, 6.2.5 Patents

1 atcii

3.17

Payment, Applications for

4.2.5, 7.3.9, 9.2, **9.3**, 9.4, 9.5, 9.6.3, 9.7, 9.8.5, 9.10.1,

14.2.3, 14.2.4, 14.4.3

**Payment, Certificates for** 

4.2.5, 4.2.9, 9.3.3, **9.4**, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1,

9.10.3, 14.1.1.3, 14.2.4

Payment, Failure of

9.5.1.3, **9.7**, 9.10.2, 13.5, 14.1.1.3, 14.2.1.2

Payment, Final

4.2.1, 4.2.9, **9.10**, 12.3, 14.2.4, 14.4.3

Payment Bond, Performance Bond and

7.3.4.4, 9.6.7, 9.10.3, **11.1.2** 

Payments, Progress

9.3, **9.6**, 9.8.5, 9.10.3, 14.2.3, 15.1.4

PAYMENTS AND COMPLETION

9

Payments to Subcontractors

5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, 14.2.1.2

PCB

10.3.1

**Performance Bond and Payment Bond** 

7.3.4.4, 9.6.7, 9.10.3, **11.1.2** 

Permits, Fees, Notices and Compliance with Laws

2.3.1, **3.7**, 3.13, 7.3.4.4, 10.2.2

PERSONS AND PROPERTY, PROTECTION OF

10

Polychlorinated Biphenyl

10.3.1

Product Data, Definition of

3.12.2

**Product Data and Samples, Shop Drawings** 

3.11, **3.12**, 4.2.7

**Progress and Completion** 

4.2.2, **8.2**, 9.8, 9.9.1, 14.1.4, 15.1.4

**Progress Payments** 

9.3, **9.6**, 9.8.5, 9.10.3, 14.2.3, 15.1.4

6

Project, Definition of Separate Contracts and Contractors 1.1.4, 3.12.5, 3.14.2, 4.2.4, 4.2.7, 6, 8.3.1, 12.1.2 1.1.4 **Project Representatives** Separate Contractors, Definition of 4.2.10 6.1.1 **Property Insurance** Shop Drawings, Definition of 10.2.5, **11.2** 3.12.1 **Proposal Requirements Shop Drawings, Product Data and Samples** 3.11, **3.12**, 4.2.7 PROTECTION OF PERSONS AND PROPERTY Site, Use of 10 **3.13**, 6.1.1, 6.2.1 Regulations and Laws Site Inspections 1.5, 2.3.2, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 9.6.4, 9.9.1, 3.2.2, 3.3.3, 3.7.1, 3.7.4, 4.2, 9.9.2, 9.4.2, 9.10.1, 13.4 10.2.2, 13.1, 13.3, 13.4.1, 13.4.2, 13.5, 14, 15.2.8, 15.4 Site Visits, Architect's Rejection of Work 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.4 4.2.6, 12.2.1 Special Inspections and Testing Releases and Waivers of Liens 4.2.6, 12.2.1, 13.4 9.3.1, 9.10.2 Specifications, Definition of Representations 1.1.6 3.2.1, 3.5, 3.12.6, 8.2.1, 9.3.3, 9.4.2, 9.5.1, 9.10.1**Specifications** Representatives 1.1.1, **1.1.6**, 1.2.2, 1.5, 3.12.10, 3.17, 4.2.14 2.1.1, 3.1.1, 3.9, 4.1.1, 4.2.10, 13.2.1Statute of Limitations Responsibility for Those Performing the Work 15.1.2, 15.4.1.1 3.3.2, 3.18, 4.2.2, 4.2.3, 5.3, 6.1.3, 6.2, 6.3, 9.5.1, 10 Stopping the Work 2.2.2, 2.4, 9.7, 10.3, 14.1 9.3.1, 9.6.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3 Stored Materials **Review of Contract Documents and Field** 6.2.1, 9.3.2, 10.2.1.2, 10.2.4 **Conditions by Contractor** Subcontractor, Definition of **3.2**, 3.12.7, 6.1.3 5.1.1 Review of Contractor's Submittals by Owner and **SUBCONTRACTORS** Architect 3.10.1, 3.10.2, 3.11, 3.12, 4.2, 5.2, 6.1.3, 9.2, 9.8.2 Subcontractors, Work by Review of Shop Drawings, Product Data and Samples 1.2.2, 3.3.2, 3.12.1, 3.18, 4.2.3, 5.2.3, 5.3, 5.4, 9.3.1.2, by Contractor 3.12 **Subcontractual Relations Rights and Remedies 5.3**, 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, 14.1, 14.2.1 1.1.2, 2.4, 2.5, 3.5, 3.7.4, 3.15.2, 4.2.6, 5.3, 5.4, 6.1, Submittals 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.1, 12.2.2, 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 7.3.4, 9.2, 9.3, 9.8, 9.9.1, 9.10.2, 9.10.3 12.2.4, **13.3**, 14, 15.4 **Royalties, Patents and Copyrights** Submittal Schedule 3.17 3.10.2, 3.12.5, 4.2.7 Rules and Notices for Arbitration Subrogation, Waivers of 15.4.1 6.1.1, **11.3** Safety of Persons and Property Substances, Hazardous 10.2, 10.4 10.3 **Safety Precautions and Programs Substantial Completion** 3.3.1, 4.2.2, 4.2.7, 5.3, **10.1**, 10.2, 10.4 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, **9.8**, 9.9.1, 9.10.3, 12.2, Samples, Definition of 15.1.2 Substantial Completion, Definition of Samples, Shop Drawings, Product Data and 9.8.1 3.11, **3.12**, 4.2.7 Substitution of Subcontractors Samples at the Site, Documents and 5.2.3, 5.2.4 Substitution of Architect 3.11 Schedule of Values 2.3.3 **9.2**, 9.3.1 Substitutions of Materials Schedules, Construction 3.4.2, 3.5, 7.3.8 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.6.2 Sub-subcontractor, Definition of

Init.

**User Notes:** 

(1986546025)

**Subsurface Conditions** 

3.7.4

**Successors and Assigns** 

13.2

**Superintendent** 

**3.9**, 10.2.6

**Supervision and Construction Procedures** 

1.2.2, **3.3**, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3,

7.3.4, 8.2, 8.3.1, 9.4.2, 10, 12, 14, 15.1.4

Suppliers

1.5, 3.12.1, 4.2.4, 4.2.6, 5.2.1, 9.3, 9.4.2, 9.5.4, 9.6,

9.10.5, 14.2.1

Surety

5.4.1.2, 9.6.8, 9.8.5, 9.10.2, 9.10.3, 11.1.2, 14.2.2,

15.2.7

Surety, Consent of

9.8.5, 9.10.2, 9.10.3

Surveys

1.1.7, 2.3.4

Suspension by the Owner for Convenience

Suspension of the Work

3.7.5, 5.4.2, 14.3

Suspension or Termination of the Contract

5.4.1.1, 14

**Taxes** 

3.6, 3.8.2.1, 7.3.4.4

**Termination by the Contractor** 

**14.1**, 15.1.7

**Termination by the Owner for Cause** 

5.4.1.1, 14.2, 15.1.7

Termination by the Owner for Convenience

Termination of the Architect

Termination of the Contractor Employment

14.2.2

TERMINATION OR SUSPENSION OF THE CONTRACT

14

**Tests and Inspections** 

3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3,

9.9.2, 9.10.1, 10.3.2, 12.2.1, **13.4** 

TIME

Time, Delays and Extensions of

3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, **8.3**, 9.5.1, 9.7,

10.3.2, 10.4, 14.3.2, 15.1.6, 15.2.5

Time Limits

2.1.2, 2.2, 2.5, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2,

5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1,

9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 12.2, 13.4, 14, 15.1.2,

15.1.3, 15.4

**Time Limits on Claims** 

3.7.4, 10.2.8, 15.1.2, 15.1.3

Title to Work

9.3.2, 9.3.3

UNCOVERING AND CORRECTION OF WORK

**Uncovering of Work** 

12.1

Unforeseen Conditions, Concealed or Unknown

3.7.4, 8.3.1, 10.3

**Unit Prices** 

7.3.3.2, 9.1.2

Use of Documents

1.1.1, 1.5, 2.3.6, 3.12.6, 5.3

Use of Site

**3.13**, 6.1.1, 6.2.1

Values, Schedule of

**9.2**, 9.3.1

Waiver of Claims by the Architect

13.3.2

Waiver of Claims by the Contractor

9.10.5, 13.3.2, 15.1.7

Waiver of Claims by the Owner

9.9.3, 9.10.3, 9.10.4, 12.2.2.1, 13.3.2, 14.2.4, **15.1.7** 

Waiver of Consequential Damages

14.2.4, 15.1.7

Waiver of Liens

9.3, 9.10.2, 9.10.4

Waivers of Subrogation

6.1.1, **11.3** 

Warranty

**3.5**, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.2, 9.10.4, 12.2.2,

15.1.2

Weather Delays

8.3, 15.1.6.2

Work, Definition of

1.1.3

Written Consent

1.5.2, 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.10.3,

13.2, 13.3.2, 15.4.4.2

Written Interpretations

4.2.11, 4.2.12

Written Orders

1.1.1, 2.4, 3.9, 7, 8.2.2, 12.1, 12.2, 13.4.2, 14.3.1

8

#### ARTICLE 1 GENERAL PROVISIONS

#### § 1.1 Basic Definitions

#### § 1.1.1 The Contract Documents

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement, and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive, or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding or proposal requirements.

#### § 1.1.2 The Contract

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect's consultants, or (4) between any persons or entities other than the Owner and the Contractor. Contractor shall, however, perform the obligations under the Contract Documents intended to facilitate performance of the Architect's duties.

#### § 1.1.3 The Work

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment, and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project. Where the basis of payment is the cost of the Work, wherever the Contract Documents require Contractor to furnish, supply, pay for, or like terms are used, it is understood that such costs shall be included as a Cost of the Work as the same is defined in the Agreement for which the Contractor shall be paid by Owner, unless otherwise specifically provided.

#### § 1.1.4 The Project

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by Separate Contractors.

#### § 1.1.5 The Drawings

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules, and diagrams.

#### § 1.1.6 The Specifications

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

#### § 1.1.7 Instruments of Service

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

#### § 1.1.8 Initial Decision Maker

The Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2. The Initial Decision Maker shall not show partiality to the Owner or Contractor and shall not be liable for results of interpretations or decisions rendered in good faith.

#### § 1.2 Correlation and Intent of the Contract Documents

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as

binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

- § 1.2.1.1 The invalidity of any provision of the Contract Documents shall not invalidate the Contract or its remaining provisions. If it is determined that any provision of the Contract Documents violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Contract Documents shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Contract.
- § 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.
- § 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.

#### § 1.3 Capitalization

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles, or (3) the titles of other documents published by the American Institute of Architects.

#### § 1.4 Interpretation

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

#### § 1.5 Ownership and Use of Drawings, Specifications, and Other Instruments of Service

- § 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and retain all common law, statutory, and other reserved rights in their Instruments of Service, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with the Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights. The Owner shall be given an irrevocable, royalty free license to the Drawings, Specifications and other Instruments of Service for the purposes of any future renovations, adjustments or any other purpose necessary to Owner's interest in the Work.
- § 1.5.2 The Contractor, Subcontractors, Sub-subcontractors, and suppliers are authorized to use and reproduce the Instruments of Service provided to them, subject to any protocols established pursuant to Sections 1.7 and 1.8, solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and suppliers may not use the Instruments of Service on other projects or for additions to the Project outside the scope of the Work without the specific written consent of the Owner, Architect, and the Architect's consultants.

#### § 1.6 Notice

- § 1.6.1 Except as otherwise provided in Section 1.6.2, where the Contract Documents require one party to notify or give notice to the other party, such notice shall be provided in writing to the designated representative of the party to whom the notice is addressed and shall be deemed to have been duly served if delivered in person, by mail, by courier, or by electronic transmission if a method for electronic transmission is set forth in the Agreement.
- § 1.6.2 Notice of Claims as provided in Section 15.1.3 shall be provided in writing and shall be deemed to have been duly served only if delivered to the designated representative of the party to whom the notice is addressed by certified or registered mail, or by courier providing proof of delivery.

#### § 1.7 Digital Data Use and Transmission

The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form.

#### § 1.8 Building Information Models Use and Reliance

(Paragraph deleted)

ARTICLE 2 OWNER

§ 2.1 General

§ 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. All parties understand that only the Board of Directors for the Owner acting as a body corporate has the authority to bind the Owner with respect to all matters requiring the Board's approval under current policy of the Board of Directors for the Owner, including, but not limited to, Change Orders, Except as otherwise provided in Section 4.2.1 the Architect does not have such authority to bind the Owner with respect to matters requiring the Owner's approval or authorization. The term "Owner" means the Owner or the Owner's authorized representative.

#### (Paragraph deleted)

#### § 2.2 Evidence of the Owner's Financial Arrangements

- § 2.2.1 Prior to commencement of the Work and upon written request by the Contractor, the Owner shall furnish to the Contractor reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. The Contractor shall have no obligation to commence the Work until the Owner provides such evidence. If commencement of the Work is delayed under this Section 2.2.1, the Contract Time shall be extended appropriately.
- § 2.2.2 Following commencement of the Work and upon written request by the Contractor, the Owner shall furnish to the Contractor reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract only if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due; or (3) a change in the Work materially changes the Contract Sum. If the Owner fails to provide such evidence, as required, within fourteen days of the Contractor's request, the Contractor may immediately stop the Work and, in that event, shall notify the Owner that the Work has stopped. However, if the request is made because a change in the Work materially changes the Contract Sum under (3) above, the Contractor may immediately stop only that portion of the Work affected by the change until reasonable evidence is provided. If the Work is stopped under this Section 2.2.2, the Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided in the Contract Documents.
- **§ 2.2.3** After the Owner furnishes evidence of financial arrangements under this Section 2.2, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.
- § 2.2.4 Where the Owner has designated information furnished under this Section 2.2 as "confidential," the Contractor shall keep the information confidential and shall not disclose it to any other person. However, the Contractor may disclose "confidential" information, after seven (7) days' notice to the Owner, where disclosure is required by law, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or by court or arbitrator(s) order. The Contractor may also disclose "confidential" information to its employees, consultants, sureties, Subcontractors and their employees, Sub-subcontractors, and others who need to know the content of such information solely and exclusively for the Project and who agree to maintain the confidentiality of such information.

#### § 2.3 Information and Services Required of the Owner

- § 2.3.1 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.
- § 2.3.2 The Owner shall retain an architect lawfully licensed to practice architecture, or an entity lawfully practicing architecture, in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be entitled to rely on the sufficiency of the Contract Documents prepared by the Architect and its consultants and other information provided to the Contractor by the Owner or its separate consultants. Should Contractor notice or determine what Contractor to be errors in the Contract Documents prepared by the Architect. Contract shall notify the Architect and Owner, as soon as practicable, of such errors.

- § 2.3.3 If the employment of the Architect terminates, the Owner shall employ a successor to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.
- § 2.3.4 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.
- § 2.3.5 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.
- § 2.3.6 Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.

#### § 2.4 Owner's Right to Stop the Work

If the Contractor fails to commence or to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or provide a plan for correction of such defective work within ten days after receipt of written notice from the Owner to commence and continue correction of such defective work or persistently and willfully fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work which is not performed in accordance with the Contract Documents, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3.

#### § 2.5 Owner's Right to Carry Out the Work

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten-day period after receipt of notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such default or neglect. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect and the Architect may, pursuant to Section 9.5.1, withhold or nullify a Certificate for Payment in whole or in part, to the extent reasonably necessary to reimburse the Owner for the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Architect's additional services made necessary by such default, neglect, or failure. If current and future payments are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner. If the Contractor disagrees with the actions of the Owner or the Architect, or the amounts claimed as costs to the Owner, the Contractor may file a Claim pursuant to Article 15.

#### ARTICLE 3 CONTRACTOR

#### § 3.1 General

- § 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative. The Contractor may also be referred to herein and in other Contract Documents as the "Construction Manager".
- § 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents.
- § 3.1.3 The Contractor shall not be relieved of its obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor except to the extent non-conforming Work may be specifically approved by the Architect.

#### § 3.2 Review of Contract Documents and Field Conditions by Contractor

- § 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed, and correlated personal observations with requirements of the Contract Documents. Change Orders shall be allowed for such changes as are required by any permitting or regulatory agencies, or requested by Owner, and for unforeseen or changed conditions, including, but not limited to, subsurface conditions, except to the extent such conditions were discovered or should have been discovered by the Construction Manager using reasonable means prior fixing the Guaranteed Maximum Price.
- § 3.2.2 The Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.3.4, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Architect any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents. Neither the Contractor, nor any of its subcontractors shall be required to conduct an independent geological investigation of the site and the Contractor and its subcontractors shall be entitled to rely on information provided by the Owner for subsurface conditions on the site.
- § 3.2.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require. Contractor's review of the Contract Documents is solely in its capacity as a contractor for purposes of establishing the Cost of Work, schedule and constructability. Contractor makes no representations regarding the Project's design including, but not limited, any representation that the Contract Documents are in accordance with applicable laws, statues, ordinances, codes, rules and regulations, or lawful orders of public authorities or that the design, as represented by the Contract Documents, is suitable for the purpose intended by the Design Professional preparing said documents. Contractor shall be entitled to rely on the Contract Documents provided by the Owner and its Architect and consultants for all such matters. Nevertheless, should Contractor notice or determine there are errors in the Contract Documents, Contractor shall notify Architect and Owner of such errors as soon as practicable.
- § 3.2.4 Notwithstanding anything herein to the contrary, the Contractor shall not be liable for default or delay because of labor shortages due to Covid-19 aka the Novel Coronavirus, whether due to voluntary quarantine, any public health order, or illness of personnel, interruptions in the supply chain for materials, or otherwise, such delays being a force majeure, which shall excuse Contractor's strict compliance with any completion schedule requirements and application of Owner's contractual remedies by way of liquidated damages or supplementation of Contractor's Work, or as otherwise provided in the Contract Documents. The Contractor shall notify Owner in writing after the commencement of any excusable delay, including force majeure as described above. Should contractor not be able to continue within the completion time frames established, Owner may terminate the Contractor for Convenience and seek another party and replace Contractor.

#### § 3.3 Supervision and Construction Procedures

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences, and procedures, and for coordinating all portions of the Work under the Contract. The above notwithstanding, Contractor's Subcontractors shall be independently responsible for the safety of their employees' job-site safety in connection with their subcontracted portions of the Work. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences, or procedures, the Contractor shall evaluate the jobsite safety thereof and shall be solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures may not be safe, the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely notice to the Owner and Architect, and shall propose alternative means, methods, techniques, sequences, or procedures. The Architect shall evaluate the proposed alternative solely for conformance with the design intent for the completed construction. If the Architect approves of the Contractor's

proposed alternative, the Contractor shall perform the Work using its alternative means, methods, techniques, sequences, or procedures.

- § 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.
- § 3.3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

#### § 3.4 Labor and Materials

- § 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.
- § 3.4.2 Except in the case of minor changes in the Work approved by the Architect in accordance with Section 3.12.8 or ordered by the Architect in accordance with Section 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Architect and in accordance with a Change Order or Construction Change Directive.
- § 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

#### § 3.5 Warranty

- § 3.5.1 The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.
- § 3.5.2 All material, equipment, or other special warranties required by the Contract Documents shall be issued in the name of the Owner, or shall be transferable to the Owner, and shall commence in accordance with Section 9.8.4.

#### § 3.6 Taxes

The Owner qualifies for exemption from State and Local Sales and Use Taxes pursuant to the provisions of Article 20.04(f) of the Texas Limited Sales, Excise and use Tax Act. Exemption certificates comply with State Comptroller of Public accounts Ruling No. 95-0.07. Any such exemption certificate issued in lieu of tax shall be subject to State Comptroller of Public Accounts Ruling No. 95-0.09, as amended. Failure by the Construction Manager or Subcontractors to take advantage of the Owner's exemption and to obtain such exemption certificate shall make him responsible for paying taxes incurred without additional cost to or reimbursement by the Owner.

#### § 3.7 Permits, Fees, Notices and Compliance with Laws

- § 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded, provided that Contractor is not responsible for temporary utility charges, utility district/company inspection and permit fees, temporary tap charges, and temporary water meter charges and any other similar fees assessed by jurisdictional authority having control over this Project.
- § 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.

- § 3.7.2.1 It is not the Construction Manager's reasonability to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities. However, if the Construction manager observes that portions of the Contract Documents are at variance therewith, the construction manager shall promptly notify the Architect and owner in writing, and necessary changes shall be accomplished by appropriate Modification.
- § 3.7.3 If the Construction Manager performs Work which he knew or should have known to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, without notifying the Architect and Owner, the Construction Manager shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

#### § 3.7.4 Concealed or Unknown Conditions

If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 14 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend that an equitable adjustment be made in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor, stating the reasons. If either party disputes the Architect's determination or recommendation, that party may submit a Claim as provided in Article 15.

- § 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15.
- § 3.7.6 The Construction Manager shall comply with the provisions of Section 22.08341 of the Texas Education Code and Section 153.1117 of the Texas Administrative Code. The form of certification by the Construction Manager shall be supplied by the Owner, and must supplemented by the Construction manager as required by law, or as requested by the Owner.

#### § 3.8 Allowances

- § 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Architect or Owner may direct and approve in writing.
- § 3.8.2 Unless otherwise provided in the Contract Documents,
  - .1 allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
  - .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit, and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
  - .3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.
- § 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

#### § 3.9 Superintendent

- § 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor.
- § 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall notify the Owner and Architect of the name and qualifications of a proposed superintendent. Within 14 days of receipt of the information, the Architect may notify the Contractor, stating whether the Owner or the Architect (1) has reasonable objection to the proposed superintendent or (2) requires additional time for review. Failure of the Architect to provide notice within the 14-day period shall constitute notice of no reasonable objection.
- § 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed.

# § 3.10 Contractor's Construction and Submittal Schedules

- § 3.10.1 The Contractor, promptly after being awarded the Contract, shall submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall contain detail appropriate for the Project, including (1) the date of commencement of the Work, interim schedule milestone dates, and the date of Substantial Completion; (2) an apportionment of the Work by construction activity; and (3) the time required for completion of each portion of the Work. The schedule shall provide for the orderly progression of the Work to completion and shall not exceed time limits current under the Contract Documents. The schedule shall be revised at appropriate intervals as required by the conditions of the Work and Project.
- § 3.10.2 The Contractor, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, shall submit a submittal schedule for the Architect's approval. The Architect's approval shall not be unreasonably delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, or fails to provide submittals in accordance with the approved submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.
- § 3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Architect.

# § 3.11 Documents and Samples at the Site

The Contractor shall make available, at the Project site, the Contract Documents, including Change Orders, Construction Change Directives, and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and the approved Shop Drawings, Product Data, Samples, and similar required submittals. These shall be in electronic form or paper copy, available to the Architect and Owner, and delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

# § 3.12 Shop Drawings, Product Data and Samples

- § 3.12.1 Shop Drawings are drawings, diagrams, schedules, and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier, or distributor to illustrate some portion of the Work.
- § 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.
- § 3.12.3 Samples are physical examples that illustrate materials, equipment, or workmanship, and establish standards by which the Work will be judged.
- § 3.12.4 Shop Drawings, Product Data, Samples, and similar submittals are not Contract Documents. Their purpose is to demonstrate how the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review

by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.

- § 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve, and submit to the Architect, Shop Drawings, Product Data, Samples, and similar submittals required by the Contract Documents, in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of Separate Contractors.
- § 3.12.6 By submitting Shop Drawings, Product Data, Samples, and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents. The above notwithstanding, the Contractor shall not be responsible for cost to correct Work as a result of errors or omissions in the contract Documents.
- § 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples, or similar submittals, until the respective submittal has been approved by the Architect.
- § 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from the requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples, or similar submittals, unless the Contractor has specifically notified the Architect of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples, or similar submittals, by the Architect's approval thereof.
- § 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples, or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such notice, the Architect's approval of a resubmission shall not apply to such revisions.
- § 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work. The Contractor shall not be required to provide professional services in violation of applicable law.
- § 3.12.10.1 If professional design services or certifications by a design professional related to systems, materials, or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall be entitled to rely upon the adequacy and accuracy of the performance and design criteria provided in the Contract Documents. The Contractor shall cause such services or certifications to be provided by an appropriately licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings, and other submittals prepared by such professional. Shop Drawings, and other submittals related to the Work, designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor the performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review and approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents.
- § 3.12.10.2 If the Contract Documents require the Contractor's design professional to certify that the Work has been performed in accordance with the design criteria, the Contractor shall furnish such certifications to the Architect at the time and in the form reasonably specified by the Architect.

# § 3.13 Use of Site

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

# § 3.14 Cutting and Patching

- § 3.14.1 The Contractor shall be responsible for cutting, fitting, or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting, or patching shall be restored to the condition existing prior to the cutting, fitting, or patching, unless otherwise required by the Contract Documents.
- § 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or Separate Contractors by cutting, patching, or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter construction by the Owner or a Separate Contractor except with written consent of the Owner and of the Separate Contractor. Consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold, from the Owner or a Separate Contractor, its consent to cutting or otherwise altering the Work.

# § 3.15 Cleaning Up

- § 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials and rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery, and surplus materials from and about the Project.
- § 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and the Owner shall be entitled to reimbursement from the Contractor.

# § 3.16 Access to Work

The Contractor shall provide the Owner and Architect with access to the Work in preparation and progress wherever located.

#### § 3.17 Royalties, Patents and Copyrights

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications, or other documents prepared by the Owner or Architect. However, if an infringement of a copyright or patent is discovered by, or made known to, the Contractor, the Contractor shall be responsible for the loss unless the information is promptly furnished to the Architect.

#### § 3.18 Indemnification, Acknowledgment of Limited Liability, Etc.

§ 3.18.1 To the fullest extent permitted by law, the Construction Manager shall indemnify and hold harmless the Owner and its directors, officers, agents and employees (the "Indemnitees") from and against claims, damages, losses and expenses, including without limitation, attorney's fees, arising out of or relating to the Work of this Construct, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease, or death, or injury to or destruction of tangible property (other than the Work itself), including loss of use resulting therefrom, to the extent such claim, damage, loss or expense is caused by the negligence or fault by the Construction Manager, a Subcontractor of any tier, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable.

# § 3.18.2 Reserved

§ 3.18.3 In the event of any claim alleging partial, contributory, concurrent, or joint wrongful conduct of the Indemnitees that is not covered under the Construction Manager's indemnity obligations under Section 3.18.2, The Construction manager shall be obligated to reimburse the Owner for its reasonable attorney's fees in proportion to the Construction Manager's liability, as such may be agreed to by the Construction manager or found by a trier of fact.

- § 3.18.4 The defense, indemnity and reimbursement obligations in Section 3.18.1 and 3.18.3 shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity which would otherwise exist as to the party or persons described in this Section 3.18, or elsewhere in the Contract Documents. These defense, indemnity and reimbursement obligations shall not be limited by any limitation on the amount or type of compensation, benefits or damages payable by or for the construction manager under any workers' compensation or other benefits laws, or by the limits of any insurance of the Construction Manager. These defense, indemnity and reimbursement obligations shall survive completion of the Work or early termination of the Contract.
- § 3.18.5 The Construction Manager acknowledges that the Owner shall have no obligation to supervise performance of the Work or Work Site for safety, nor does the Owner exercise any control over the means and methods of construction employed by the Construction manager, Subcontractor of any tier, or anyone directly or indirectly employed by them. The Construction Manger further acknowledges that the Owner's liability for claims of personal injury, death, or property damage are limited by Section 95.003 of the Texas Civil Practice and Remedies Code, as well as the privileges and immunities enjoyed by Owner as a governmental unit of the State of Texas.
- § 3.18.6 The Construction Manager shall provide worker's compensation insurance coverage for each employee of the Construction Manager employed on the Project, and shall require the same of its Subcontractors of any tier. The Construction Manager and each Subcontractor of any tier shall be required to provide certificates of this coverage to the Owner.
- § 3.18.7 To the extent any of the obligations in this Section 3.18 violate appliable law, the obligation(s) will be reformed or severed to the minimum extent necessary to comply with applicable law in order to provide the maximum protection to the Indemnitees.

#### § 3.19 Antitrust Violations

§ 3.19.1 Construction Manager hereby assigns to owner any and all claims for overcharges associated with this Contract which arise under the antitrust laws of the United States, 15 A.S.C.A. Section 1 et. seq. (1973).

#### ARTICLE 4 ARCHITECT

# § 4.1 General

- § 4.1.1 The Architect is the person or entity retained by the Owner pursuant to Section 2.3.2 and identified as such in the Agreement.
- § 4.1.2 Duties, responsibilities, and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified, or extended without written consent of the Owner, Contractor, and Architect. Consent shall not be unreasonably withheld.

#### § 4.2 Administration of the Contract

- § 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate for Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.
- § 4.2.2 The Architect, as a representative of the Owner, will visit the site at intervals appropriate to the stage of the Construction manager's operations (1) to become generally familiar with and to keep the Owner informed about the progress and quality of the portion of the Work completed, (2) to endeavor to guard the owner against defects and deficiencies in the Work, and (3) to determine in general if the Work is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. The Architect will be required to make on-site inspections as necessary to keep the Owner informed or the progress of the Work and as necessary to guard the Owner against defects and deficiencies in the Work. The Architect will neither have control over or charge or, nor be responsible for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Construction Manager's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.
- § 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's

Powered by BoardOnTrack

failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of, and will not be responsible for acts or omissions of, the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

#### § 4.2.4 Communications

The Owner and Contractor shall include the Architect in all communications that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and suppliers shall be through the Contractor. Communications by and with Separate Contractors shall be through the Owner. The Contract Documents may specify other communication protocols.

- § 4.2.5 Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.
- § 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. The Architect shall be required to promptly notify the Owner of any non-conforming Work and, unless otherwise notified in writing by the Owner, shall reject such non-conforming Work. Whenever the Architect considers it necessary or advisable for implementation of the intent of the Contract Documents, the Architect will have authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not such Work is fabricated, installed or completed. Performance of any additional inspection or testing, which would result in additional cost to the Owner, shall require advance notice to and approval of the Owner. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Construction Manager, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work, except when the Construction Manager's inability to perform the Work is a result of design flaw, error or omission.
- § 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data, and Samples. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review, but, in any event with sufficient promptness so as not to delay the Work. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5, and 3.12. The Architect's review shall not constitute approval of safety precautions or of any construction means, methods, techniques, sequences, or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- § 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may order minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.
- § 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.
- § 4.2.10 If the Owner and Architect agree, the Architect will provide one or more Project representatives to assist in carrying out the Architect's responsibilities at the site. The Owner shall notify the Contractor of any change in the duties, responsibilities and limitations of authority of the Project representatives.
- § 4.2.11 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.

**User Notes:** 

- § 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either, and will not be liable for results of interpretations or decisions rendered in good faith.
- § 4.2.13 All decisions on matters relating to aesthetic effect shall initially be made by the Architect; however, all such decisions are subject to the Owner's written approval.
- § 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

#### ARTICLE 5 SUBCONTRACTORS

### § 5.1 Definitions

- § 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a Separate Contractor or the subcontractors of a Separate Contractor.
- § 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

#### § 5.2 Award of Subcontracts and Other Contracts for Portions of the Work

Refer to the terms of the Agreement AIA Document A133-2009.

(Paragraphs deleted)

#### § 5.3 Subcontractual Relations

By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work that the Contractor, by these Contract Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies, and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

# § 5.4 Contingent Assignment of Subcontracts

- § 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that
  - assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor; and
  - .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract.

§ 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for increases in cost resulting from the suspension.

§ 5.4.3 Upon assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract.

### ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

### § 6.1 Owner's Right to Perform Construction and to Award Separate Contracts

- § 6.1.1 The term "Separate Contractor(s)" shall mean other contractors retained by the Owner under separate agreements. The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and with Separate Contractors retained under Conditions of the Contract substantially similar to those of this Contract, including those provisions of the Conditions of the Contract related to insurance and waiver of subrogation.
- § 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.
- § 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each Separate Contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with any Separate Contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to its construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, Separate Contractors, and the Owner until subsequently revised.
- § 6.1.4 Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces or with Separate Contractors, the Owner or its Separate Contractors shall have the same obligations and rights that the Contractor has under the Conditions of the Contract, including, without excluding others, those stated in Article 3, this Article 6, and Articles 10, 11, and 12.

#### § 6.2 Mutual Responsibility

- § 6.2.1 The Contractor shall afford the Owner and Separate Contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.
- § 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a Separate Contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly notify the Architect of apparent discrepancies or defects in the construction or operations by the Owner or Separate Contractor that would render it unsuitable for proper execution and results of the Contractor's Work. Failure of the Contractor to notify the Architect of apparent discrepancies or defects prior to proceeding with the Work shall constitute an acknowledgment that the Owner's or Separate Contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work. The Contractor shall not be responsible for discrepancies or defects in the construction or operations by the Owner or Separate Contractor that are not apparent.
- § 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a Separate Contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for costs the Contractor incurs because of a Separate Contractor's delays, improperly timed activities, damage to the Work or defective construction.
- **§ 6.2.4** The Contractor shall promptly remedy damage that the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or Separate Contractor as provided in Section 10.2.5.
- **§ 6.2.5** The Owner and each Separate Contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.

**User Notes:** 

# § 6.3 Owner's Right to Clean Up

If a dispute arises among the Contractor, Separate Contractors, and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect will allocate the cost among those responsible.

#### ARTICLE 7 CHANGES IN THE WORK

#### § 7.1 General

- § 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive not affecting time or the Cost of the Work or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.
- § 7.1.2 A Change Order shall be based upon agreement among the Owner's Board of Directors, Construction Manager, and Architect. A Construction Change Directive requires agreement by the Owner or the Owner's representative and Architect, and may or may not be agreed to by the Construction Manager; an order for a minor change not affecting time or the Cost of the work may be issued by the Architect alone.
- § 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents. The Contractor shall proceed promptly with changes in the Work, unless otherwise provided in the Change Order, Construction Change Directive, or order for a minor change in the Work.

# § 7.2 Change Orders

- § 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor, and Architect stating their agreement upon all of the following:
  - .1 The change in the Work;
  - .2 The amount of the adjustment, if any, in the Contract Sum; and
  - .3 The extent of the adjustment, if any, in the Contract Time.

# § 7.3 Construction Change Directives

- § 7.3.1 A Construction Change Directive is a written order prepared by the Architect and signed by the Owner and Architect, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions, the Contract Sum and Contract Time being adjusted accordingly.
- § 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.
- § 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:
  - .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
  - .2 Unit prices stated in the Contract Documents or subsequently agreed upon;
  - 3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
  - .4 As provided in Section 7.3.4.
- § 7.3.4 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.4 shall be limited to the following:
  - .1 Costs of labor, including applicable payroll taxes, fringe benefits required by agreement or custom, workers' compensation insurance, and other employee costs approved by the Architect;
  - .2 Costs of materials, supplies, and equipment, including cost of transportation, whether incorporated or consumed;

- 3 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others:
- .4 Costs of premiums for all bonds and insurance, permit fees, and sales, use, or similar taxes, directly related to the change; and
- .5 Costs of supervision and field office personnel directly attributable to the change.
- § 7.3.5 If the Contractor disagrees with the adjustment in the Contract Time, the Contractor may make a Claim in accordance with applicable provisions of Article 15.
- § 7.3.6 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.
- § 7.3.7 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.
- § 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- § 7.3.9 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.
- § 7.3.10 When the Owner and Contractor agree with a determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive.

### (Paragraphs deleted)

#### ARTICLE 8 TIME

# § 8.1 Definitions

- **§ 8.1.1** Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.
- § 8.1.2 The date of commencement of the Work is the date established in the Agreement.
- § 8.1.3 The date of Substantial Completion is the date certified by the Architect in accordance with Section 9.8.
- § 8.1.4 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

# § 8.2 Progress and Completion

- § 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work.
- § 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, commence the Work prior to the effective date of insurance required to be furnished by the Contractor and Owner.
- § 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

# § 8.3 Delays and Extensions of Time

§ 8.3.1 If the Construction Manager is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a Separate Contractor employed by the Owner, or by changes ordered in the Work, or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Construction Manager's Control, or by other causes which the Architect determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine, and an adjustment may be made to the Guaranteed maximum Price, for such extended general conditions expenses and other proven loss resulting from such delay, which the Construction manager must substantiate with supporting documentation, subject to approval by the Architect.

- § 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.
- § 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under other provisions of the Contract Documents.
- § 8.3.4 The parties hereto agree that time is of the essence of this contract, and that pecuniary damages would be suffered by the Owner if the Construction Manager does not substantially complete all Work called for in the Contract Document by the specified date, which damages are, by their very nature, difficult of ascertainment. It is therefore expressly agreed, as a part of the consideration inducing the Owner to execute this Contract that the Owner may deduct from the final payment made to the Construction Manager a sum equal to Five Hundred Dollars (\$500.00) per day, for each and every Calendar Day beyond the agreed date which the Construction Manager has agreed to for Substantial Completion of the Work included in the Contract Document. It is expressly understood that said sum per day is agreed upon as a fair estimate of the pecuniary damages which will be sustained by the Owner in the event that the Work is not substantially completed within the agreed time, or with the legally extended time, if any, otherwise provided for herein. Said sum shall be considered as liquidated damages only, and in no sense shall be considered a penalty or forfeiture; said damage being caused by additional compensation to personnel, and other miscellaneous increased costs, all of which are difficult of exact ascertainment. Delays related to COVID 19 are subject to the provisions of §3.2.4 and shall not be included in any calculation of liquidated damages.

#### ARTICLE 9 PAYMENTS AND COMPLETION

# § 9.1 Contract Sum

- § 9.1.1 The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.
- § 9.1.2 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed so that application of such unit prices to the actual quantities causes substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

#### § 9.2 Schedule of Values

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit a schedule of values to the Architect before the first Application for Payment, allocating the entire Contract Sum to the various portions of the Work. The schedule of values shall be prepared in the form, and supported by the data to substantiate its accuracy, required by the Architect. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment. Any changes to the schedule of values shall be submitted to the Architect and supported by such data to substantiate its accuracy as the Architect may require, and unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's subsequent Applications for Payment.

# § 9.3 Applications for Payment

§ 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work. The application shall be notarized, if required, and supported by all data substantiating the Contractor's right to payment that the Owner or Architect require, such as copies of requisitions, and releases and waivers of liens from Subcontractors and suppliers, and shall reflect retainage if provided for in the Contract Documents.

- § 9.3.1.1 As provided in Section 7.3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders.
- § 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or supplier, unless such Work has been performed by others whom the Contractor intends to pay.
- § 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment (1) incorporated in the Work, (2) suitably stored at the site, or (3) suitably stored at some off-site location shall be conditioned upon compliance by the Construction Manager with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest for all-site storage, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site as follows:
- .1 The location must be agreed to, in writing, by the Owner and the Surety.
- .2 The location must be a bonded warehouse.
- .3 The Surety must agree, in writing, to each request for payment.
- .4 The Construction Manager must bear the cost of the Owner's and Architect's reasonable expenses related to visiting the off-site storage area.

Payments for materials or equipment stored on or off the site shall be conditioned upon submission by the Construction Manager of Bills of Sale or such other procedures satisfactory to the Owner to establish the Owner's title to such materials or equipment or otherwise protect the Owner's interest, including applicable insurance (naming the Owner as insured) and transportation to the site for those materials and equipment stored off the site. Under no circumstances will the Owner reimburse the Construction Manager for down payments, deposits, or other advance payments for materials or equipment without prior written consent from the Owner, which shall not be unreasonably withheld.

§ 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information, and belief, be free and clear of liens, claims, security interests, or encumbrances, in favor of the Contractor, Subcontractors, suppliers, or other persons or entities that provided labor, materials, and equipment relating to the Work.

# § 9.4 Certificates for Payment

- § 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either (1) issue to the Owner a Certificate for Payment in the full amount of the Application for Payment, with a copy to the Contractor; or (2) issue to the Owner a Certificate for Payment for such amount as the Architect determines is properly due, and notify the Contractor and Owner of the Architect's reasons for withholding certification in part as provided in Section 9.5.1; or (3) withhold certification of the entire Application for Payment, and notify the Contractor and Owner of the Architect's reason for withholding certification in whole as provided in Section 9.5.1.
- § 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data in the Application for Payment, that, to the best of the Architect's knowledge, information, and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion, and to specific qualifications expressed by the Architect. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work; (2) reviewed construction means, methods, techniques, sequences, or procedures; (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment;

or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

#### § 9.5 Decisions to Withhold Certification

- § 9.5.1 The Architect may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of
  - defective Work not remedied;
  - .2 third party claims filed or reasonable evidence indicating probable filing of such claims, unless security acceptable to the Owner is provided by the Contractor;
  - .3 failure of the Contractor to make payments properly to Subcontractors or suppliers for labor, materials or equipment;
  - damage to the Owner or a Separate Contractor;

(Paragraph deleted)

- repeated failure to carry out the Work in accordance with the Contract Documents. .5
- § 9.5.2 When either party disputes the Architect's decision regarding a Certificate for Payment under Section 9.5.1, in whole or in part, that party may submit a Claim in accordance with Article 15.
- § 9.5.3 When the reasons for withholding certification are removed, certification will be made for amounts previously withheld.
- § 9.5.4 If the Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or supplier to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Contractor shall reflect such payment on its next Application for Payment.

#### § 9.6 Progress Payments

- § 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.
- § 9.6.2 The Contractor shall pay each Subcontractor, no later than seven days after receipt of payment from the Owner, the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.
- § 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.
- § 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors and suppliers to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay, or to see to the payment of money to, a Subcontractor or supplier, except as may otherwise be required by law.
- § 9.6.5 The Contractor's payments to suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

- § 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.
- § 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors or provided by suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, create any fiduciary liability or tort liability on the part of the Contractor for breach of trust, or entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.
- § 9.6.8 Provided the Owner has fulfilled its payment obligations under the Contract Documents, the Contractor shall defend and indemnify the Owner from all loss, liability, damage or expense, including reasonable attorney's fees and litigation expenses, arising out of any lien claim or other claim for payment by any Subcontractor or supplier of any tier. Upon receipt of notice of a lien claim or other claim for payment, the Owner shall notify the Contractor. If approved by the applicable court, when required, the Contractor may substitute a surety bond for the property against which the lien or other claim for payment has been asserted.

### § 9.7 Failure of Payment

If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents, the amount certified by the Architect then the Contractor may, upon seven additional days' notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided for in the Contract Documents.

# § 9.8 Substantial Completion

- § 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use. Issuance of a temporary or final Certificate of Occupancy shall be prima facie evidence of Substantial Completion. However, should the Work or designated portion thereof be sufficiently complete so that the Owner can occupy or utilize the Work for its intended use, Substantial Competition shall be deemed to have occurred, regardless of the failure or delay, through no fault of the Contractor, by any governmental agency to issue a temporary or final Certificate of Occupancy.
- § 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.
- § 9.8.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.
- § 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion; establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance; and fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

**User Notes:** 

§ 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in the Certificate. Upon such acceptance, and consent of surety if any, the Owner shall make payment of retainage applying to the Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

### § 9.9 Partial Occupancy or Use

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect. For any partial occupancy or use, the Owner shall reduce retainage proportionately to the Contractor at the time of partial occupancy or use.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor, and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

# § 9.10 Final Completion and Final Payment

§ 9.10.1 Upon receipt of the Contractor's notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection. When the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect, (3) a written statement that the Contractor knows of no reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment, (5) documentation of any special warranties, such as manufacturers' warranties or specific Subcontractor warranties, and (6) if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts and releases and waivers of liens, claims, security interests, or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien, claim, security interest, or encumbrance. If a lien, claim, security interest, or encumbrance remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging the lien, claim, security interest, or encumbrance, including all costs and reasonable attorneys' fees.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed, corrected, and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of the surety to payment of the balance due for that

portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of Claims.

- § 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from
  - liens, Claims, security interests, or encumbrances arising out of the Contract and unsettled;
  - .2 failure of the Work to comply with the requirements of the Contract Documents;
  - .3 terms of special warranties required by the Contract Documents; or
  - audits performed by the Owner, if permitted by the Contract Documents, after final payment.
- § 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor, or a supplier, shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.
- § 9.10.6 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be paid by the Owner to the Construction Manager thirty-one (31) days after Substantial Completion of the Work unless otherwise stipulated in the Certificate of Substantial Completion, provided the work has then been completed, the Contract fully performed, all Contract Close Out Documents have been submitted, and the Final Certificate for Payment has been issued by the Architect. The final payment will not be made until all of these conditions have been satisfied.

#### ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

# § 10.1 Safety Precautions and Programs

The Contractor shall be responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the performance of the Contract.

# § 10.2 Safety of Persons and Property

- § 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury, or loss to
  - employees on the Work and other persons who may be affected thereby;
  - the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, .2 under care, custody, or control of the Contractor, a Subcontractor, or a Sub-subcontractor; and
  - other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures, and utilities not designated for removal, relocation, or replacement in the course of construction.
- § 10.2.2 The Contractor shall comply with, and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, bearing on safety of persons or property or their protection from damage, injury, or loss.
- § 10.2.3 The Contractor shall implement, erect, and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards; promulgating safety regulations; and notifying the owners and users of adjacent sites and utilities of the safeguards.
- § 10.2.4 When use or storage of explosives or other hazardous materials or equipment, or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- § 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3. The Contractor may make a Claim for the cost to remedy the damage or loss to the extent such damage or loss is attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.

- § 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.
- § 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

# § 10.2.8 Injury or Damage to Person or Property

If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, notice of the injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

§ 10.2.9 The Construction Manager shall be responsible for taking reasonable precautions to protect the Work in place from foreseeable weather conditions which could cause any potential damage to portions or all Work in place. The Construction Manager shall be responsible for performing all repairs and/or replacement of any Work that is damaged as a result of the Construction manager's failure to take such reasonable precautions.

#### § 10.3 Hazardous Materials and Substances

- § 10.3.1 The Construction Manager is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials or substances. If the Construction Manager encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance encountered on the site by the Construction Manager, the Construction Manager shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner and Architect in writing. The Owner, Construction Manager and Architect shall then proceed in the same manner described in Section 10.3.2.
- § 10.3.2 Upon receipt of the Contractor's notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of the material or substance or who are to perform the task of removal or safe containment of the material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable additional costs of shutdown, delay, and start-up.
- § 10.3.3 If, without negligence on the part of the Construction Manager, the Construction Manager is held liable by a governmental agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall indemnify the Construction Manager for all costs and expenses thereby incurred, but only to the extent provided by law.
- § 10.3.4 The Owner shall not be responsible under this Section 10.3 for hazardous materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for hazardous materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling of such materials or substances.
- § 10.3.5 The Contractor shall reimburse the Owner for the cost and expense the Owner incurs (1) for remediation of hazardous materials or substances the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

§ 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall reimburse the Contractor for all cost and expense thereby incurred.

# § 10.4 Emergencies

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury, or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

#### ARTICLE 11 INSURANCE AND BONDS

#### § 11.1 Contractor's Insurance and Bonds

- § 11.1.1 The Contractor shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in the Agreement or elsewhere in the Contract Documents. The Contractor shall purchase and maintain the required insurance from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located. The Owner, Architect, and Architect's consultants shall be named as additional insureds under the Contractor's commercial general liability policy or as otherwise described in the Contract Documents.
- § 11.1.2 The Contractor shall provide surety bonds of the types, for such penal sums, and subject to such terms and conditions as required by the Contract Documents. The Contractor shall purchase and maintain the required bonds from a company or companies lawfully authorized to issue surety bonds in the jurisdiction where the Project is located.
- § 11.1.3 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.
- § 11.1.4 The insurance required by Section 11.1.1 shall be written for not less than the following limits or the limits required by law, whichever coverage is greater:
  - .1 Workers' Compensation and Employer's Liability Insurance. Statutory Limits with All States endorsement and minimum Employer's Liability Limits will be provided as follows:
    - .1 \$1,000,000 Bodily Injury with Accident Each Accident;
    - .2 \$1,000,000 Bodily Injury by Disease Policy Limit; and
    - .3 \$1,000,000 Bodily Injury by Disease Each Employee.
    - .4 A waiver of subrogation shall be granted in favor of the Owner.

#### .2 Commercial General Liability Insurance.

- .1 Limits of coverages:
  - .1 \$1,000,000 Each Occurrence
  - .2 \$2,000,000 Products/Completed Operations Aggregate
  - .3 \$1,000,000 Personal and Advertising Injury
  - **.4** \$50,000 Fire Damage
  - .5 \$5,000 Medical Expenses
- .2 Coverage shall include the following:
  - .1 Occurrence Basis;
  - .2 Premises Operations;
  - .3 Contractual Liability;
  - .4 Products/Completed Operations;
  - .5 Broad Form Property Damage; and
  - **.6** Independent Contractors.
- .3 This policy will be written and evidenced on an occurrence basis. One year Products-Completed Operation hazard shall take in effect at completion of the project.
- **Business Automobile Liability Insurance.** Comprehensive Automobile Liability Insurance to cover all vehicles owned by, hired by, or used on behalf of Construction Manager, with minimum combined single limit in the amount of not less than \$1,000,000 per occurrence. Owner and its

directors, officers, representatives, agents and employees shall be endorsed as Additional Insureds, ATIMA (As Their Interest May Appear).

#### .4 Excess (Umbrella) Liability Insurance.

- Limits of coverages:
  - .1 \$5,000,000 Each Occurrence
  - .2 \$5,000,000 Aggregate
- The Umbrella shall provide coverage over the General Liability and Comprehensive Automobile Liability Insurance Coverages. The Owner, the Architect and all Consultants listed on the Title Page of the Project Manual shall be listed as Additional Insureds on the Construction Manager's policy associated with the Project. Provisions shall be included for Waiver of Subrogation against the Owner and Architect and his Consultants, where allowed by
- § 11.1.5 Original certificates of insurance evidencing all required insurance hereunder shall be filed with the Owner upon prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance. All insurance policies and certificates required hereunder shall be in form and content satisfactory to Owner. The Construction Manager (and each Subcontractor) shall maintain the insurance coverages specified herein at all times during the term of this Agreement or such later date specified herein. If any of the foregoing insurance coverages are required to remain in force after final payment and are reasonably available, an additional certificate evidencing continuation of such coverage shall be submitted with the final Application for Payment as required by Section 9.10.2 and thereafter, upon renewal or replacement of such policy of insurance. All required insurance shall:
  - include a provision stating that coverages afforded under the policies will not be terminated, materially modified, or allowed to expire until at least 30 days' prior written notice has been given to the Owner;
  - .2 be written by an insurance company or companies satisfactory to owner and licensed to do business in Texas, with a rating of not less than B+ VII in the last available Best's Rating guide;
  - .3 excluding professional liability and employer's liability coverage, include clauses whereby each underwriter agrees to waive its rights of subrogation against the Owner; and
- ACORD Certificate of Insurance 25 (2016/03) signed by Contract Manager's Insurance Representative/Producer. Policy exclusions and restrictions should be clearly explained on the Certificate or in an attached letter from the Issuing Agency. Blank areas on the Certificate should have "not covered" written across the printed areas when coverage is not provided.
- § 11.1.6 The Contractor shall cause the commercial liability coverage required by the Contract Documents to include (1) the Owner, the Architect and the Architect's consultants as additional insureds for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's operations; and (2) the Owner as an additional insured for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's completed operations.

# § 11.1.7 Property Insurance

§ 11.1.7.1 The Contractor shall purchase and maintain, from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located, property insurance written on a builder's risk "all-risks" completed value or equivalent policy form and sufficient to cover the total value of the Work on a replacement cost basis. The Contractor's property insurance coverage shall be no less than the amount of the initial Contract Sum, plus the value of subsequent Modifications and labor performed and materials or equipment supplied by others. The property insurance shall be maintained for one until Substantial Completion unless otherwise provided in the Contract Documents or otherwise agreed in writing by the parties to this Agreement. This insurance shall include the interests of the Owner, Contractor, Subcontractors, and Sub-subcontractors in the Project as insureds. This insurance shall include the interests of mortgagees as loss payees.

§ 11.1.7.2 Occupancy or Use Prior to Substantial Completion. The Owner's occupancy or use of any completed or partially completed portion of the Work prior to Substantial Completion shall not commence until the insurance company or companies providing the insurance under this Section 11.1.7 have consented in writing to the continuance of coverage. The Owner shall take no action with respect to partial occupancy or use that would cause cancellation, lapse, or reduction of insurance, unless agreed otherwise in writing by Contractor.

# § 11.1.7.3 Insurance for Existing Structures

If the Work involves remodeling an existing structure or constructing an addition to an existing structure, the Owner shall purchase and maintain, until the expiration of the period for correction of Work as set forth in Section 12.2.2 of the General Conditions, "all-risks" property insurance, on a replacement cost basis, protecting the existing structure against direct physical loss or damage from the causes of loss identified in Section 11.2.3.1, notwithstanding the undertaking of the Work. The Owner shall be responsible for all co-insurance penalties.

§ 11.1.8 Notice of Cancellation or Expiration of Contractor's Required Insurance. Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by the Contract Documents; the Contractor shall provide notice to the Owner of such impending or actual cancellation or expiration. Upon receipt of notice from the Contractor, the Owner shall, unless the lapse in coverage arises from an act or omission of the Owner, have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by the Contractor. The furnishing of notice by the Contractor shall not relieve the Contractor of any contractual obligation to provide any required coverage.

#### § 11.2 Owner's Insurance

§ 11.2.1 The Owner shall be responsible for purchasing and maintaining the Owner's general liability insurance.

§ 11.2.2 Failure to Purchase Required Property Insurance. If the Owner fails to purchase and maintain the required property insurance, with all of the coverages and in the amounts described in the Agreement or elsewhere in the Contract Documents, the Owner shall inform the Contractor in writing prior to commencement of the Work. Upon receipt of notice from the Owner, the Contractor may delay commencement of the Work and may obtain insurance that will protect the interests of the Contractor, Subcontractors, and Sub-Subcontractors in the Work. When the failure to provide coverage has been cured or resolved, the Contract Sum and Contract Time shall be equitably adjusted. In the event the Owner fails to procure coverage, the Owner waives all rights against the Contractor, Subcontractors, and Sub-subcontractors to the extent the loss to the Owner would have been covered by the insurance to have been procured by the Owner. The cost of the insurance shall be charged to the Owner by a Change Order. If the Owner does not provide written notice, and the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain the required insurance, the Owner shall reimburse the Contractor for all reasonable costs and damages attributable thereto.

### (Paragraph deleted)

§ 11.2.5 Notice of Cancellation or Expiration of Owner's Required Property Insurance. Within three (3) business days of the date the Owner becomes aware of an impending or actual cancellation or expiration of any property insurance required by the Contract Documents, the Owner shall provide notice to the Contractor of such impending or actual cancellation or expiration. Unless the lapse in coverage arises from an act or omission of the Contractor: (1) the Contractor, upon receipt of notice from the Owner, shall have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by either the Owner or the Contractor; (2) the Contract Time and Contract Sum shall be equitably adjusted; and (3) the Owner waives all rights against the Contractor, Subcontractors, and Sub-subcontractors to the extent any loss to the Owner would have been covered by the insurance had it not expired or been cancelled. If the Contractor purchases replacement coverage, the cost of the insurance shall be charged to the Owner by an appropriate Change Order. The furnishing of notice by the Owner shall not relieve the Owner of any contractual obligation to provide required insurance.

#### § 11.3 Waivers of Subrogation

§ 11.3.1 The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents, and employees, each of the other; (2) the Architect and Architect's consultants; and (3) Separate Contractors, if any, and any of their subcontractors, sub-subcontractors, agents, and employees, for damages caused by fire, or other causes of loss, to the extent those losses are covered by property insurance required by the Agreement or other property insurance applicable to the Project, except such rights as they have to proceeds of such insurance. The Owner or Contractor, as appropriate, shall require similar written waivers in favor of the individuals and entities identified above from the Architect, Architect's consultants, Separate Contractors, subcontractors, and sub-subcontractors. The policies of insurance purchased and maintained by each person or entity agreeing to waive claims pursuant to this section 11.3.1 shall not prohibit this waiver of subrogation. This waiver of subrogation shall be effective as to a person or entity (1) even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, (2) even though that person or entity did not pay the insurance premium directly or indirectly, or (3) whether or not the person or entity had an insurable interest in the damaged property.

§ 11.3.2 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, to the extent permissible by such policies, the Owner waives all rights in accordance with the terms of Section 11.3.1 for damages caused by fire or other causes of loss covered by this separate property insurance.

#### § 11.4 Loss of Use, Business Interruption, and Delay in Completion Insurance

The Owner, at the Owner's option, may purchase and maintain insurance that will protect the Owner against loss of use of the Owner's property, or the inability to conduct normal operations, due to fire or other causes of loss. The Owner waives all rights of action against the Contractor and Architect for loss of use of the Owner's property, due to fire or other hazards however caused.

# §11.5 Adjustment and Settlement of Insured Loss

§ 11.5.1 A loss insured under the property insurance required by the Agreement shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.5.2. The Owner shall pay the Architect and Contractor their just shares of insurance proceeds received by the Owner, and by appropriate agreements the Architect and Contractor shall make payments to their consultants and Subcontractors in similar manner.

§ 11.5.2 Prior to settlement of an insured loss, the Owner shall notify the Contractor of the terms of the proposed settlement as well as the proposed allocation of the insurance proceeds. The Contractor shall have 14 days from receipt of notice to object to the proposed settlement or allocation of the proceeds. If the Contractor does not object, the Owner shall settle the loss and the Contractor shall be bound by the settlement and allocation. Upon receipt, the Owner shall deposit the insurance proceeds in a separate account and make the appropriate distributions. Thereafter, if no other agreement is made or the Owner does not terminate the Contract for convenience, the Owner and Contractor shall execute a Change Order for reconstruction of the damaged or destroyed Work in the amount allocated for that purpose. If the Contractor timely objects to either the terms of the proposed settlement or the allocation of the proceeds, the Owner may proceed to settle the insured loss, and any dispute between the Owner and Contractor arising out of the settlement or allocation of the proceeds shall be resolved pursuant to Article 15. Pending resolution of any dispute, the Owner may issue a Construction Change Directive for the reconstruction of the damaged or destroyed Work.

# § 11.6 Performance Bond and Payment Bond

§ 11.6.1 The Contractor shall furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.

§ 11.6.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

§ 11.6.3 The Performance Bond Form and the Payment Bond form required herein shall be executed and submitted to the Architect in duplicate prior to commencement of the Work. The surety companies must be acceptable to the Owner and licensed admitted carriers in the State of Texas; and the surety shall be listed, with appropriate underwriting limitation, on the U.S. Treasury Department Circular No. 570 (Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies).

§ 11.6.4 Each bond shall be of penal sum equal to 100% of the estimated cost of construction and shall be compatible with the provisions of the governing authority. The Construction Manager shall file copies of each bond with the county clerk and furnish the Owner with a file receipt. The bonds shall remain in full force throughout the contractual correction period of the Agreement. The Work will not be started until the bonds and issuing companies have been accepted as satisfactory by the Owner. The original bonds will be delivered to the Owner with an authorized power of attorney attached thereto. If a fixed contract amount or guaranteed maximum price has not been determined at the time the Contract is awarded, the penal sums of the performance and payment bonds delivered to the Owner must each be in an amount equal to the project budget, as specified in the request for qualifications or request for sealed proposal.

The Construction Manager shall deliver the bonds not later than the tenth (10<sup>th</sup>) day after the date the Construction Manager executes this Agreement unless the Construction Manager furnishes a bid bond or other financial security acceptable to the Owner to ensure that the Construction Manager will furnish the required performance and payment bonds when a guaranteed maximum price is established.

Notwithstanding any other provision, the Performance Bond and the Payment Bond shall be provided in accordance with the following. Upon acceptance by Owner of a Guaranteed Maximum Price Proposal, Contractor shall provide performance and payment bonds on forms in accordance with the requirements of the Texas Government Code and this contract. The penal sum of the payment and performance bonds shall be equal to the GMP. If construction is phased or staged with different Guaranteed Maximum Prices established at different times, the penal sum of the bonds shall be increased at the start of each stage or phase based on the cumulative total value of all Guaranteed Maximum Prices in effect.

§ 11.6.5 Claims must be sent to the Construction Manager and his Surety, in accordance with Texas Government Code, Chapter 2253. The Owner will furnish in accordance with such Article, a copy of the payment bond, as provided therein, to claimants upon request. All claimants are cautioned that no lien exists on the funds unpaid to the Construction Manager on such Contract, and that reliance on notices sent to the Owner may result in loss of their rights against the Construction Manager and/or his Surety. The Owner is not responsible in any manner to a claimant for collection of unpaid bills, and accepts no responsibility because of any representation by any agent or employee.

### ARTICLE 12 UNCOVERING AND CORRECTION OF WORK

# § 12.1 Uncovering of Work

§ 12.1.1 If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.

§ 12.1.2 If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, the Contractor shall be entitled to an equitable adjustment to the Contract Sum and Contract Time as may be appropriate. If such Work is not in accordance with the Contract Documents, the costs of uncovering the Work, and the cost of correction, shall be at the Contractor's expense.

# § 12.2 Correction of Work

#### § 12.2.1 Before Substantial Completion

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, discovered before Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

#### § 12.2.2 After Substantial Completion

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of any applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of notice from the Owner to do so, unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.5.

§ 12.2.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

- § 12.2.2.3 The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2.
- § 12.2.3 The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.
- § 12.2.4 The Contractor shall bear the cost of correcting destroyed or damaged construction of the Owner or Separate Contractors, whether completed or partially completed, caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.
- § 12.2.5 Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

# § 12.3 Acceptance of Nonconforming Work

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

#### ARTICLE 13 MISCELLANEOUS PROVISIONS

### § 13.1 Governing Law

The Contract shall be governed by the law of the place where the Project is located.

# § 13.2 Successors and Assigns

- § 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns, and legal representatives to covenants, agreements, and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.
- § 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate the assignment.

# § 13.3 Rights and Remedies

- § 13.3.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights, and remedies otherwise imposed or available by law.
- § 13.3.2 No action or failure to act by the Owner, Architect, or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed upon in writing.

#### § 13.4 Tests and Inspections

§ 13.4.1 Tests, inspections, and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules, and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections, and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections, and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of tests, inspections, or approvals that do not become requirements until after bids are received or negotiations concluded. The Owner shall directly arrange and pay for tests, inspections, or approvals where building codes or applicable laws or regulations so require.

- § 13.4.2 If the Architect, Owner, or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection, or approval not included under Section 13.4.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection, or approval, by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.4.3, shall be at the Owner's expense.
- § 13.4.3 If procedures for testing, inspection, or approval under Sections 13.4.1 and 13.4.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure, including those of repeated procedures and compensation for the Architect's services and expenses, shall be at the Contractor's expense.
- § 13.4.4 Required certificates of testing, inspection, or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.
- § 13.4.5 If the Architect is to observe tests, inspections, or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.
- § 13.4.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

#### § 13.5 Interest

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at the rate the parties agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

### § 13.6 Written Notice

Written notice shall be deemed to have been duly served if delivered in person to the individual or a member of the firm or entity or to an officer at the corporation for which it was intended, or if delivered at or sent by registered or certified mail, or by courier service providing proof of delivery, to the last business address known to the party giving notice, or if delivered by facsimile or email to the offices of the person or corporation for which it was intended. For facsimiles or emails received after 5:00 p.m. on a business day, or on a weekend or legal holiday on which the recipient's offices are closed, notice shall be deemed to have been duly served on the next business day.

#### ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT

# § 14.1 Termination by the Contractor

- § 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Contractor, a Sub-contractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work, for any of the following reasons:
  - Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped;
  - .2 An act of government, such as a declaration of national emergency, that requires all Work to be stopped;
  - Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents; or
  - The Owner has failed to furnish to the Contractor reasonable evidence as required by Section 2.2.
- § 14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work, repeated suspensions, delays, or interruptions of the entire Work by the Owner as described in Section 14.3, constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.
- § 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, as well as reasonable overhead and profit on Work not executed, and costs incurred by reason of such termination.

§ 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor, a Sub-subcontractor, or their agents or employees or any other persons or entities performing portions of the Work because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

### § 14.2 Termination by the Owner for Cause

- § 14.2.1 The Owner may terminate the Contract if the Contractor
  - .1 repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
  - .2 fails to make payment to Subcontractors or suppliers in accordance with the respective agreements between the Contractor and the Subcontractors or suppliers;
  - .3 repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
  - 4 otherwise is guilty of substantial breach of a provision of the Contract Documents.
- § 14.2.2 When any of the reasons described in Section 14.2.1 exist, and upon certification by the Architect that sufficient cause exists to justify such action, the Owner may, without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:
  - .1 Accept assignment of subcontracts pursuant to Section 5.4; and
  - .2 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.
- § 14.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.
- § 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, upon application, and this obligation for payment shall survive termination of the Contract.

# § 14.3 Suspension by the Owner for Convenience

- § 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work, in whole or in part for such period of time as the Owner may determine.
- § 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay, or interruption under Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent
  - .1 that performance is, was, or would have been, so suspended, delayed, or interrupted, by another cause for which the Contractor is responsible; or
  - .2 that an equitable adjustment is made or denied under another provision of the Contract.

# § 14.4 Termination by the Owner for Convenience

- § 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.
- § 14.4.2 Upon receipt of notice from the Owner of such termination for the Owner's convenience, the Contractor shall
  - .1 cease operations as directed by the Owner in the notice;
  - .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and
  - .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

§ 14.4.3 In case of such termination for the Owner's convenience, the Construction Manager shall be entitled to receive payment for Work in accordance with the contract documents executed up to the effective date of termination.

#### ARTICLE 15 CLAIMS AND DISPUTES

§ 15.1 Claims

#### § 15.1.1 Definition

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, a change in the Contract Time, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim. This Section 15.1.1 does not require the Owner to file a Claim in order to impose liquidated damages in accordance with the Contract Documents.

# § 15.1.2 Time Limits on Claims

The Owner and Contractor shall commence all Claims and causes of action against the other and arising out of or related to the Contract, whether in contract, tort, breach of warranty or otherwise, in accordance with the requirements of the binding dispute resolution method selected in the Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work.

#### § 15.1.3 Notice of Claims

- § 15.1.3.1 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered prior to expiration of the period for correction of the Work set forth in Section 12.2.2, shall be initiated by notice to the other party and to the Initial Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. Claims by either party must be initiated within ninety (90) days after occurrence of the event giving rise to such Claim or within ninety (90) days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.
- § 15.1.3.2 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered after expiration of the period for correction of the Work set forth in Section 12.2.2, shall be initiated by notice to the other party. In such event, no decision by the Initial Decision Maker is required.

#### § 15.1.4 Continuing Contract Performance

- § 15.1.4.1 Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.
- § 15.1.4.2 The Contract Sum and Contract Time shall be adjusted in accordance with the Initial Decision Maker's decision, subject to the right of either party to proceed in accordance with this Article 15. The Architect will issue Certificates for Payment in accordance with the decision of the Initial Decision Maker.

#### § 15.1.5 Claims for Additional Cost

If the Contractor wishes to make a Claim for an increase in the Contract Sum, notice as provided in Section 15.1.3 shall be given before proceeding to execute the portion of the Work that is the subject of the Claim. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

# § 15.1.6 Claims for Additional Time

- § 15.1.6.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, notice as provided in Section 15.1.3 shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.
- § 15.1.6.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time and had an adverse effect on the scheduled construction.

(Paragraphs deleted)

§ 15.1.7 Limitation on Liability for Consequential Damages Contractor's liability for consequential damages arising out of or in connection with the performance or nonperformance of the Contractor's responsibilities or other obligations under this Agreement shall not exceed the total amount paid to Contractor under this Agreement, whether base on contract, tort, negligence, strict liability, or error and omission, provided that such limitation of liability for consequential damages: (1) shall not apply to Contractor's indemnity obligations under the Agreement; and (2) shall not apply to the policy limits or any amounts actually paid from any insurance Contractor is required to maintain pursuant to the terms of this Agreement.

# § 15.1.7 Waiver of Claims for Consequential Damages

The Contractor and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit, except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Section 15.1.7 shall be deemed to preclude assessment of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

#### § 15.2 Initial Decision

- § 15.2.1 Claims, excluding those arising under Sections 11.3.9 and 11.3.10, or claims alleging an error or omission by the Architect, shall be referred to the Initial Decision maker for initial decision. The Architect will serve as the Initial Decision maker, unless otherwise indicated in the Agreement. If the parties are unable to agree, any claim, dispute or matters arising out of the contract between the Architect, Owner and Construction Manager or any combination of those parties shall be submitted to a court of appropriate jurisdiction.
- § 15.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to resolve the Claim.
- § 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision. The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense.
- § 15.2.4 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of the request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished, or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part.
- § 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the so agree, and subject to legal or equitable proceedings in a court having jurisdiction thereof.

§ 15.2.6 (intentionally deleted)

# § 15.2.6.1 (intentionally deleted)

- § 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.
- § 15.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.
- § 15.2.9 The prevailing party in any judicial proceeding arising from the Contract Documents shall recover its reasonable and necessary attorneys' fees.

# § 15.3 Mediation

§ 15.3.1 If the parties to a dispute arising out of or related to the Contract agree to submit the Claim to mediation following a decision by the Initial Decision maker, the parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof. Nothing in the Contract Documents shall be construed as requiring mandatory mediation prior to litigation.

(Paragraphs deleted)